Cover Page



Universiteit Leiden



The handle <u>http://hdl.handle.net/1887/80955</u> holds various files of this Leiden University dissertation.

Author: Albulescu, I.C. Title: Targeting chikungunya virus replication : insights into chikungunya virus replication and the antiviral activity of suramin in vitro Issue Date: 2019-11-27

Targeting chikungunya virus replication

Insights into chikungunya virus replication and the antiviral activity of suramin *in vitro*

Irina Cristina Albulescu

PhD thesis, Leiden University, 2019.

The research described in this thesis was performed at Leiden University Medical Center, Department of Medical Microbiology, Leiden, The Netherlands.

This work was financially supported by the European Union 7th Framework Programme EUVIRNA (a Marie Curie Initial Training Network), grant agreement no. 264286, and the by the European Union's Horizon 2020 Research and Innovation Programme under ZIKAlliance grant agreement no. 734548.

Cover design by Roxana Drăgușel.

Printed by: Proefschrift-AIO.nl ISBN: 978-94-93184-16-9 Copyright © 2019, Irina C. Albulescu, all rights reserved

Targeting chikungunya virus replication

Insights into chikungunya virus replication and

the antiviral activity of suramin in vitro

Proefschrift

ter verkrijging van de graad van Doctor aan de Universiteit Leiden, op gezag van Rector Magnificus prof.mr. C.J.J.M. Stolker, volgens besluit van het College voor Promoties te verdedigen op woensdag 27 november 2019 klokke 16:15 uur

door

Irina Cristina Albulescu geboren te Petroșani, Romania in 1985

Promotor:	Prof. Dr. E.J. Snijder
Co-promotor:	Dr. M.J. van Hemert
Leden Promotiecommissie:	Prof. Dr. A.E. Gorbalenya
	Prof. Dr. J.M. Smit, UMC Groningen
	Dr. L. Delang, KU Leuven

"I know, I know the sun is hot Mosquitos come suck your blood Leave you there all alone Just skin and bone ... Swallow and chew Eat you alive All of us food That hasn't died"

(lyrics from the "Mosquito Song", by Queens of the Stone Age)

Contents

Chapter 1 - General introduction, aim, and outline of the thesis	9
Chapter 2 - An <i>in vitro</i> assay to study chikungunya virus RNA synthesis and the mode of action of inhibitors	41
Chapter 3 - Suramin inhibits chikungunya virus replication through multiple mechanisms	59
Chapter 4 - Suramin inhibits Zika virus replication through multiple mechanisms	79
Chapter 5 - Suramin inhibits chikungunya virus replication by interacting with virions and blocking the early events in infection (in preparation for submission)	95
Chapter 6 - General discussion and conclusion	117
Appendix	
List of Commonly Used Abbreviations	140
Summary	141
Samenvatting	143
Rezumat	145
Curriculum Vitae	147
List of Publications	148

Chapter 1

General introduction, aim and chapter outline

Introduction and aim

Viruses are submicroscopic entities that can infect organisms from all kingdoms of life. They are obligatory parasites and do not grow or multiply by division, but instead are assembled from components produced by their host, which also supplies them with the energy and protein synthesis required for virus replication. There are numerous viruses that can cause serious disease in humans, sometimes leading to problematic recovery or even death. They can be spread by contact with an infected individual via bodily fluids, rely on airborne transmission, or the bite of a virus-containing vector (mosquitos, ticks, etc.). The latter viruses are classified as arboviruses (arthropod-borne viruses). In this era of increased travel, urbanization and climate change, viruses have the potential to cause larger and faster spreading outbreaks. The recent chikungunya, Ebola, yellow fever, and Zika virus epidemics are only a few examples of this trend. Unfortunately, for many infections no specific vaccine or antiviral treatment is available.

Chikungunya virus (CHIKV) belongs to the arbovirus group and is a member of the *Togaviridae* family. CHIKV is transmitted to susceptible hosts through the bite of an infected mosquito. Although this virus was discovered in 1952 in what is currently Tanzania [1], research has intensified in the past two decades due to the large epidemics (and painful burden) that the virus has caused outside of Africa [2]. This led to increased interest in this neglected tropical virus, which is classified as a biosafety level 3 pathogen, meaning that special containment facilities are required to work with it.

This thesis aims to provide a deeper understanding of CHIKV replication in cell culture and to uncover particular characteristics that can facilitate the development of targeted inhibitors. It includes studies on the inhibition of CHIKV replication by the registered antiparasitic drug suramin. Suramin was found to also inhibit the replication of Zika virus, an arbovirus that emerged and caused a massive epidemic while the research on CHIKV described in this thesis was ongoing. More in-depth studies into suramin's mode of action were performed and the possibilities to repurpose this antiparasitic drug as a broadspectrum antiviral compound are discussed.

CHIKV and the Togaviridae family

CHIKV is a member of the *Togaviridae* family, which based on the Baltimore classification of viruses belongs to group IV, the positive-sense and single-stranded RNA (+ssRNA) viruses [3].

The *Togaviridae* family (toga=cloak in Greek) was established in 1974 [4], and to date it contains 32 members that are subdivided into two genera: *Rubivirus*, with rubella virus as its only member, and *Alphavirus* containing the other 31 species, listed at [5, 6]. Representative species and isolates are depicted in the phylogenetic tree presented in Fig. 1.



Figure 1. Phylogenetic tree with representative species and isolates from the *Alphavirus* genus. The tree was generated based on a conserved region of 2184 nt encoding part of the envelope proteins; the alphaviruses are further grouped depending on the antigenic complex they belong to, as well as their geographic distribution (adapted from [9]).

Togavirus nomenclature is based predominantly on their geographical origin, or, alternatively, refers to a feature of the disease they cause (*e.g.*, rubella virus – rubis = red in latin, causes a red rash; Venezuelan equine encephalitis virus or VEEV requires no further explanation). Sindbis virus (named after an Egyptian village) and Semliki Forest virus (originating from Uganda) are two of the most-studied alphaviruses, because their low

pathogenicity in humans allows them to be studied at lower containment facilities. Studies on SINV and SFV have led to many advancements in the understanding of the replication cycle of alphaviruses and viruses in general [7].

Togaviruses differ from one another from an antigenic point of view. They can, however, be grouped into seven antigenic complexes (established based on cross-neutralization assays), containing members that are more closely related to each other [8].

Host range and transmission

The host range of togaviruses is quite broad, as they are able to replicate in both arthropod vectors (mosquitoes, ticks, lice, and cliff swallow bugs) as well as vertebrate host such as humans, non-human primates, equines, seals, fish, birds, rodents, and swine [8, 10]. In the case of CHIKV, there is a sylvatic cycle with the virus circulating between mosquitoes (*Aedes* sp.) and non-human primates, with occasional spillover transmission into humans occurring in urbanized areas. Transmission is horizontal, predominantly via the bite of an infected vector, with the exception of Rubella virus that has airborne transmission and infects only humans [11], and Eilat virus, which only replicates in insect cells and is thought to be vertically transmitted in mosquitoes [12].

Geographical distribution of alphaviruses and epidemiology of CHIKV

The *Togaviridae* family distribution relies on favorable ecological conditions and preferred vectors as well as host availability. Rubella virus can be found worldwide, but is more prevalent in underdeveloped countries where vaccination is scarce infection is characterized by mild fever and rash, and is considered a measles-like disease that can have severe consequences for fetal development during pregnancy when maternal immunity is lacking [13]. The alphaviruses are thought to have a marine origin based on phylogenetic analysis of members infecting fish and marine mammals, and likely spread from the southern ocean across the rest of the world [14]. They are further separated into new-world and old-world alphaviruses (Fig. 1). New-world alphaviruses are mainly present in the Americas, and are known to cause encephalitis (in equines and humans) and arthritic/rheumatic syndromes. The Old-world alphaviruses, including CHIKV, are found in Africa, Asia, and Australia, and in humans predominantly cause arthritic/rheumatic syndromes [10].

The most prominent alphavirus members, which have raised awareness due to their pathology and high mortality rate, are the encephalitic viruses: Eastern equine encephalitis virus (EEEV), Western equine encephalitis virus (WEEV), and Venezuelan equine encephalitis virus or VEEV [15]. The arthritogenic viruses like CHIKV, Ross River virus (RRV), Barmah Forest virus (BFV), o'nyong-nyong virus (ONNV), Mayaro virus (MAV), SINV and SFV have caused occasional epidemics leading to a much larger number of infections [16]. Among the arthritogenic alphaviruses, CHIKV has caused the largest epidemics so far, exhibiting highly effective spread within immunologically naïve populations. The first time that CHIKV attracted widespread attention was in 2004-2008, when the epidemics that originated from Africa reached the island of La Reunion, after which CHIKV further spread into India and other countries surrounding the Indian Ocean, causing at least 6 million cases [17, 18]. This was followed by the 2013-2014 'New-world epidemic' with an estimated number of 2.9 million human infections [19]. As far as we know, CHIKV is also the only Old-world alphavirus that has crossed into the New world on two different occasions: in 2013 via the Caribbean island of Saint Martin [20] and in 2014 in Brazil [21].

Based on phylogenetic analysis, CHIKV is considered to have originated in Africa, about 500 years ago [22] and it can now be classified into three major genotypes (or lineages): West African (WA), East, central and South African (ECSA; including the Indian Ocean lineage (IOL) as a subgenotype) and Asian (including the Asian/American subgenotype) [23]. Aedes aegypti was considered to be the main vector for transmission, but in the early 2000s the IOL evolved from the ECSA genotype after acquiring a mutation (A226V) in the E1 envelope protein that allowed CHIKV to replicate in, and hence be transmitted by, *Aedes albopictus*, the Asian tiger mosquito, an avid daytime biter [24]. There are differences in virulence between CHIKV genotypes, with the Asian/American type being the less virulent as compared to the Asian, ECSA and WA viruses [25]. The difference in virulence of the Asian/American genotype is presumably caused by its shorter 3' UTR – as explained later in this chapter. The distribution of CHIKV genotypes is depicted in Fig. 2, alongside the territories of their main vectors, *Aedes aegypti* and *albopictus*.



Figure 2. Global distribution of mosquito vectors and areas where CHIKV is endemic (with permission from [23]).

Morphology and replication cycle of alphaviruses and CHIKV in particular

Virion morphology and characteristics

A typical alphavirus particle is icosahedral (T=4 symmetry) and has a diameter of approximately 70 nm (Fig. 3A). The outer layer consists of a lipid bilayer originating from the host cell, in which the envelope glycoproteins E1 and E2 are embedded as groups of three heterodimers, forming projections or spikes that give the particle its coarse surface (Fig. 3B). Below the envelope, the nucleocapsid core has a regular, icosahedral structure, and contains 240 copies of the capsid protein, which interacts with the cytoplasmic tails of the envelope proteins while packaging the genome. In the virion the stoichiometric ratio between the major structural proteins capsid: E1: E2 is presumed to be 1:1:1 [8].

Regarding their physical properties, it is known that alphaviruses have a buoyant density of 1.20 g/cm^3 in sucrose gradients and a sedimentation coefficient of ~280S, corresponding to a molecular mass of 5.2×10^7 Daltons [26, 27]. The infectivity of the particle is lost upon treatment with denaturing agents (formaldehyde, 70% ethanol) or detergents, acids, and lipid solvents [28]. Moreover, exposure to heat, UV light, or radiation greatly decreases infectivity ([29] and WHO Technical Report, Series No. 924, 2004, Annex 4).



Figure 3. Morphology of a CHIKV particle with ultrastructure details. **A.** Structure of the virion. **B.** View through a section of the virus based on a cryolectron microscopic reconstruction of CHIKV and VEEV particles. The insert shows the structure of an immature envelope spike [30], [31], while the lipid bilayer and the nucleocapsid are shown at atomic resolution (adapted from [2]).

Besides the basic structural components described above, some alphaviruses can contain other components, as is the case for Aura virus that can have a subgenomic RNA packaged alongside the genome [32].

Other alphavirus preparations have been found to contain additional virus proteins, such as the envelope protein E3 (as a sign that the particles contain immature spikes). The 6K or transframe protein TF [33] can also be present in sub-stoichiometric amounts, as determined

1

by mass spectrometry. However, further studies, e.g. using cryolectron microscopy, are needed to understand their arrangement (perhaps as oligomeric channels) within the virion [8]. Surprisingly, the non-structural protein 2 (nsP2) has been postulated to be important for virus encapsidation in the case of VEEV [34], and very recently has been shown to actually be incorporated into the virion, though not as a structural component [36]. Besides these additional virus proteins, virions can also contain small amounts of host proteins, such as ribosomal subunits [35] or phospholipid scramblase [36]. Particles generated in mosquito cells have a homogeneous density, while those obtained from vertebrate cells include two populations: light and heavy particles [35]. Furthermore, particles derived from mammalian cells contain considerably more cholesterol in their envelope. In general the mass of an alphavirus particle consists of 6% RNA, 62% proteins, 26% lipids, and 6% glycans, with the abundance of the latter two components depending on the cell type from which the particles were generated [27].

Alphavirus genome organization and protein function

In general, the alphavirus genome is a ≈ 11.8 -kilobase, single-stranded RNA equipped with a 5'-terminal m7G cap and a 3'-terminal poly-A tail (Fig. 3), functional features that help it mimic a host cell/eukaryotic mRNA in order to facilitate its direct translation by the host cell ribosomes. The genome contains two open reading frames, ORF1 and ORF2, encoding the non-structural (nsPs) or the structural proteins, respectively.

Alphavirus genomes have a type-0 cap that consists of a 7meGpppA and in the cytoplasm of the infected cell its formation is mediated by the activities of the non-structural proteins (nsP1 and nsP2, as explained below in more detail). It differs from the cellular type-1 cap structure, 7meGppp2meG, as it lacks the second methyl group. Interestingly, it should be noted that for the particles produced early during the infection of mammalian cells, not all genomes are capped, leading to and impairment in their infectivity because of this; it also implies that alphaviruses, for some reason, rely on a mix t of infectious and non-infectious particles for infectivity [37]. The poly(A) tail is ~70 nt long and requires a minimum length of 11-12 nt in order to support the interaction with the host's poly (A) binding protein (PABP), which in turn interacts with initiation factors (eIF α , eIF4E, etc.) bound to the 5' cap in order to circularize the RNA for efficient translation [38].

Besides the ORFs, the genome also contains untranslated regions or UTRs (some authors use the term NTR, for non-translated region) that show a high variability in length and sequence within the alphavirus genus, but also between strains of the same species. UTRs are located at either end of the genome and are important for the regulation of viral gene expression. Conserved sequence elements CSE1-4 play important roles in the alphavirus replication cycle and the nsP1- or nsP2-encoding region of the genome contains a sequence that is crucial for RNA encapsidation (table 1). These RNA sequence elements interact with the nsPs, capsid or host proteins and are crucial for key processes in the virus replication cycle.

Element	Location	Functions
CSE1	Within UTR1	Promoter for genomic RNA synthesis; recognized on the complementary -RNA by RTCs consisting of partly or fully processed nsPs
CSE2	Beginning of ORF1	Important for -RNA synthesis
CSE3	Between ORF1 and ORF2	Promoter for subgenomic RNA synthesis; recognized on the complementary -RNA by RTCs containing fully processed nsPs
CSE4	Within UTR2	Promoter for -RNA synthesis, perhaps in conjunction with CSE2, recognized RTCs composed of nsP123 and nsP4
Virion packaging sequences	Within nsP1 or nsP2	Allow specific packaging of the genomic RNA in the presence of a the more abundant sgRNA

 Table 1. Conserved sequence elements identified within the genome of alphaviruses, adapted from [8].

The 3' UTR (or UTR2) of CHIKV is about 498-732 nt long and is located upstream of the poly(A) tail. Its sequence contains a series of direct-repeat elements (DRs 1-3, also known as RSEs, repeated sequence elements) and stem-loop structures, as well as a conserved sequence element, CSE4, immediately upstream of the poly A tail [39]. Although there is great sequence variation within UTR2 of the different CHIKV genotypes, these conserved elements are required for genome replication and stability, and efficient transmission. Therefore, the genetic changes that have occurred in the UTR2 sequence of the CHIKV Asian/American sub-genotype (namely a gain in length) are responsible for the impaired replication in the vector and consequently these viruses are less virulent compared to the other genotypes [40].

The 5' UTR (or UTR1) is about 76-77 nt long in the case of CHIKV, and contains CSE1 that acts as a promoter for +RNA synthesis from its complement that forms the 3' sequence of the negative strand. This element, as well as several stable secondary structures in the region, are involved in the interactions with host factors, leading to initiation of translation. This region also contributes to evasion of the innate immune response by antagonizing IFIT1, an IFN-stimulated gene encoding a protein that limits translation of non-self RNA [41].

The proteins involved in RNA replication and transcription are produced/translated from ORF1 as a polyprotein consisting of the four non-structural proteins (nsP1-4). In many alphaviruses an opal stop codon is present towards the end of the coding sequence of nsP3,

in order to downregulate the production of the viral RNA-dependent RNA polymerase, nsP4. Ribosomes can read through this relatively weak stop codon in ~10% of the cases, leading to the production of the nsP1234 and nsP123 polyproteins in a ratio of ~1:10 [42]. For SFV, ONNV, and CHIKV, it has been shown that not all isolates have the opal stop codon and that it is sometimes replaced by an arginine codon. This is attributed to evolutionary pressures allowing for the existence of both variants and suggesting that both are necessary for virus fitness in the vertebrate and invertebrate host [43-45]. The proteolytic processing of the non-structural polyproteins is carried out by nsP2, the largest of the proteins, with multiple functions that are listed in table 2. The N-terminal Cys protease activity of nsP2 is responsible for the following cleavages: rapid *cis*-cleavage of the site RAGG/Y between nsP3/4, *trans*-cleavage of the sites RAGA/G and RAGC/A between nsP1/2 and nsP2/3 respectively. The latter requires the accumulation of nsP123 [43], thereby controlling the switch from negative strand synthesis done by RTCs composed of nsP123 and nsP4 to synthesis of +RNA by complexes composed of fully cleaved nsPs.

The structural proteins are not produced by direct translation of the genome, but are derived from a polyprotein expressed from the single sgRNA. Also in this situation two polyproteins can be translated: capsid-E3-E2-6K-E1 and, following a frameshift within the 6K protein-coding region, the minor product capsid-E3-E2-TF is produced. The processing of these polyproteins starts with the release of the capsid protein, which catalyzes its own liberation, and next the other sites are gradually cleaved by host proteases (signal peptidase, furin) while trafficking through the secretory pathway.

1



Figure 4. Schematic representation of the alphavirus genome, its translation products, polyprotein processing, and the types of RNA produced during replication/transcription. Upon genome translation the polyproteins nsP123 and nsP1234 are produced, the latter after read-through of the opal stop codon between nsp3 and nsP4. The cis-cleavage of nsP1234 produces the replicase complex (nsP123+nsP4) that synthesizes only the negative sense RNA (upper part of the figure). As the level of nsP123 increases, trans-cleavage occurs generating a short-lived nsP1+nsP23+nsP4 complex and the final nsP1+nsP2+nsP3+nsP4 complex, which is involved in synthesizing only positive sense RNA (g and sgRNA) – lower part of the figure. The structural proteins are translated from the sgRNA, also as two polyproteins due to a frameshift within the 6K protein-coding region: capsid-E3-E2-6K-E1 and capsid-E3-E2-TF. Capsid catalyzes its release from the polyprotein and the remaining polyprotein is processed by host proteases into individual proteins while trafficking along the secretory pathway. (Adapted from <u>https://talk.ictvonline.org/ictv-reports/ictv_online_report/positive-sense-rna-viruses/w/togaviridae</u>)

As outlined above, the alphavirus genome is relatively small, encoding only 10 proteins - four non-structural and six structural components (Fig. 4); therefore, like the majority of viruses, alphaviruses strongly rely on host factors, metabolic pathways and cellular building blocks for their replication. The proteins encoded by CHIKV, their size, and functions are summarized in table 2 and are further detailed in the description of the replicative cycle.

The alphavirus replication cycle: general and CHIKV-specific features

The CHIKV replicative cycle, as is the case for all alphaviruses, exhibits some host cellspecific differences in invertebrate cell and vertebrate cells [46]. Alphaviruses cause cytolytic infections in mammalian cells while establishing a noncytopathic, persistent infection in mosquito cells. The replication cycle in invertebrate cells is therefore different than in the vertebrate ones [36]. Regardless of the host cell, several key steps can be distinguished in the replicative cycle of alphaviruses: entry, genome translation, RNA replication and transcription, and biogenesis of new particles. These steps are represented in Fig. 5, together with the multitude of stages that occur in vertebrate and invertebrate host cells.

Virus entry and its stages

Entry, the process by which a virion delivers its genome to the viral replication site, in this case the cytoplasm, can be divided in several stages: virus attachment, uptake via receptormediated endocytosis (RME), and membrane fusion in order to release the nucleocapsid (numbered 1 to 5 in Fig. 5). This topic is covered in much detail in [47].

The E2 protein likely plays a role in the broad host range of alphaviruses, as it is involved in receptor interaction, and could have multiple binding sites for various cellular receptors. Alphavirus E2 proteins likely interact with receptors that are ubiquitously expressed and highly conserved among the various host species [8].

Attachment is thought to require co-factors as GAGs (such as heparan sulfate) or adhesion molecules (such as DC-SIGN, L-SING, C-type lectins, NRAMP, NRAMP2) that are very abundant on the surface of vertebrate cells [8]. This can facilitate the subsequent interaction with a specific receptor, such as matrix remodeling-associated protein 8 (Mxra8) [48] or the epidermal growth factor receptor pathway substrate (EPS15) [49], in the case of CHIKV. The choice of co-factor and/or receptor is thought to be cell type-dependent as well as CHIKV strain-specific.

Once bound to a receptor, protein rearrangements can occur while the virus is taken up by receptor-mediated endocytosis (RME) in early endosomes that will mature by fusing with lysosomes. Endosomal acidification will trigger conformational changes that destabilize the spike heterotrimers leading to the exposure of the E1 fusion loop. This loop will be inserted in the target membrane in a cholesterol-dependent fashion, followed by trimerization of E1 protein and subsequent fusion with the (late) endosomal membrane. Surprisingly, alphavirus infection could be achieved in the absence of clathrin-mediated endocytosis as has been shown for SINV [50], and the fusion events can also take place directly at the plasma membrane of the host cell.

Following fusion with the endosomal or the plasma membranes, the nucleocapsid is released in the cytoplasm where it undergoes disassembly in order to proceed to genome translation. It is unknown how this happens, but it might be triggered by the ribosomes, more specifically ribosomal RNA that could interact with the capsid protein, destabilizing the core [8]. This hypothesis is plausible, since it has been shown for SINV that the presence of host-derived ribosomal subunits within the virion can speed up the start of the translation [35].

Protein	Size (amino acids)	Main enzymatic activities and other functions
nsP1	535	Methyltransferase and guanylyl transferase activity for capping of viral RNA; Membrane anchor for the replicase complex
nsP2	798	N-terminal NTPase, helicase, and RNA triphosphatase activities; involved in processing of nonstructural polyproteins (either in cis or trans) with its C-terminal cysteine protease activity
nsP3	523 or 529(*)	Phosphoprotein that contains macro domain and SH3-binding regions and an opal stop codon; it interacts with host factors and is involved in negative-strand RNA synthesis;
nsP123	1856 or 1863(*)	Polyprotein that together with nsP4 is involved in negative-strand RNA synthesis (replication)
nsP4	611	RNA-dependent RNA polymerase (RdRp) with terminal transferase activity; it is unstable and produced in lower amounts, as compared to the other nsPs
capsid	261	Structural protein that together with the RNA form the nucleocapsid core; its carboxyl domain has autocatalytic serine protease activity
pE2	487	Precursor polyprotein containing E3 and E2; processed by host furin-like protease in the trans-Golgi vesicles; incorporated into virions at a low level
E3	64	N-terminal domain is uncleaved leader peptide for E2; glycosylated; may help shield fusion peptide in E1 during egress
E2	423	Component of the envelope projection mediating binding to receptors and attachment factors on the host cell membrane; glycosylated and palmitoylated; main target of neutralizing antibodies
6K	61	Leader peptide for E1; functions as an ion channel and may enhance particle release
TF	76	Transframe protein alternatively produced by ribosomal frameshifting; has the N-terminus and the ion channel function as 6K; may enhance particle release
El	439	Type II fusion protein and component of the envelope surface projections; mediates fusion of viral envelope with cellular membranes via fusion peptide at acidic pH

 Table 2.
 Summary of the CHIKV proteins, polyproteins and intermediates, with their individual functions (adapted from [7, 8, 23].

 (*) The longer variant is generated in the cases when the opal stop codon is read-through

RNA replication and its stages

The next 6-9 stages of the virus replication cycle as depicted in Fig. 5 are the steps required for translation, replication and transcription of viral RNA. Translation of genomic RNA (gRNA) leads to the production of the non-structural proteins 1-4, and is initiated by recruiting the host translation machinery. It is hypothesized that alphavirus genomes are translated in the same fashion as cellular mRNAs (a process assisted by the eukaryotic initiation factors, eIFs), since later in infection their translation is downregulated [51, 52]. At first, the polyproteins nsP123 and nsP1234 are produced (in different amounts, as explained earlier), which will be gradually processed by the Cys-protease activity of nsP2. Out of these four proteins, only nsP1 is believed to interact with membranes, either through palmitoylated cysteine residues (C417-419) or by using an amphipathic α -helix formed by amino acid residues 245-264 [8]. By interacting with the plasma membrane these polyproteins are involved in the formation spherules, which are the sites of viral RNA replication and transcription [53]. Because nsP3 has been shown to interact with ampiphysin-1 and -2, proteins involved in membrane curvature, it might have an important role in the generation of these structures [54]. In the spherule environment, the RNA is protected from degradation by host RNases, miRNA targeting or/and detection by the innate immune system. The nsP123 and nsP4 complex is thought to form the bottleneck of the spherule and is responsible, as previously mentioned, for initiating replication by interacting with CSE4 and CSE2, leading to the production of the negative-strand RNA. This will be the template from which more genomes, as well as the sgRNA will be transcribed. The synthesis of the aforementioned three types of viral RNA, requires the functions of all four nsPs as listed in table 2.

Besides their direct involvement in RNA replication and transcription, the nsPs have additional activities within the infected cell that are beneficial for the viral replication cycle. The interplay between host factors and CHIKV nsPs was extensively discussed by Wong and Chu [55] and in particular for nsP2 and nsP3 a variety of interactions with the host have been described. Despite lacking a known nuclear localization signal, nsP2 can translocate to the nucleus where it mediates the degradation of Rpb1 and many other subunits of the RNApolymerase II complex [56, 57], thus leading to a transcriptional host shut-off that could contribute to evasion of innate immune responses [58]. NsP3 was shown to interact with G3BP1 and G3BP2, components of stress granules, which are believed to be involved in the inhibition of translation. nsP3 sequesters G3BPs, which blocks the formation of true stress granules during infection, while early in infection this interaction appears to facilitate the switch from translation to viral RNA replication [59, 60]. Activation of the PI3K-Akt-mTOR pathway by nsP3, which is dependent on its C-terminal part (last 50 residues), causes spherules to pinch off into early endosomes and by fusion with lysosomes they mature in the cytoplasm into type I cytopathic vacuoles [61].

Virion assembly and egress

Stages 9-14 as depicted in Fig. 5 concern the events that lead to the assembly and release of new particles.

The structural proteins are produced upon translation of the sgRNA at a later stage in infection, when cellular and genomic RNA translation are downregulated [52]. Therefore, it is possible that the sgRNA is translated through a different mechanism, perhaps relying on the interaction with host proteins such as the heterogeneous nuclear ribonucleoproteins (hnRNPs) [62]. During translation, the capsid protein liberates itself from the polyprotein, after which the rest of the polyprotein is partially translocated across the ER membrane, a process that is assisted by the transmembrane domains of the E proteins and dependent on an N-terminal signal sequence located in E3. Following its membrane association, the polyprotein undergoes cleavage of the 6K or TF protein by signal-peptidases (step 11) and, as it travels across the secretory pathway towards the plasma membrane, it also undergoes posttranslational modifications (N-glycosylation and palmitoylation), as well as cleavage of the E3-E2 proteins by furin (step 12).

In parallel, newly produced gRNA is packaged by capsid subunits in the cytoplasm (step 13). Another type of cytopathic vacuoles are formed, CPV type II, which originate from the trans-Golgi network. CPV II can interact with nucleocapsids, and contain E1 and E2 glycoproteins arranged as helical tube-like structures resembling the trimeric organization from the envelope of the mature virions. These vacuoles are located closer to the plasma membrane, which is the site of virus budding, suggesting that they might be involved in the transport of structural components to the budding site [63].

The nucleocapsids reach the plasma membrane and interact with the cytoplasmic sides of the envelope proteins that will trigger changes in the curvature of the membrane leading to the budding of newly formed particles.

The steps of virus replication in invertebrate cells are quite similar to those in vertebrate cells, except that replication complexes are associated with endosomes in so called "virus factories", which are also the site of protein synthesis and assembly of new particles. The newly assembled virions are released into the extracellular space by fusion of the endosome with the plasma membrane [46].



Figure 5. Schematic representation of the alphavirus replication cycle in mammalian (A) and insect (B) cells. (A) Upon attachment and receptor binding (steps 1 and 2), alphaviruses are internalized via endocytosis (step 3). At low pH the virus fuses (step 4) with the late endosome membrane and the nucleocapsid is released in the cytosol. After disassembly (step 5), the non-structural proteins (nsPs) are expressed from the genomic RNA (Step 6). Together with host proteins, the nsPs form RTCs (step 7) that will replicate and transcribe the viral RNA (step 8) producing the -RNA that serves as a template from which genomic and subgenomic RNA are transcribed. The RTCs perform their activity in spherules located at the plasma membrane or on modified endosomes termed cytopathic vacuoles type I (CPV I). The structural proteins are expressed from the sgRNA (step 9) and after capsid self-cleavage, the rest of the polyprotein is translocated to the ER (step 10) where it undergoes processing by signalase (Step 11) and several post-translational modifications (glycosylation) when the proteins reach the Golgi complex. Here the host protease furin cleaves the E3 from E2 (step 12) and the E1-E2 pre-formed heterodimers are trafficked towards the plasma membrane or to CPV-II. The capsid protein will interact with genomic RNA and form nucleocapsids (step 13) that will reach CPV-II or directly the plasma membrane in order to bud off into new virus particles (step 14). (B) In the mosquito cell, replication progresses in a similar manner, but with some differences. These concern the site of replication, which takes place on internal large vesicles that also contain nucleocapsids and internally budded virus particles (step 15). The newly formed virus particles accumulate in large vesicles that are released from the plasma membrane (with permission from [46]).

Not all released particles will be infectious. On the one hand, this can be due to immature surface projections from which E3 is not removed or due to the lack of proper post-translational modifications. On the other hand, it is possible that new virions with properly processed and arranged proteins are formed, which carry a defective genome that is incapable of starting a new round of replication when it is released into the cytoplasm of a new host cell. This is a consequence of the high error rate of the RdRp, which leads to the accumulation of mutations that can be detrimental (e.g. introducing stop codons, changing the reading frame, or point mutations that affect structure or function of proteins or regulatory RNA elements). However, the relatively high error rate of the RdRp also

provides an advantage, as it allows the generation of a large genetic diversity – quasispecies, which enhances the capability of the virus to rapidly adapt to changing selective pressures, e.g. host immune responses or replication in a novel host cell [64].

Clinical signs and pathogenesis of CHIKV infection

In humans CHIKV causes a non-lethal but debilitating, untreatable, and usually selflimiting disease characterized by extreme joint pains that can persist for months.

Clinical symptoms

Upon the bite of an infected mosquito, around 85 % of infected people develop chikunguya fever (CHIKF) symptoms, while 15% or less have asymptomatic seroconversion [2]. The onset of symptoms is rapid, after a mean incubation period of 3 days (range 2 to 6 days), and is characterized by high fever (39-40°C), intense asthenia, myalgia, rash, and headache. The polyarthralgia is the specific symptom to which CHIKV owes its name, as the word chikungunya describes a person bent over by the debilitating joint pain, in Kimakonde, the language of the Makonde tribe that lives in the border region of Mozambique and Tanzania. Polyarthralgia is symmetrical and localized in distant extremities (arms and legs). Acute arthritis may also occur in the interphalangeal joints, wrists, and ankles. This acute phase lasts around 7 days and is dominated by joint pain or/and arthritis. Some unusual symptoms were also noted, such as lymphadenopathy, digestive disorders, and pruritus. Complications of this disease were described for the extreme age groups (neonates and elderly) as well as for those with underlying comorbidities. They are rare, and include myocarditis, ocular disease, hepatitis, acute renal failure, and neurological disorders (meningoencephalitis, Guillane-Barre syndrome). In many patients arthritis can persist for months to years, suggesting the ability of CHIKV to induce immune pathogenic mechanisms specific for chronic disease, by potentially persisting in the joint tissues and replicating at a low level, leading to continuous stimulation of the immune system.



Figure 6. Representation of acute and chronic chikungunya fever symptoms with underlying events (with permission from [23]).

Mechanism of dissemination and pathogenesis

An important role in virus dissemination can be attributed to the mosquito saliva that has been shown to enhance the infectivity of arboviruses [65]. After reaching the skin of its host, CHIKV infects and replicates in fibroblasts as well as macrophages, then spreads via the lymphatic system to specific sites: lymph nodes, spleen, muscle, skin, peripheral joints, tendons, but also to aspecific sites as brain and liver (see Fig. 6, left side).

Viral replication can lead to high levels of viremia, >10° particles/mL in the blood, which is correlated with the acute onset of the disease. This will initiate the immune response elevation of pro-inflammatory cytokines, chemokines, growth factors, and type I IFNs. It also leads to activation and proliferation of T lymphocytes (CD8- and CD4-positive), which contribute to virus clearance. IgM antibody levels are detected within 5-6 days (overlapping with the acute phase of the disease) and IgG antibodies are detected 7-10 days after the onset of chikungunya fever, and their presence aids with the infection clearing. The chronic phase (Fig. 6, right side), follows a partial clearance of the infection (no CHIKV titers detected), but perhaps antigens still remain in many tissues because of persistence in infected cells: endothelial cells, macrophages, mononuclear cells, satellite cells in the muscle [23]. In some individuals this can last for years, leading to long-term impairment of the quality of life [66]. In rare cases, some patients can suffer from bone erosions due to the CHIKV-induced arthritis [67].

Much of the above information has been acquired by long-term monitoring of patients recovering from CHIKV infection, but also by using animal models – a wide range of mouse and non-human primate models are available to study the progression of CHIKV infection [68]. These approaches have offered a more in-depth view of the molecular mechanisms involved in CHIKV pathogenesis, persistence, and (difficulties in) clearance.

Chikungunya fever diagnosis

Accurate chikungunya fever diagnosis has been an issue for decades, because it has been mistaken for malaria or dengue fever, the major infections in tropical areas. It can also be misdiagnosed as rheumatoid arthritis, since CHIKV-induced joint pain can mimic this autoimmune disease, though the onset of arthritis is usually abrupt in CHIKF [69]. Another contributing factor is the absence of fast and cost-effective diagnostic methods, and unlike flaviviruses (Zika virus), CHIKV can be detected reliably only in blood, and not saliva or urine [70].

At the moment, the World Health Organization recommends several methods for diagnosis. For samples collected soon after the onset of symptoms virus-detection methods like RT-PCR (that can also allow proper genotyping of the virus) can be used as well as serological methods. At later stages, the detection of infection can be done only by serological tests to determine the presence of IgM and IgG anti-CHIKV antibodies [71]. Virus isolation can also be performed from blood shortly after the symptoms have appeared and viral RNA extracted from blood should directly be used for genotyping, preferably by NGS, to avoid the accumulation of adaptive mutations due to repeated passaging in cell culture. Passaging of the isolated virus on mammalian cells or in various animal models, as was conventionally done, could lead to accumulation of adaptive mutations that would hamper proper characterization of the original clinal isolate.

Recently a new and promising method for rapid diagnosis of CHIKV infection, even in DENV co-infected samples, has been proposed, which is based on CHIKV antigen detection by immune-chromatography [72].

Preventive and antiviral strategies against CHIKV infection

Preventive strategies/prophylaxis

In the case of untreatable (infectious) diseases, prevention is the best strategy for controlling the spread of the infection. At the same time, substantial focus should also be on developing proper treatments in order to reduce the symptoms and cure those already infected. Prevention can be achieved by vector control, and by vaccinating the population in endemic areas.

Since CHIKV is an arbovirus, vector control is an important strategy for containing an epidemic. Wearing long clothes and using insect repellent sprays are recommended measures to be taken in areas with endemic CHIKV (or other arboviruses and parasites). Mosquito nets and insecticide-impregnated curtains should be used to limit vector entry into homes and prevent nighttime bites. Additionally, insecticides can be sprayed or added to treat the water from containers (in order to kill the immature larvae or adult mosquitoes) during outbreaks. The use of insecticides can be detrimental for public health, as well as the environment (affecting multiple ecosystems), and it can ultimately become inefficient due to insecticide resistance. A more recently developed approach for vector control is the production of genetically engineered (GE) male mosquitoes that upon release will produce offspring unable to grow in the absence of tetracycline [73]. Another promising strategy could be the use of *Wolbachia*-infected mosquitos, which appears to diminish vector competence for CHIKV (as well as dengue virus). More research should be dedicated to these approaches in order to establish whether the use of GE insects poses a risk to the environment [74].

Passive immunization could also be considered as a treatment alternative, since the humoral immune response is the one limiting the infection. Several mouse studies have shown the success of administering anti-CHIKV antibodies (either as antisera or isolated monoclonal antibodies) in protecting against the disease [23, 75]. Nevertheless, in general vaccines have been one of the most efficient ways of infectious disease prevention, and even eradication. The factors that contribute to efficiency of a vaccine efficient are still not completely understood today and the requirements for bringing a vaccine to the market, for use in humans, have become increasingly strict and time-consuming.

The most successful attempt, in the case of alphavirus vaccines, that led to a marketed product is re the TC-83 IND vaccine against the encephalitic alphavirus VEEV, which was shown to have good efficiency in equines [76]. In the case of CHIKV, multiple attempts have been made (reviewed by [77]), but no vaccine has made it to the market, e.g. due to reasons such as insufficient market interest or financing. One of the most promising attempts was done by the US Army Medical Research Institute, that have produced the live-attenuated vaccine (LAV) candidate TS1-GSD-218 (or 181/clone25) based on an Asian

CHIKV strain, AF15561. The production was stopped after the phase II clinical trial, when 8% of the human subjects developed mild arthralgia [78]. Also the dominant attenuation mutation of that strain, G82R in E2, was observed to revert to wild-type in mice and humans leading to safety concerns ([23] and references herein). However, due to the massive last two epidemics, the global market for a Chikungunya vaccine is estimated at approximately €500 million annually (estimate supported by independent market studies – www.valneva. com), and therefor new vaccine candidates are being tested in clinical trials (see table 3).

Vaccine	Platform	Phase of clinical research	Company
VRC-CHKVLP059-	Measles vectored	Phase II	National Institute of
00-VP			Allergy and Infectious
			Diseases (NIAID)
MV-CHIK	Measles vectored	Phase II	Themis Bioscience GmbH
VLA1553	LAV	Phase I; data available early 2019	Valneva
PXVX0317 CHIKV- VLP	VLP	Phase II; with or without Alhydrogel adjuvant; study completion December 2020	PaxVax Inc.
VAL-181388	Not specified	Phase I	Moderna Therapeutics
ChAdOx1 Chik	Not specified	Phase I	University of Oxford

Table 3. A summary of the CHIKV vaccines currently in clinical trials.

New promising avenues in vaccine development are also being explored, such as the production of a CHIKV/IRESv1 vaccine candidate that contains an encephalomyocarditis virus internal ribosomal entry site (IRES) instead of the subgenomic promoter. This limits the production of the structural proteins and at the same time blocks the replication in insect cells, where the IRES is not functional [25].

Antiviral treatment perspectives

It has been historically proven that vaccines are one of the most efficient ways to prevent or limit outbreaks of infectious disease. Unfortunately, there are issues with their largescale production, especially in a short timeframe, and there are concerns related to storage/ distribution in tropical areas. Another downside of vaccines is that their target must be known or related to a (pandemic) pathogen variant that is already known, otherwise a long development process will have to be taken into consideration. Vaccines can become obsolete when (immune) escape mutants of the target virus emerge.

Consequently, antiviral drugs are needed as a therapeutic alternative, offering the advantages that they are relatively easy to produce and store, compared to most vaccines. However, designing effective and specific antiviral drugs is challenging. Viruses extensively rely on their host for replication and thus it is difficult to target virus replication without affecting cellular processes/pathways.

However, viruses express proteins with unique virus-specific functions, which can be targeted by direct acting antiviral compounds (DAAs) and such compounds have been successfully developed for the treatment of viral infections caused by e.g. flu, human immunodeficiency virus (HIV), hepatitis C virus (HCV), human cytomegalovirus (hCMV), varicella zoster virus (VZV), respiratory syncytial virus (RSV), and others [79]. Alternatively, cellular factors crucial for certain steps of the viral replication cycle can be targeted with host-directed antivirals (HDAs), circumventing the potentially rapid development of antiviral resistance, which is often observed when using DAAs. In the case of CHIKV, compounds targeting host factors (HDAs) as well as viral proteins (DAAs) have been reviewed extensively [55, 80-82]. Some of the more relevant therapeutic approaches are discussed here.

Host-targeting antivirals (HDAs) against (alpha)viruses

Of the many approved drugs that are on the market for other conditions, several have been shown to also have antiviral properties, likely because their targets are important factors for the replication of certain viruses. Such compounds have been identified for several viruses, by e.g. screening of FDA-approved drug libraries.

Based on knowledge of host factors involved in replication, specific cellular functions can be targeted. For instance, inhibition of the activities of furin or heat shock protein 90 (HSP90) – involved in particle maturation and (viral) protein folding, respectively, has been investigated ([81], references herein). However, a major bottleneck for the success of this approach is the associated side-effects due a general cytotoxicity of these compounds. Side effects also can be expected when considering to modulate the activity of kinases, which are involved in regulation of a multitude of complex signaling pathways within cells. Inhibition of sphingosine kinase 2 (SK2), a host factor that co-localizes with CHIKV RTCs and is

essential during infection, has been proposed as a potential therapeutic strategy [83]. Not only inhibition of kinases should be considered, as activation of protein kinase C (PKC) by prostatin had a negative impact on CHIKV replication [84].

Another host-directed anti-CHIKV strategy concerns interference with the metabolism of polyamines by the use of difluoromethylornithine (DFMO or effornithine, marketed as Vaniqa[®] for the treatment of second stage African trypanosomiasis) [85].

The immunosuppressing drug mycophenolic acid (MPA), sold as Myfortic[®] or CellCept[®] for its immunosuppressant role, was also shown to inhibit CHIKV replication [86]. This is due to its inhibitory action on the inosine monophosphate dehydrogenase (IMPDH), which leads to a strong depletion of the intracellular GTP pool, hence indirectly affecting virus replication. However, the immunosuppressive properties of this compound complicate its application in the treatment of infections.

Arguably more promising host-directed antiviral strategies would be to use drugs that upregulate host factors with antiviral activities, such as virperin, a protein that participates in the synthesis of the chain terminator ddhCTP. In this manner the toxic effects could perhaps be bypassed, since viperin is naturally produced in the human body [87].

Immune modulators that stimulate the activity of retinoic acid inducible gene I (RIG-I), a host protein that recognizes dsRNA and triggers antiviral pathways that will lead to the inhibition of virus replication, have also shown to be promising molecules [88].

Recently, the inhibition of Granzyme A, a promoter of arthritic inflammation, with Serpinb6b has been identified as a potential treatment for CHIKV infections. Granzyme A levels were elevated in CHIKV-infected non-human primates as well as a cohort of human patients and treatment with Serpinb6b reduced CHIKV-induced arthritic inflammation in a mouse model [89].

In clinical practice, several marketed therapeutics have been used with moderate success to alleviate the symptoms of patients with chronic CHIKV-induced arthritis. These include non-steroidal anti-inflammatory drugs (NSAIDs), short-term course of corticosteroids and disease modifying antirheumatic drugs (DMARDs) [90].

Direct acting antivirals (DAAs) against alphaviruses

Chloroquine, marketed as Aralen[®] for the treatment of malaria and additionally displaying a broad antiviral activity in cell culture, was shown to be an entry inhibitor of CHIKV as well, but in clinical trials it did not prove to be effective and even led to increased arthralgia complaints from treated patients [91]. In a non-human primate model chloroquine was even stimulating CHIKV replication while delaying cell and humoral immune responses [92].

CHIKV protein synthesis, and subsequently virus replication, was strongly inhibited by the use of siRNA or shRNA targeting combinations of non-structural and structural genes, and these results were successfully confirmed in animal experiments [93, 94].

Small molecules targeting the capping functions of nsP1 are a promising strategy, since alphaviruses employ unique enzymatic activities for their RNA capping and therefore this cytoplasmic process could be specifically targeted with minimal cytotoxic effects [95, 96]. Most efforts involved targeting of the Cys-protease function of nsP2, mainly through the in silico identification of inhibitors, followed by assessment of their activity using *in vitro* enzymatic assays [80]. The challenge with protease inhibitors is to obtain high selectivity, i.e. they should not inhibit host Cys-proteases.

Ribavirin is a nucleoside analogue, marketed as Rebetol[®] (and various other names) that has proven to effectively inhibit CHIKV replication in patients. It can act through various mechanisms, either by blocking the activity of IMPDH, an enzyme involved in GTP metabolisms (so, through an indirect effect). Additionally, it could also interfere with viral RNA capping or, by being incorporated into newly synthesized viral RNA, it can induce lethal mutagenesis ([82], references herein).

Similarly, 6-Azauridine (a uridine analogue), marketed for the treatment of psoriasis, has been shown to have a broad-spectrum antiviral activity. Its anti-CHIKV activity has been validated in multiple *in vitro* studies, and it is suspected to act indirectly by interfering with cellular UTP metabolism, or to act as DAA upon its incorporation into viral RNA leading to lethal mutagenesis ([97] and references herein). It requires further pre-clinical evaluation in an appropriate animal model, before it can be tested in the clinic.

Though many of the nucleoside analogues with anti-CHIKV activity that have been identified so far, seem to inhibit virus replication by blocking enzymes involved in the nucleoside metabolism, there are some that potentially act directly on the alphavirus RNA polymerase. β -D-N(4)-hydroxycytidine (NHC) is a potent anti-VEEV drug that causes accumulation of mutations in virus-specific RNAs, and could potentially have a broader anti-alphavirus activity [98]. Similarly, favipiravir is a compound marketed as an anti-influenza A drug, but it has been shown to have a broader antiviral activity, since it can inhibit the replication of several noroviruses, bunyaviruses, flaviviruses, and alphaviruses [99]. Favipiravir specifically targets the viral RdRps, including that of CHIKV [100]. The compound has a broad spectrum of activity in cell culture and animal models, is already approved for the treatment of influenza infections in Japan and clinical trials for other infections are ongoing. Therefore, it could be an inexpensive and fast option for the development of anti-CHIKV therapy.

Due to the lengthy process of bringing a new compound to the market, there is great interest in antiviral screening using previously approved drugs or those considered safe in clinical trials but not yet on the market. This approach is called drug repositioning or repurposing [101] and is an expanding field of research. A number of compounds that have become "block busters' are described by Naylor S. *et al* [102], and an interesting example (that the opposite route is possible as well) is Gemzar[®] (Eli Lilly), a nucleoside-based inhibitor that was intended as an antiviral compound but ended up on the market for the treatment of various cancers. Suramin is one of the oldest drugs on the market, intensively studied in the past 30 years for its incredibly broad spectrum, and it is anti-CHIKV and anti-ZIKV properties are discussed in chapters 3, 4, and 5 of this thesis.



Figure 7. The chemical structure of suramin hexasodium. The full IUPAC nomenclature is available at PubChem (https://pubchem.ncbi.nlm.nih.gov/).

Suramin was synthesized in 1916 by scientists working at Bayer AG, and is still commercialized, also under the original name Germanin[®]. Its discovery was urged by the colonial powers being exposed to vast epidemics of sleeping sickness in Africa. The disease is transmitted through the bite of the Tsetse flies infected with *Trypanosoma* sp. Suramin, a "colorless dye" related to trypan blue, became the therapy for the first stage of the infection [103]. Soon after, it was proven to be effective also in the treatment of another parasitic infection, onchocerciasis. The mode of action of suramin against these parasites remains unclear until now. It is suspected that the compound impairs the parasites' energy metabolism, leading to their death.

The chemical properties, pharmacokinetics, toxicity and pharmacological actions of suramin have been reviewed in detail [104]. It has a high affinity for proteins, including serum proteins, which leads to a long half-life in humans. Studies on the administration of suramin at high doses and or over a long period of time, which led to accumulation in the plasma, revealed a number of side effects, including fever, skin reactions, malaise, reversible polyneuropathy, adrenal insufficiency, vomiting, blood clotting inhibition, etc. Nevertheless, due to its success in treating African trypanosomiasis, suramin is on the WHO list of essential medicines, 20th edition [105], a list summarizing the most effective and safe medicines required in a health-care system.

Additional pharmacological activities of suramin have been discovered [106, 107]. Due to its capacity to interact with growth factor receptors, suramin has anti-proliferative effects and therefore has been evaluated in cancer therapy [108]. In low doses, suramin has been shown to improve the outcome of children with autism spectrum disorder (ASD), and therefore it is also considered a promising new therapy in the field of neurological disorders [109].

Last but not least, the antiviral activity of suramin has also been explored, and the findings are summarized in the introduction of chapter 3 with all the relevant references [110]. Briefly, through *in vitro* studies it was identified as an inhibitor of viral helicases, polymerases, and reverse-transcriptases. In cell-based assays, the mechanism of action seemed to be different from the one observed *in vitro*, and surprisingly similar for a multitude of viruses - suramin blocks the early events of virus entry. Suramin was proven to be effective against enterovirus A71, the etiological agent of hand foot and mouth disease in humans. This finding was also confirmed in mouse and non-human primate animal models, which showed that suramin treatment lowered the viral burden and decreased mortality, leading to a faster recovery and better outcome [111].

Thesis outline

The main focus of this thesis is on CHIKV replication and the anti-CHIKV activity of suramin, a compound which also inhibited replication of ZIKV, another medically important arbovirus that emerged more recently.

Chapter 1 contains the general introduction that summarizes the current knowledge regarding alphavirus genome organization, protein function, replication, CHIKV pathogenesis and preventive or therapeutic strategies.

Chapter 2 presents the development of an *in vitro* replication assay (IVRA) that can be used to study CHIKV RNA synthesis, as well as to identify inhibitors of this process and perform mode of action studies on these compounds.

Chapter 3 describes the identification of suramin as a direct-acting inhibitor of CHIKV RNA synthesis *in vitro*. Cell-based assays revealed that suramin's main mode of action is dependent on the inhibition of an early step of the replicative cycle and this is also the hypothesis for its antiviral activity *in vivo*.

Chapter 4 shows the antiviral effect of suramin is indeed quite broad, as it can also inhibit ZIKV replication by interfering with virus entry and biogenesis of virions, at a later stage of the replication cycle.

Chapter 5 describes the identification of the CHIKV E2 protein as the target of suramin and mode of action studies that suggest that suramin blocks attachment to cells, and subsequent fusion of the viral particle with cellular membranes.

Chapter 6 contains the general discussion based on the key findings presented in this thesis and their implications, followed by a short conclusion.

References

- Robinson MC. 1955. An epidemic of virus disease in Southern Province, Tanganyika Territory, in 1952-53.
 I. Clinical features. Trans R Soc Trop Med Hyg 49:28-32.
- Weaver SC, Lecuit M. 2015. Chikungunya virus and the global spread of a mosquito-borne disease. N Engl J Med 372:1231-9.
- 3. Baltimore D. 1971. Expression of animal virus genomes. Bacteriol Rev 35:235-41.
- Fenner F, Pereira, H. G., Poerterfield, J. S., Joklik, W. K., Downie, A. W. 1974. Family and Generic Names for Viruses Approved by the International Committee on Taxonomy of Viruses, June 1974. https://talk. ictvonline.org/ICTV/proposals/Ratification_1974.pdf. Accessed
- ICTV. 2018. List of species in the Alphavirus genus, on ICTV. https://talk.ictvonline.org/ictv-reports/ ictv_9th_report/positive-sense-rna-viruses-2011/w/posrna_viruses/275/Togaviridae. Accessed
- Chen R, Mukhopadhyay S, Merits A, Bolling B, Nasar F, Coffey LL, Powers A, Weaver SC, Ictv Report C. 2018. ICTV Virus Taxonomy Profile: *Togaviridae*. J Gen Virol 99:761-762.
- 7. Acheson NH. 2011. Fundamentals of molecular virology, 2nd ed. John Wiley & Sons, Hoboken, NJ.
- Kuhn RJ. 2013. Fields' Virology (Chapter 22 Togaviride), p 629-650. In D. M. Knipe & P. M. Howley. Philadelphia ULWW (ed), Fields' Virology.
- ICTV. 2018. Togaviridae. https://talk.ictvonline.org/ictv-reports/ictv_online_report/positive-sense-rnaviruses/w/Togaviridae. Accessed
- Fros JJ, Pijlman GP. 2016. Alphavirus Infection: Host Cell Shut-Off and Inhibition of Antiviral Responses. Viruses 8.
- 11. Parkman PD. 1996. Chapter 55 Togaviruses: Rubella Virus. *In* Baron S (ed), Medical Microbiology, 4th edition, 4th ed, Galveston (TX).
- 12. Nasar F, Palacios G, Gorchakov RV, Guzman H, Da Rosa AP, Savji N, Popov VL, Sherman MB, Lipkin WI, Tesh RB, Weaver SC. 2012. Eilat virus, a unique alphavirus with host range restricted to insects by RNA replication. Proc Natl Acad Sci U S A 109:14622-7.
- 13. Lambert N, Strebel P, Orenstein W, Icenogle J, Poland GA. 2015. Rubella. Lancet 385:2297-307.
- Forrester NL, Palacios G, Tesh RB, Savji N, Guzman H, Sherman M, Weaver SC, Lipkin WI. 2012. Genome-scale phylogeny of the alphavirus genus suggests a marine origin. J Virol 86:2729-38.
- 15. Zacks MA, Paessler S. 2010. Encephalitic alphaviruses. Vet Microbiol 140:281-6.
- 16. Suhrbier A, Jaffar-Bandjee MC, Gasque P. 2012. Arthritogenic alphaviruses--an overview. Nat Rev Rheumatol 8:420-9.
- Schwartz O, Albert ML. 2010. Biology and pathogenesis of chikungunya virus. Nat Rev Microbiol 8:491-500.
- Staples JE, Breiman RF, Powers AM. 2009. Chikungunya fever: an epidemiological review of a re-emerging infectious disease. Clin Infect Dis 49:942-8.
- Yactayo S, Staples JE, Millot V, Cibrelus L, Ramon-Pardo P. 2016. Epidemiology of Chikungunya in the Americas. J Infect Dis 214:S441-S445.
- Cassadou S, Boucau S, Petit-Sinturel M, Huc P, Leparc-Goffart I, Ledrans M. 2014. Emergence of chikungunya fever on the French side of Saint Martin island, October to December 2013. Euro Surveill 19.
- 21. Nunes MR, Faria NR, de Vasconcelos JM, Golding N, Kraemer MU, de Oliveira LF, Azevedo Rdo S, da Silva DE, da Silva EV, da Silva SP, Carvalho VL, Coelho GE, Cruz AC, Rodrigues SG, Vianez JL, Jr., Nunes BT, Cardoso JF, Tesh RB, Hay SI, Pybus OG, Vasconcelos PF. 2015. Emergence and potential for spread of Chikungunya virus in Brazil. BMC Med 13:102.
- 22. Volk SM, Chen R, Tsetsarkin KA, Adams AP, Garcia TI, Sall AA, Nasar F, Schuh AJ, Holmes EC, Higgs S, Maharaj PD, Brault AC, Weaver SC. 2010. Genome-scale phylogenetic analyses of chikungunya virus reveal independent emergences of recent epidemics and various evolutionary rates. J Virol 84:6497-504.
- 23. Silva LA, Dermody TS. 2017. Chikungunya virus: epidemiology, replication, disease mechanisms, and prospective intervention strategies. J Clin Invest 127:737-749.

- 24. Tsetsarkin KA, Vanlandingham DL, McGee CE, Higgs S. 2007. A single mutation in chikungunya virus affects vector specificity and epidemic potential. PLoS Pathog 3:e201.
- Langsjoen RM, Haller SL, Roy CJ, Vinet-Oliphant H, Bergren NA, Erasmus JH, Livengood JA, Powell TD, Weaver SC, Rossi SL. 2018. Chikungunya Virus Strains Show Lineage-Specific Variations in Virulence and Cross-Protective Ability in Murine and Nonhuman Primate Models. MBio 9.
- Jose J, Snyder JE, Kuhn RJ. 2009. A structural and functional perspective of alphavirus replication and assembly. Future Microbiol 4:837-56.
- 27. Koblet H. 1990. The "Merry-Go-Round": Alphaviruses Between Vertebrate and Invertebrate Cells, p 343-402. *In* KARL MARAMOROSCH FAM, AARON J. SHATKIN (ed), Advances in Virus Research.
- Roberts PL, Lloyd D. 2007. Virus inactivation by protein denaturants used in affinity chromatography. Biologicals 35:343-7.
- Park SL, Huang YJ, Hsu WW, Hettenbach SM, Higgs S, Vanlandingham DL. 2016. Virus-specific thermostability and heat inactivation profiles of alphaviruses. J Virol Methods 234:152-5.
- Voss JE, Vaney MC, Duquerroy S, Vonrhein C, Girard-Blanc C, Crublet E, Thompson A, Bricogne G, Rey FA. 2010. Glycoprotein organization of Chikungunya virus particles revealed by X-ray crystallography. Nature 468:709-12.
- Zhang R, Hryc CF, Cong Y, Liu X, Jakana J, Gorchakov R, Baker ML, Weaver SC, Chiu W. 2011. 4.4 A cryo-EM structure of an enveloped alphavirus Venezuelan equine encephalitis virus. EMBO J 30:3854-63.
- 32. Rumenapf T, Strauss EG, Strauss JH. 1994. Subgenomic mRNA of Aura alphavirus is packaged into virions. J Virol 68:56-62.
- Snyder JE, Kulcsar KA, Schultz KL, Riley CP, Neary JT, Marr S, Jose J, Griffin DE, Kuhn RJ. 2013. Functional characterization of the alphavirus TF protein. J Virol 87:8511-23.
- Kim DY, Atasheva S, Frolova EI, Frolov I. 2013. Venezuelan equine encephalitis virus nsP2 protein regulates packaging of the viral genome into infectious virions. J Virol 87:4202-13.
- Sokoloski KJ, Snyder AJ, Liu NH, Hayes CA, Mukhopadhyay S, Hardy RW. 2013. Encapsidation of hostderived factors correlates with enhanced infectivity of Sindbis virus. J Virol 87:12216-26.
- 36. Schuchman R, Kilianski A, Piper A, Vancini R, Ribeiro JMC, Sprague TR, Nasar F, Boyd G, Hernandez R, Glaros T. 2018. Comparative Characterization of the Sindbis Virus Proteome from Mammalian and Invertebrate Hosts Identifies nsP2 as a Component of the Virus Nucleocapsid and Sorting Nexin 5 as a Significant Host Factor for Alphavirus Replication. J Virol doi:10.1128/JVI.00694-18.
- Sokoloski KJ, Haist KC, Morrison TE, Mukhopadhyay S, Hardy RW. 2015. Noncapped Alphavirus Genomic RNAs and Their Role during Infection. J Virol 89:6080-92.
- Lemay JF, Lemieux C, St-Andre O, Bachand F. 2010. Crossing the borders: poly(A)-binding proteins working on both sides of the fence. RNA Biol 7:291-5.
- Chen R, Wang E, Tsetsarkin KA, Weaver SC. 2013. Chikungunya virus 3' untranslated region: adaptation to mosquitoes and a population bottleneck as major evolutionary forces. PLoS Pathog 9:e1003591.
- 40. Morley VJ, Noval MG, Chen R, Weaver SC, Vignuzzi M, Stapleford KA, Turner PE. 2018. Chikungunya virus evolution following a large 3'UTR deletion results in host-specific molecular changes in protein-coding regions. Virus Evol 4:vey012.
- Hyde JL, Chen R, Trobaugh DW, Diamond MS, Weaver SC, Klimstra WB, Wilusz J. 2015. The 5' and 3' ends of alphavirus RNAs--Non-coding is not non-functional. Virus Res 206:99-107.
- 42. Li G, Rice CM. 1993. The signal for translational readthrough of a UGA codon in Sindbis virus RNA involves a single cytidine residue immediately downstream of the termination codon. J Virol 67:5062-7.
- Rupp JC, Sokoloski KJ, Gebhart NN, Hardy RW. 2015. Alphavirus RNA synthesis and non-structural protein functions. J Gen Virol 96:2483-500.
- Myles KM, Kelly CL, Ledermann JP, Powers AM. 2006. Effects of an opal termination codon preceding the nsP4 gene sequence in the O'Nyong-Nyong virus genome on *Anopheles gambiae* infectivity. J Virol 80:4992-7.
- 45. Jones JE, Long KM, Whitmore AC, Sanders W, Thurlow LR, Brown JA, Morrison CR, Vincent H, Peck KM, Browning C, Moorman N, Lim JK, Heise MT. 2017. Disruption of the Opal Stop Codon Attenuates Chikungunya Virus-Induced Arthritis and Pathology. MBio 8.
- 46. Jose J, Taylor AB, Kuhn RJ. 2017. Spatial and Temporal Analysis of Alphavirus Replication and Assembly in Mammalian and Mosquito Cells. MBio 8.
- van Duijl-Richter MK, Hoornweg TE, Rodenhuis-Zybert IA, Smit JM. 2015. Early Events in Chikungunya Virus Infection-From Virus Cell Binding to Membrane Fusion. Viruses 7:3647-74.
- Zhang R, Kim AS, Fox JM, Nair S, Basore K, Klimstra WB, Rimkunas R, Fong RH, Lin H, Poddar S, Crowe JE, Jr., Doranz BJ, Fremont DH, Diamond MS. 2018. Mxra8 is a receptor for multiple arthritogenic alphaviruses. Nature 557:570-574.
- Bernard E, Solignat M, Gay B, Chazal N, Higgs S, Devaux C, Briant L. 2010. Endocytosis of chikungunya virus into mammalian cells: role of clathrin and early endosomal compartments. PLoS One 5:e11479.
- 50. Vancini R, Wang G, Ferreira D, Hernandez R, Brown DT. 2013. Alphavirus genome delivery occurs directly at the plasma membrane in a time- and temperature-dependent process. J Virol 87:4352-9.
- Toribio R, Diaz-Lopez I, Boskovic J, Ventoso I. 2018. Translation initiation of alphavirus mRNA reveals new insights into the topology of the 48S initiation complex. Nucleic Acids Res 46:4176-4187.
- 52. Carrasco L, Sanz MA, Gonzalez-Almela E. 2018. The Regulation of Translation in Alphavirus-Infected Cells. Viruses 10.
- Pietila MK, Hellstrom K, Ahola T. 2017. Alphavirus polymerase and RNA replication. Virus Res 234:44-57.
- Neuvonen M, Kazlauskas A, Martikainen M, Hinkkanen A, Ahola T, Saksela K. 2011. SH3 domainmediated recruitment of host cell amphiphysins by alphavirus nsP3 promotes viral RNA replication. PLoS Pathog 7:e1002383.
- 55. Wong KZ, Chu JJH. 2018. The Interplay of Viral and Host Factors in Chikungunya Virus Infection: Targets for Antiviral Strategies. Viruses 10.
- Akhrymuk I, Kulemzin SV, Frolova EI. 2012. Evasion of the innate immune response: the Old World alphavirus nsP2 protein induces rapid degradation of Rpb1, a catalytic subunit of RNA polymerase II. J Virol 86:7180-91.
- 57. Treffers EE, Tas A, Scholte FE, Van MN, Heemskerk MT, de Ru AH, Snijder EJ, van Hemert MJ, van Veelen PA. 2015. Temporal SILAC-based quantitative proteomics identifies host factors involved in chikungunya virus replication. Proteomics 15:2267-80.
- Akhrymuk I, Frolov I, Frolova EI. 2018. Sindbis Virus Infection Causes Cell Death by nsP2-Induced Transcriptional Shutoff or by nsP3-Dependent Translational Shutoff. J Virol 92.
- Fros JJ, Domeradzka NE, Baggen J, Geertsema C, Flipse J, Vlak JM, Pijlman GP. 2012. Chikungunya virus nsP3 blocks stress granule assembly by recruitment of G3BP into cytoplasmic foci. J Virol 86:10873-9.
- Scholte FE, Tas A, Albulescu IC, Zusinaite E, Merits A, Snijder EJ, van Hemert MJ. 2015. Stress Granule Components G3BP1 and G3BP2 Play a Proviral Role Early in Chikungunya Virus Replication. J Virol 89:4457-69.
- 61. Thaa B, Biasiotto R, Eng K, Neuvonen M, Gotte B, Rheinemann L, Mutso M, Utt A, Varghese F, Balistreri G, Merits A, Ahola T, McInerney GM. 2015. Differential Phosphatidylinositol-3-Kinase-Akt-mTOR Activation by Semliki Forest and Chikungunya Viruses Is Dependent on nsP3 and Connected to Replication Complex Internalization. J Virol 89:11420-37.
- LaPointe AT, Gebhart NN, Meller ME, Hardy RW, Sokoloski KJ. 2018. The Identification and Characterization of Sindbis Virus RNA:Host Protein Interactions. J Virol doi:10.1128/JVI.02171-17.
- Soonsawad P, Xing L, Milla E, Espinoza JM, Kawano M, Marko M, Hsieh C, Furukawa H, Kawasaki M, Weerachatyanukul W, Srivastava R, Barnett SW, Srivastava IK, Cheng RH. 2010. Structural evidence of glycoprotein assembly in cellular membrane compartments prior to Alphavirus budding. J Virol 84:11145-51.
- 64. Coffey LL, Beeharry Y, Borderia AV, Blanc H, Vignuzzi M. 2011. Arbovirus high fidelity variant loses fitness in mosquitoes and mice. Proc Natl Acad Sci U S A 108:16038-43.
- 65. Fong SW, Kini RM, Ng LFP. 2018. Mosquito Saliva Reshapes Alphavirus Infection and Immunopathogenesis. J Virol 92.
- 66. van Aalst M, Nelen CM, Goorhuis A, Stijnis C, Grobusch MP. 2017. Long-term sequelae of chikungunya virus disease: A systematic review. Travel Med Infect Dis 15:8-22.

- 67. Manimunda SP, Vijayachari P, Uppoor R, Sugunan AP, Singh SS, Rai SK, Sudeep AB, Muruganandam N, Chaitanya IK, Guruprasad DR. 2010. Clinical progression of chikungunya fever during acute and chronic arthritic stages and the changes in joint morphology as revealed by imaging. Trans R Soc Trop Med Hyg 104:392-9.
- Haese NN, Broeckel RM, Hawman DW, Heise MT, Morrison TE, Streblow DN. 2016. Animal Models of Chikungunya Virus Infection and Disease. J Infect Dis 214:S482-S487.
- Burt FJ, Chen W, Miner JJ, Lenschow DJ, Merits A, Schnettler E, Kohl A, Rudd PA, Taylor A, Herrero LJ, Zaid A, Ng LFP, Mahalingam S. 2017. Chikungunya virus: an update on the biology and pathogenesis of this emerging pathogen. Lancet Infect Dis 17:e107-e117.
- 70. Musso D, Teissier A, Rouault E, Teururai S, de Pina JJ, Nhan TX. 2016. Detection of chikungunya virus in saliva and urine. Virol J 13:102.
- 71. WHO. 2017. Chikungunya. http://www.who.int/news-room/fact-sheets/detail/chikungunya. Accessed
- 72. Jain J, Okabayashi T, Kaur N, Nakayama E, Shioda T, Gaind R, Kurosu T, Sunil S. 2018. Evaluation of an immunochromatography rapid diagnosis kit for detection of chikungunya virus antigen in India, a dengue-endemic country. Virol J 15:84.
- Phuc HK, Andreasen MH, Burton RS, Vass C, Epton MJ, Pape G, Fu G, Condon KC, Scaife S, Donnelly CA, Coleman PG, White-Cooper H, Alphey L. 2007. Late-acting dominant lethal genetic systems and mosquito control. BMC Biol 5:11.
- Meghani Z, Boete C. 2018. Genetically engineered mosquitoes, Zika and other arboviruses, community engagement, costs, and patents: Ethical issues. PLoS Negl Trop Dis 12:e0006501.
- Clayton AM. 2016. Monoclonal Antibodies as Prophylactic and Therapeutic Agents Against Chikungunya Virus. J Infect Dis 214:S506-S509.
- 76. Paessler S, Weaver SC. 2009. Vaccines for Venezuelan equine encephalitis. Vaccine 27 Suppl 4:D80-5.
- Schwameis M, Buchtele N, Wadowski PP, Schoergenhofer C, Jilma B. 2016. Chikungunya vaccines in development. Hum Vaccin Immunother 12:716-31.
- Edelman R, Tacket CO, Wasserman SS, Bodison SA, Perry JG, Mangiafico JA. 2000. Phase II safety and immunogenicity study of live chikungunya virus vaccine TSI-GSD-218. Am J Trop Med Hyg 62:681-5.
- 79. De Clercq E, Li G. 2016. Approved Antiviral Drugs over the Past 50 Years. Clin Microbiol Rev 29:695-747.
- Ching KC, L FPN, Chai CLL. 2017. A compendium of small molecule direct-acting and host-targeting inhibitors as therapies against alphaviruses. J Antimicrob Chemother 72:2973-2989.
- Subudhi BB, Chattopadhyay S, Mishra P, Kumar A. 2018. Current Strategies for Inhibition of Chikungunya Infection. Viruses 10.
- Abdelnabi R, Neyts J, Delang L. 2015. Towards antivirals against chikungunya virus. Antiviral Res 121:59-68.
- Reid SP, Tritsch SR, Kota K, Chiang CY, Dong L, Kenny T, Brueggemann EE, Ward MD, Cazares LH, Bavari S. 2015. Sphingosine kinase 2 is a chikungunya virus host factor co-localized with the viral replication complex. Emerg Microbes Infect 4:e61.
- Abdelnabi R, Amrun SN, Ng LF, Leyssen P, Neyts J, Delang L. 2017. Protein kinases C as potential host targets for the inhibition of chikungunya virus replication. Antiviral Res 139:79-87.
- Mounce BC, Cesaro T, Moratorio G, Hooikaas PJ, Yakovleva A, Werneke SW, Smith EC, Poirier EZ, Simon-Loriere E, Prot M, Tamietti C, Vitry S, Volle R, Khou C, Frenkiel MP, Sakuntabhai A, Delpeyroux F, Pardigon N, Flamand M, Barba-Spaeth G, Lafon M, Denison MR, Albert ML, Vignuzzi M. 2016. Inhibition of Polyamine Biosynthesis Is a Broad-Spectrum Strategy against RNA Viruses. J Virol 90:9683-9692.
- 86. Khan M, Dhanwani R, Patro IK, Rao PV, Parida MM. 2011. Cellular IMPDH enzyme activity is a potential target for the inhibition of Chikungunya virus replication and virus induced apoptosis in cultured mammalian cells. Antiviral Res 89:1-8.
- Gizzi AS, Grove TL, Arnold JJ, Jose J, Jangra RK, Garforth SJ, Du Q, Cahill SM, Dulyaninova NG, Love JD, Chandran K, Bresnick AR, Cameron CE, Almo SC. 2018. A naturally occurring antiviral ribonucleotide encoded by the human genome. Nature 558:610-614.

- Olagnier D, Scholte FE, Chiang C, Albulescu IC, Nichols C, He Z, Lin R, Snijder EJ, van Hemert MJ, Hiscott J. 2014. Inhibition of dengue and chikungunya virus infections by RIG-I-mediated type I interferon-independent stimulation of the innate antiviral response. J Virol 88:4180-94.
- 89. Wilson JA, Prow NA, Schroder WA, Ellis JJ, Cumming HE, Gearing LJ, Poo YS, Taylor A, Hertzog PJ, Di Giallonardo F, Hueston L, Le Grand R, Tang B, Le TT, Gardner J, Mahalingam S, Roques P, Bird PI, Suhrbier A. 2017. RNA-Seq analysis of chikungunya virus infection and identification of granzyme A as a major promoter of arthritic inflammation. PLoS Pathog 13:e1006155.
- Goupil BA, Mores CN. 2016. A Review of Chikungunya Virus-induced Arthralgia: Clinical Manifestations, Therapeutics, and Pathogenesis. Open Rheumatol J 10:129-140.
- De Lamballerie X, Boisson V, Reynier JC, Enault S, Charrel RN, Flahault A, Roques P, Le Grand R. 2008. On chikungunya acute infection and chloroquine treatment. Vector Borne Zoonotic Dis 8:837-9.
- 92. Roques P, Thiberville SD, Dupuis-Maguiraga L, Lum FM, Labadie K, Martinon F, Gras G, Lebon P, Ng LFP, de Lamballerie X, Le Grand R. 2018. Paradoxical Effect of Chloroquine Treatment in Enhancing Chikungunya Virus Infection. Viruses 10.
- 93. Parashar D, Paingankar MS, Kumar S, Gokhale MD, Sudeep AB, Shinde SB, Arankalle VA. 2013. Administration of E2 and NS1 siRNAs inhibit chikungunya virus replication *in vitro* and protects mice infected with the virus. PLoS Negl Trop Dis 7:e2405.
- Lam S, Chen KC, Ng MM, Chu JJ. 2012. Expression of plasmid-based shRNA against the E1 and nsP1 genes effectively silenced Chikungunya virus replication. PLoS One 7:e46396.
- 95. Delang L, Li C, Tas A, Querat G, Albulescu IC, De Burghgraeve T, Guerrero NA, Gigante A, Piorkowski G, Decroly E, Jochmans D, Canard B, Snijder EJ, Perez-Perez MJ, van Hemert MJ, Coutard B, Leyssen P, Neyts J. 2016. The viral capping enzyme nsP1: a novel target for the inhibition of chikungunya virus infection. Sci Rep 6:31819.
- Feibelman KM, Fuller BP, Li L, LaBarbera DV, Geiss BJ. 2018. Identification of small molecule inhibitors of the Chikungunya virus nsP1 RNA capping enzyme. Antiviral Res 154:124-131.
- Scholte FE, Tas A, Martina BE, Cordioli P, Narayanan K, Makino S, Snijder EJ, van Hemert MJ. 2013. Characterization of synthetic Chikungunya viruses based on the consensus sequence of recent E1-226V isolates. PLoS One 8:e71047.
- Urakova N, Kuznetsova V, Crossman DK, Sokratian A, Guthrie DB, Kolykhalov AA, Lockwood MA, Natchus MG, Crowley MR, Painter GR, Frolova EI, Frolov I. 2017. beta-D-N(4)-hydroxycytidine is a potent anti-alphavirus compound that induces high level of mutations in viral genome. J Virol doi:10.1128/ JVI.01965-17.
- 99. Delang L, Segura Guerrero N, Tas A, Querat G, Pastorino B, Froeyen M, Dallmeier K, Jochmans D, Herdewijn P, Bello F, Snijder EJ, de Lamballerie X, Martina B, Neyts J, van Hemert MJ, Leyssen P. 2014. Mutations in the chikungunya virus non-structural proteins cause resistance to favipiravir (T-705), a broad-spectrum antiviral. J Antimicrob Chemother 69:2770-84.
- 100. Abdelnabi R, Morais ATS, Leyssen P, Imbert I, Beaucourt S, Blanc H, Froeyen M, Vignuzzi M, Canard B, Neyts J, Delang L. 2017. Understanding the Mechanism of the Broad-Spectrum Antiviral Activity of Favipiravir (T-705): Key Role of the F1 Motif of the Viral Polymerase. J Virol 91.
- 101. Corsello SM, Bittker JA, Liu Z, Gould J, McCarren P, Hirschman JE, Johnston SE, Vrcic A, Wong B, Khan M, Asiedu J, Narayan R, Mader CC, Subramanian A, Golub TR. 2017. The Drug Repurposing Hub: a next-generation drug library and information resource. Nat Med 23:405-408.
- Naylor S, Kauppi, D.M., Schonfeld, J.M. 2015. Therapeutic Drug Repurposing, Repositioning and Rescut. Drug Discovery World Spring.
- 103. Steverding D. 2008. The history of African trypanosomiasis. Parasit Vectors 1:3.
- Voogd TE, Vansterkenburg EL, Wilting J, Janssen LH. 1993. Recent research on the biological activity of suramin. Pharmacol Rev 45:177-203.
- 105. WHO. 2017. WHO Model LIst of Essential Medicines, 20th edition. WHO, WHO Medicines website.
- McGeary RP, Bennett AJ, Tran QB, Cosgrove KL, Ross BP. 2008. Suramin: clinical uses and structureactivity relationships. Mini Rev Med Chem 8:1384-94.

- Tsay SC, Gupta, N.K., Bachurin, S.O., Hwu, J.R. 2017. Pharmaceutical and Biological Effects of Suramin and Potential of Its Derivatives and Analogues. Organic and Medicinal Chemistry International Journal 2.
- Stein CA. 1993. Suramin: a novel antineoplastic agent with multiple potential mechanisms of action. Cancer Res 53:2239-48.
- 109. Naviaux RK, Curtis B, Li K, Naviaux JC, Bright AT, Reiner GE, Westerfield M, Goh S, Alaynick WA, Wang L, Capparelli EV, Adams C, Sun J, Jain S, He F, Arellano DA, Mash LE, Chukoskie L, Lincoln A, Townsend J. 2017. Low-dose suramin in autism spectrum disorder: a small, phase I/II, randomized clinical trial. Ann Clin Transl Neurol 4:491-505.
- 110. Albulescu IC, van Hoolwerff M, Wolters LA, Bottaro E, Nastruzzi C, Yang SC, Tsay SC, Hwu JR, Snijder EJ, van Hemert MJ. 2015. Suramin inhibits chikungunya virus replication through multiple mechanisms. Antiviral Res 121:39-46.
- 111. Ren P, Zou G, Bailly B, Xu S, Zeng M, Chen X, Shen L, Zhang Y, Guillon P, Arenzana-Seisdedos F, Buchy P, Li J, von Itzstein M, Li Q, Altmeyer R. 2014. The approved pediatric drug suramin identified as a clinical candidate for the treatment of EV71 infection[mdash]suramin inhibits EV71 infection *in vitro* and *in vivo*. Emerg Microbes Infect 3:e62.

Chapter 2

An *in vitro* assay to study chikungunya virus RNA synthesis and the mode of action of inhibitors

Irina C. Albulescu*, Ali Tas*, Florine E.M. Scholte, Eric J. Snijder and Martijn J. van Hemert[§]

Affiliation: Molecular Virology Laboratory, Department of Medical Microbiology, Leiden University Medical Center, Leiden, The Netherlands.

> *Joint first authorship. [§]Corresponding author.

Published in Journal of General Virology, 2014 Dec; 95(Pt 12):2683-92.

Summary

Chikungunya virus (CHIKV) is a re-emerging mosquito-borne alphavirus that causes severe persistent arthralgia. To better understand the molecular details of CHIKV RNA synthesis and the mode of action of inhibitors, we have developed an in vitro assay to study CHIKV replication/transcription complexes isolated from infected cells. In this assay ³²P-CTP was incorporated into CHIKV genome and subgenomic (sg) RNA, as well as into a ~7.5 kb positive-stranded RNA, termed RNA II. We mapped RNA II, which was also found in CHIKV-infected cells, to the 5'-end of the genome up to the start of the sgRNA promoter region. Most of the RNA-synthesizing activity, negative-stranded RNA and a relatively large proportion of nsP1 and nsP4 were recovered from a crude membrane fraction obtained by pelleting at 15,000 x g. Positive-stranded RNA was mainly found in the cytosolic S15 fraction, suggesting it was released from the membrane-associated RTCs. The newly synthesized RNA was relatively stable and remained protected from cellular nucleases, possibly by encapsidation. A set of compounds that inhibit CHIKV replication in cell culture was tested in the in vitro RTC assay. In contrast to 3'dNTPs, which act as chain terminators and were potent inhibitors of RTC activity, ribavirin triphosphate and 6-aza-UTP did not affect the RNA-synthesizing activity in vitro. In conclusion, this in vitro assay for CHIKV RNA synthesis is a useful tool for mechanistic studies on the CHIKV RTC and mode of action studies on compounds with anti-CHIKV activity.

Introduction

Chikungunya virus (CHIKV) is an alphavirus that is mainly transmitted by *Aedes* mosquitoes and in general causes a severe arthralgia that may persist for months. The virus re-emerged in 2005 in an epidemic form and has since affected millions of people mainly in Asia [1]. Hundreds of infected travelers have arrived in the USA [2] and Europe, which even led to local transmission in Italy in 2007 and France in 2010 [3, 4]. A massive, CHIKV outbreak that started in October 2013 on the Caribbean island of Saint Martin marked [5] the arrival of CHIKV in the Americas [6, 7] and up to July 2014 over 100,000 cases were already reported [8]. These outbreaks illustrate the increasing disease burden of CHIKV, for which there is still no registered vaccine or specific antiviral therapy.

The CHIKV replication cycle involves the synthesis of negative-stranded RNA (-RNA) early in infection, which serves as a template for the subsequent production of the genome and a subgenomic (sg) mRNA. Genome and sgRNA are capped and polyadenylated and serve to express polyproteins comprising the nonstructural proteins (nsPs) 1-4 and structural proteins C, E3, E2, 6K and E1, respectively. CHIKV RNA synthesis presumably takes place on the cytoplasmic side of the plasma membrane and/or modified endosomal membranes, as described for other alphaviruses [5, 9-11].

In vitro activity studies on replication complexes from a variety of positive-stranded RNA (+RNA) viruses, like poliovirus [12] hepatitis C virus [13-16], West Nile virus [17, 18], Dengue virus [19], and SARS-coronavirus [20], have contributed to our understanding of +RNA virus replication. For the alphaviruses Semliki forest virus (SFV) and Sindbis virus (SINV) such *in vitro* assays have been developed as well [21].

We have now established an *in vitro* assay to characterize the activity and composition of CHIKV replication/transcription complexes (RTCs) isolated from infected cells. Besides developing a useful tool to screen for inhibitors of CHIKV RNA synthesis, our study provided more insight into the molecular details of CHIKV RNA synthesis and the mode of action of several inhibitors. Our *in vitro* studies also revealed the synthesis of a previously unrecognized 7.5 kb CHIKV RNA (RNA II) that is collinear with the 5' end of the genome up to the start of the sgRNA promoter region. This RNA species was also found in CHIKV-infected cells and resembles the RNA II that has been observed in SINV-infected cells [22].

Results

Isolation of active RTCs from CHIKV-infected cells

We set out to isolate CHIKV RTCs with the highest possible RNA-synthesizing activity from infected Vero E6 cells. We therefore first analyzed CHIKV RNA synthesis *in vivo* by metabolic labeling with ³H-uridine at various time points post infection. Quantification of ³H-labeled CHIKV RNA following denaturing agarose gel electrophoresis and fluorography (Fig. 1) showed that the rate of RNA synthesis increased up to 8 h p.i., after which a modest

decline was observed. Based on these results, we begun the isolation procedure of CHIKV RTCs at 6 h p.i. to ensure that a good amount of activity would remain after the relatively lengthy procedure (1-2 h).



Figure 1. Kinetics of CHIKV RNA synthesis in Vero E6 cells. (a) CHIKV-infected cells were pulse labeled for 1 h with ³H-uridine starting 30 min before the time points indicated above the gel, followed by isolation of RNA, denaturing agarose gel electrophoresis and visualization by fluorography. The positions of genomic and sgRNA (determined by hybridization of CHIKV RNA with a specific probe) and an additional CHIKV-specific ~7.5 kb product (RNA II) are indicated. (b) The rate of CHIKV RNA synthesis (replication and transcription combined) at each time point post infection was quantified by densitometry of the bands shown in panel a (circles) or by direct liquid scintillation counting of isolated RNA (squares). Ribosomal 18S RNA levels were used to correct for variations in isolation. The graph shows ³H-uridine incorporation (relative to the highest observed value) and represents the average ±SD of two experiments.

Approximately 1 x 10⁸ CHIKV-infected Vero E6 cells were harvested by trypsinization at 6 h p.i., followed by homogenization in hypotonic buffer. Nuclei and debris were pelleted and the resulting post-nuclear supernatant (PNS) was fractionated by centrifugation yielding a 15,000 x g pellet (P15) and supernatant (S15; Fig. 2a). The PNS, P15 and S15 fractions were assayed for CHIKV RNA-synthesizing activity by measuring the incorporation of ³²P-CTP using the protocol described in the Materials section.

Reactions performed with PNS prepared from CHIKV-infected cells yielded two major ³²P-labeled reaction products with sizes corresponding the CHIKV genome and sgRNA (Fig. 2b). RNA II, a minor product running between genome and sgRNA will be discussed below. No radiolabeled RNA was detected when the assay was performed using PNS prepared from mock-infected cells (Fig. 2b). Approximately 58% of the RTC activity present in the PNS could be recovered in the P15 fraction, while 6% remained in the S15 fraction. Combining the P15 and S15 fractions yielded an activity that was comparable to the sum of their individual activities, suggesting that the RTCs in the P15 fraction did not require (host) factors from the S15 fraction for their activity. The ~35% overall activity loss compared to the PNS, was likely due to damage caused by RTC pelleting and resuspending.



Figure 2. Isolation of active CHIKV RTCs from infected Vero E6 cells. (a) Overview of the isolation procedure. (b) *In vitro* incorporation of ³²P-CTP into RNA in PNS prepared from mock-infected cells, or PNS, P15 and S15 fractions obtained from CHIKV-infected cells. RNA was extracted, separated in denaturing agarose gels and visualized using a phosphor imager. The amount of ³²P-CTP incorporated into CHIKV genome and sgRNA is indicated under each lane, expressed as % of the radioactivity incorporated by the PNS fraction.

CHIKV RTCs synthesize RNA II in addition to genome and sgRNA

During the analysis of *in vitro* synthesized RNA we noticed that besides products with the size of genome and sgRNA, a ~7.5-kb product was synthesized, which we termed RNA II (Fig. 2b and 3a). Approximately 4% of the incorporated ³²P-CTP was present in RNA II, which was only observed under reaction conditions that allowed sgRNA synthesis. RNA II was not an artifact of the *in vitro* system as it was also detected in RNA isolated from CHIKV-infected cells metabolically labeled with ³H-uridine (see Fig. 1a and 3a). At the peak of RNA synthesis in vivo, about 2% of the 3H-label was found in RNA II, while 27% and 71% of the radioactivity was found in genome and sgRNA, respectively. Based on its size and the correlation between RNA II and sgRNA synthesis, we hypothesized it represents the 5'-proximal ~7.5 kb of the genome, up to the sgRNA promoter region (starting at nt 7480). We have termed this molecule RNA II, since this name was used for a similar RNA product described for SINV [22]. To map the 3' end of CHIKV RNA II, RNA from CHIKV-infected cells was hybridized with probes that bind to +RNA between nucleotides 4572-4601 (IA1), or immediately upstream (IA2) or ~100 nucleotides downstream (IA3) of the subgenomic promoter region (Fig. 3c). As anticipated, probes IA1 and IA2 hybridized to the genome and RNA II, while probe IA3 recognized sgRNA but not RNA II (Fig. 3b and c). This result confirmed our hypothesis that RNA II corresponds to the 5'-proximal 7.5 kb of the CHIKV genome up to subgenomic promoter region.



Figure 3. Analysis of the origin of CHIKV RNA II. (a) CHIKV RNA was metabolically labeled with ³H-uridine *in vivo* or with ³²P-CTP *in vitro*. The positions of the genome, sgRNA and the minor ~7.5 kb CHIKV-specific product RNA II are indicated. (b) Hybridization analysis of CHIKV RNA using probes that specifically bind the regions of CHIKV+RNA indicated at (c). (c) Schematic representation of the 3 +RNAs synthesized by CHIKV and the locations of the probes IA1, IA2 and IA3 that were used to map the origin of RNA II.

Optimization of reaction conditions for in vitro CHIKV RNA synthesis

A number of assay parameters were tested to determine the optimal conditions for CHIKV RNA synthesis in vitro. Activity remained relatively constant over a range of temperatures from 20 to 37°C (Fig. 4a), probably reflecting that CHIKV replicates at low temperatures in mosquito vectors and at higher temperatures in vertebrate hosts. Synthesis of sgRNA decreased at higher temperatures, while the ³²P-incorporation into genomic RNA increased. In further experiments, 30°C was used as the standard assay temperature, as this allowed the analysis of both genome replication and sgRNA synthesis. Magnesium was required for RTC activity and a concentration of 3 mM was optimal (Fig. 4b). An ATP-regenerating system was essential as no radiolabeled products were observed when creatine phosphate and creatine phosphokinase were omitted (data not shown). The incorporation of ³²P-CTP into CHIKV RNA was readily detectable after a 5-min incubation at 30°C, and the signal rapidly increased up to 90 min, after which it hardly increased further (Fig. 4c and e). Addition of pyrophosphatase did not increase the yield suggesting that the decreasing activity was not due to inhibition by the pyrophosphate that is released during NTP incorporation (data not shown). The addition of fresh NTPs, creatine phosphate and creatine phosphokinase after a 90-min reaction, followed by an additional 60-min incubation led to an almost 3-fold increase in sgRNA labeling compared to a reaction that was incubated for an additional 60 min without replenishing components (Fig. 4d). The incorporation of radioactivity into genomic RNA was only 1.2-fold higher. This suggests that the reaction rate dropped after 90 min due to exhaustion of one or more reaction components, like NTPs, and not due to RTC instability. Indeed, RTCs retained most of their activity when they were first kept at 30°C for 1 h before a reaction was started (data not shown). To assess the stability of the in vitro-synthesized RNAs, a reaction was terminated after 90 min by adding the obligate chain terminator 3'dUTP, after which RNA samples were taken every 20 min during a 60min chase period. In untreated control samples the amount of radioactive CHIKV RNA increased ~1.2 fold over the chase period (Fig. 4f, squares). The level of radioactive CHIKV RNA in samples in which RNA synthesis was blocked slowly decreased (half-life >60 min) over the chase period (Fig. 4f, circles). The half-life of a (naked) control RNA was less than 5 min (Fig. 4f, triangles), suggesting that the newly synthesized CHIKV RNA was somehow protected from cellular nucleases.

While optimizing the NTP concentration, we discovered that even in a reaction to which no CTP, GTP and UTP was added, a substantial amount of mainly genomic RNA was synthesized (Fig. 3g). This was likely due to the pool of endogenous NTPs in PNS [23]. Addition of 10 μ M of NTPs to the reaction hardly had an effect, probably because the endogenous NTP concentration was already at least 10-fold higher. Addition of 1 mM of ATP increased the reaction rate and resulted especially in increased sgRNA transcription and generation of RNA II. Reactions performed with the P15 fraction (not expected to contain endogenous NTPs) were strongly dependent on the supplied NTPs, as no ³²P-CTP incorporation was observed in the absence of NTPs (Fig. 3g). A final concentration of 10 μ M CTP, UTP and GTP and 1 mM of ATP was optimal. Higher NTP concentrations did not substantially increase the activity (not shown).



Figure 4. Optimization of reaction conditions for the *in vitro* RNA-synthesizing activity of CHIKV RTCs. The temperature (a), Mg²⁺ concentration (b) or incubation time (c) were varied and ³²P-CTP-labeled RNA products were analyzed (see Fig. 1). The percentages depicted under the lanes indicate ³²P incorporation normalized to the highest observed activity in a given series (100%), after correction for variations in RNA recovery based on quantification of 18S rRNA (not shown). (d) After a 90-min standard reaction (left lane) fresh NTPs, creatine phosphate and creatine phosphokinase (+) or an equal volume of dilution buffer (-) were added and the reaction was continued for an additional hour. (e) Kinetics of the incorporation of ³²P-CTP into CHIKV genomic and sgRNA. (e) Stability of newly synthesized CHIKV RNA in PNS. The incorporation of ³²P-CTP into CHIKV RNA was allowed to proceed for 90 min, after which it was blocked by the addition of 0.1 mM 3'dUTP, followed by the quantification of the remaining radioactive CHIKV RNA at the indicated time points of the chase (circles). A reaction that was chased in the absence of 3'dUTP was included as a control (squares). The triangles show the decay of a *in vitro* transcribed control RNA that was incubated with PNS. (f) Assays performed with either PNS or the P15 fraction in the presence of various concentrations of (added) ATP and the other NTPs.

CHIKV RTCs incorporate ³²P-CTP into single-stranded +RNA

We next assessed whether *in vitro* synthesized radioactive CHIKV RNAs resulted from genuine RdRp-mediated incorporation of ³²P-label or whether they were merely the result of end-labeling of existing RNA molecules by a viral or host cell activity. This was done by hybridizing a small DNA probe at a specific position in the viral RNA, ~2.2 kb from the 3' end of both genome and sgRNA, and subsequently cleaving the DNA:RNA duplex using RNase H (Fig. 5a). The fact that, after this targeted cleavage, the ~9.6-kb 5'-terminal genome fragment was radioactively labeled confirmed that ³²P-CTP had been incorporated at other positions than merely the 3' end of the CHIKV RNAs (Fig. 5b). A radioactive fragment corresponding to the predicted ~2.1 kb 5' fragment of the sgRNA was also observed, although migrating very close to the ~2.2 kb 3'-terminal fragment. Treatment of ³²P-labeled reaction products with single strand-specific RNAse A/T1 resulted in the degradation of more than 90% of the *in vitro* synthesized RNA, along with the 18S RNA that was used as an internal control (Fig. 5c). Treatment with double-stranded RNA-specific RNase III led to a 10-20% decrease of radioactive CHIKV RNA, suggesting that radioactivity had mainly accumulated in single-stranded RNA.

To determine the polarity of the newly synthesized RNA, radioactive reaction products were hybridized to a membrane containing specific capture probes for positive- and negative-stranded CHIKV RNA and a non-specific control RNA. Figure 5d shows that mainly +RNA was synthesized *in vitro*.



Figure 5. Characterization of *in vitro* synthesized CHIKV RNAs. (a) Schematic representation of the two possible mechanisms that could lead to generation of ³²P-labeled CHIKV RNAs and the targeted RNA cleavage assay with DNA probe and RNase H that was performed to distinguish between the two. Genuine RdRp activity would incorporate ³²P-CTP throughout newly synthesized RNA and would result in 4 radioactive fragments (sizes in bold italic) after cleavage. If RNAs would only be labeled at their 3' end, due to terminal transferase activity, only the 2.2 kb 3'-terminal fragment of genome and sgRNA would be radioactively labeled. (b) Result of the targeted cleavage assay using reaction products from an *in vitro* RTC assay. RNA II was not cleaved since it was not recognized by the DNA probe. (c) Treatment of *in vitro* synthesized CHIKV RNA with single strand-specific RNase A/T1 or the double-stranded RNA-specific RNase III, followed by denaturing agarose gel electrophoresis and direct detection of ³²P-labeled products by phosphor imaging. Ribosomal 18S RNA, was included as a control. (d) Binding of radioactive *in vitro* synthesized CHIKV RNA of positive or negative polarity or a non-specific control probe.

Distribution of CHIKV RNA and proteins between the P15 and S15 fraction

To obtain more insight into the composition of the P15 fraction that contains most of the RTC activity (Fig. 2b), the distribution of CHIKV RNA, viral proteins and several cellular marker proteins between P15 and S15 was studied.

Approximately 90% of -RNA was found in the P15 fraction, where it likely serves as template for +RNA synthesis (Fig. 6a). In contrast, the P15 fraction contained about 6 times less +RNA than the cytoplasmic S15 fraction (Fig. 6a).

The P15 fraction contained 20% or 16% of the total amount of nsP2 and nsP3, respectively. In contrast, 30% of nsP4 and 50% of the nsP1 were found in P15 (Fig. 6b). The E2 envelope protein was enriched in the membrane fraction (P15), which also contained about 15% of the capsid protein (Fig 6b). The absence of the nuclear marker fibrillarin indicated that the P15 and S15 fraction (and PNS) were not notably contaminated with nuclear material. The P15 fraction did not contain detectable amounts of the cytosolic marker cyclophilin A or actin, while it contained most of the endoplasmic reticulum marker cyclophilin B and the bulk of the plasma membrane marker Na⁺/K⁺ ATPase (Fig. 6b, right panel). The early endosome marker, Rab5, was predominantly found in the S15 fraction, whereas Rab7, a late endosome marker, was present both in P15 and S15. These results indicated that P15 was a rather crude fraction that contained membranes of various cellular origins.



Figure 6. Distribution of CHIKV RNA and viral and cellular marker proteins between the P15 and S15 fractions. (a). Total RNA was isolated from PNS, P15 and S15 fractions (volumes reflecting equal cell numbers) obtained from CHIKV-infected Vero E6 cells. After denaturing agarose gel electrophoresis, -RNA and +RNA were specifically detected by hybridization with ³²P-labeled DNA probes. Samples were spiked with a control RNA to correct for variations in the recovery and/or loading of RNA. (b) Western blot analysis of samples reflecting equal cell numbers of PNS, P15, and S15 of CHIKV-infected cells or a whole cell lysate of uninfected cells using antibodies specific for the protein indicated next to each panel.

The in vitro RTC assay as a tool for mode of action studies on inhibitors of CHIKV replication To determine whether our *in vitro* assay for CHIKV RNA synthesis is a suitable tool for mode of action studies on inhibitors of CHIKV replication, we have studied the effect of a variety of compounds. We first tested a set of compounds that inhibit CHIKV replication in cell culture [24], but were not expected to directly inhibit RNA synthesis. These compounds either affect another step in the CHIKV replication cycle (chloroquine, 3-deaza-adenosine), act as antimetabolite (mycophenolic acid, ribavirin, 6-aza-uridine) and/or need to be first converted by cellular enzymes to their active form (ribavirin, 6-aza-uridine). Indeed, none of these compounds had a measurable inhibitory effect in the *in vitro* assay for CHIKV RNA synthesis (Fig. 7a).

A set of nucleoside analogs that act as obligate chain terminators strongly inhibited the synthesis of CHIKV RNA in the *in vitro* assay in a dose-dependent manner (Fig. 7b). The weaker inhibitory effect of 3'dATP compared to that of the other 3'dNTPs was likely due to an excess of ATP in the reaction (1 mM supplied and also produced by the ATP-regenerating system). In line with this notion, the inhibitory effect of 3'dUTP, which was already very strong at a concentration of 10 μ M (Fig 7b, lane 7) could be reversed by adding an excess of UTP. For example a 20-fold molar excess of UTP over 50 μ M 3'dUTP restored the RNA-synthesizing activity to a level comparable to that observed in the presence of 1 μ M of the inhibitor (Fig. 7b, compare lanes 6 and 11). Finally, we analyzed whether 500 μ M of ribavirin triphosphate or 100 μ M 6-aza-UTP had an effect on the kinetics of CHIKV RNA synthesis *in vitro*. Figure 7c shows that the kinetics of RNA synthesis in the presence of these compounds was indistinguishable from that of the control reaction, suggesting these compounds do not (directly) affect the reaction rate.



Figure 7. Effect of various compounds on the *in vitro* RNA synthesizing activity of CHIKV RTCs. (a) Standard assays were performed in the presence of 0.5% DMSO, 50 μ M 6-azauridine, 100 μ M chloroquine, 60 μ M 3'-deaza-adenosine, 150 μ M ribavirin, 500 μ M fluorouracil, or 6 μ M mycophenolic acid. (b) Standard reactions performed in the presence of the obligate chain terminators 3'dATP, 3'dGTP and 3'dUTP at the concentrations indicated above the lanes. The rightmost lane contains RNA synthesized in the presence of 50 μ M 3'dUTP and 1 mM UTP. (c) Reaction kinetics of *in vitro* RNA synthesis in the presence of ribavirin triphosphate or 6-azaUTP or in the absence of inhibitor (control) analyzed over a time course of 5-90 min. (d) Quantified *in vitro* ³²P-CTP incorporation into CHIKV RNA in the presence of ribavirin triphosphate (squares) or 6-aza-UTP (triangles) represented as % of the maximum incorporation in the control reaction (circles).

Discussion

We have successfully developed an *in vitro* system to study CHIKV RNA synthesis, in which ³²P-CTP is incorporated into CHIKV genome and sgRNA. Mainly single-stranded +RNA accumulated, which was not unexpected considering lysates were prepared at 6 h p.i., when -RNA synthesis has ceased and RTCs are exclusively involved in +RNA synthesis [24]. A crude membrane fraction (P15) had an approximately 10-fold higher RNA synthesizing activity than the cytosolic S15 fraction. It remains to be determined whether the low activity in S15 represents a pool of structurally and functionally distinct RTCs or whether the activity is merely due to RTCs that were released (dissociated) from the pool of P15-associated complexes. We are currently developing more elaborate subcellular fractionation procedures to study these possibilities in detail. Our fractionation data showed that the pelleted CHIKV RTCs were not dependent on cytosolic host factors for their activity, in contrast to what was found for nidoviruses [20, 25]. This is in line with previous studies on SINV and SFV [21].

The P15 fraction contained most of the -RNA, the template for +RNA synthesis. The bulk of the +RNA was found in the S15 fraction, suggesting it was released from the membraneassociated RTCs. The *in vitro* synthesized CHIKV RNA was relatively stable in the lysate and appeared to be somehow protected from cellular nucleases. This might be due to its structure, membrane association, presence within polysomes, or encapsidation soon after its synthesis (capsid protein was also detected in P15). The S15 fraction contained about 80% of nsP2 and nsP3, proteins that besides their role in the RTC are known to have several other functions, like inducing shut-off of host transcription and translation and interacting with a variety of host proteins. Compared to nsP2 and nsP3, the RdRp nsP4 and especially nsP1 were enriched in the P15 fraction. The latter protein is presumably involved in the anchoring of RTCs to membranes.

We discovered that a previously unrecognized CHIKV RNA, which we termed RNA II, was produced besides genome and sgRNA. RNA II was synthesized both *in vitro* and *in vivo*, and we demonstrated that it represents the first 7.5 kb of the genome.

Earlier metabolic labeling studies already suggested the presence of similar molecules in SINV- and SFV-infected cells [26, 27] and Wielgosz *et al.* [22] have identified and mapped RNA II in SINV-infected cells [22]. RNA II was then believed to be unique for SINV, but we have now also identified it in CHIKV-infected cells. RNA II probably results from premature termination (near the sgRNA promoter) of RNAs initiated on the 3' end of a -RNA template that is also involved in transcription. This is supported by the observation that RNA II was only found under conditions that favored sgRNA synthesis. We are currently investigating whether RNA II is capped and polyadenylated and whether it is merely a by-product of genome replication or has a function in the infected cell.

We have tested several compounds in the in vitro assay for CHIKV RNA synthesis to demonstrate it is a useful addition to the toolbox for mode of action studies. As expected chloroquine showed no effect, because in vivo it blocks CHIKV entry. Ribavirin and 6-azauridine had no effect as their conversion to an active (triphosphorylated) form by cellular enzymes probably does not occur (to a sufficient level) in vitro. These compounds may also act as antimetabolites that in vivo affect the cellular NTP pool, like 5-fluorouracil and mycophenolic acid, and hence have no effect in the *in vitro* assay, as NTPs are supplied in the reaction buffer. In cell culture 3-deaza-adenosine inhibits CHIKV replication and it was hypothesized that this might be due to its effect on the capping of CHIKV RNA [24]. In line with this, we found that 3-deaza-adenosine had no direct effect on RNA synthesis in vitro. The obligate chain terminators 3'dATP, 3'dGTP and 3'dUTP inhibited CHIKV RNA synthesis in vitro at low micromolar concentrations. Their inhibitory effect could be reversed by addition of NTPs, which stresses the importance of not supplying a large excess of NTPs in *in vitro* assays. Otherwise the inhibitory effect of (novel) compounds might be underestimated or missed completely. Using the newly developed assay, we found that ribavirin triphosphate and 6-aza-UTP had no measurable effect on the kinetics of CHIKV RNA synthesis. This might indicate that either these compounds were not incorporated by the RTC or they were incorporated without affecting the reaction rate. In addition, these compounds could also indirectly affect RNA synthesis in vivo through their effect on cellular NTP pools. Obviously, this would have no effect in the *in vitro* system as NTPs are supplied. This is further supported by the fact that the IMPDH inhibitor mycophenolic acid had no effect in vitro, while it inhibited CHIKV in cell culture. Since ribavirin and 6-azauridine increase the viral mutation frequency [28, 29], at least part of their mode of action can probably be attributed to their incorporation into viral RNA.

The in vitro assay described here may be used to enhance our understanding of the molecular details of CHIKV RNA synthesis, to screen for inhibitors, and to study the mode of action of antiviral compounds originating from other (cell-based) screens.

Methods

Cells, virus, antisera and compounds

Vero E6 cells were infected with CHIKV strain LS3 at an MOI of 5 as described [24]. Antisera against CHIKV nsP1, nsP2, nsP3, nsP4, and capsid protein were a generous gift from prof. Andres Merits (Tartu, Estonia). The E2 antiserum [30] was obtained from dr. Gorben Pijlman (Wageningen, The Netherlands). Antisera against cellular markers were purchased from Sigma (actin), Santa Cruz (fibrillarin, cyclophilins A&B), and Cell Signalling (Rab5, Rab7 and Na/K ATPase). Ribavirin triphosphate, 3'dATP, 3'dGTP and 3'dUTP were from TriLink BioTechnologies and 6-aza-UTP from Jena Bioscience. Stock solutions of ribavirin, 6-azauridine, 5-fluorouracil, 3'-deaza-adenosine, chloroquine and mycophenolic acid were prepared as described [24].

RNA isolation, denaturing agarose gel electrophoresis, in gel hybridization and detection of ³²P- and ³H-labeled RNA

RNA isolation [25] and denaturing formaldehyde agarose gel electrophoresis were performed as described [24]. For detection of ³²P-labeled RNA Phosphor Imager screens were directly exposed to dried gels, followed by scanning with a Typhoon 9410 imager (GE Healthcare). CHIKV -RNA and +RNA were specifically detected by in-gel hybridization with ³²P-labeled probe Hyb2 and Hyb4 [24]. Probes IA1 (5'-CAATATCGCAGTCTA-TGGAGATGTGCTCAT-3'), IA2 (5'-GTTATGACG-GGTCCTCTGAGCTTCTCGA-3') and IA3 (5'- CCTCCTATT-GTAAAAAGTTT-GGGTTGGG-ATG-3') are complementary to nt 4572-4601, 7452-7479, and 7575-7605 of the genome, resp. The 18S ribosomal RNA, detected with probe 5'-ATGCCCCGGCCGTCCCTCT-3', was used as isolation efficiency and loading control for RNA isolated from cells or PNS. P15 and S15 fractions were spiked with an *in vitro* transcribed RNA containing the 3' end of the CHIKV genome that was detected with probe Hyb4, to correct for variations in isolation and/or loading.

Metabolic labeling of CHIKV-infected cells

Metabolic labeling of CHIKV-infected cells with 40 μ Ci ³H-uridine was done basically as described [24] except that we used Vero E6 cells seeded in 12-well clusters at a density of $2x10^5$ cells/well and infected at an MOI of 5.

Isolation of RTCs from CHIKV-infected cells

Approximately 1 x 10⁸ CHIKV-infected VeroE6 cells were harvested by trypsinization at 6 h p.i. Cells were resuspended in 4 ml of hypotonic buffer containing 20 mM HEPES pH 7.5, 10 mM KCl, 1 mM DTT, 0.2 U/ μ l RiboLock (Thermo Scientific), and 2 μ g/ml ActD. After incubation at 4°C for 15 min, the cells were disrupted using a Dounce homogenizer. To increase the osmotic value sucrose was added to a final concentration of 250 mM. HEPES and DTT were also added to a final concentration of 35 mM and 2.5 mM, respectively. The lysate was cleared by centrifugation at 1,000 x g for 10 min to remove unlysed cells, nuclei and cellular debris, yielding a post-nuclear supernatant (PNS). Part of the PNS was further separated by centrifugation at 15,000 x g for 15 min into a heavy membrane pellet fraction (P15) and a supernatant fraction S15. The P15 pellet was resuspended in dilution buffer (35 mM HEPES pH 7.5, 250 mM Sucrose, 2.7 mM DTT, 7 mM KCl, 2 μ g/ml ActD, and 0.2 U/ μ l RiboLock) using 1/5 of the volume of the PNS from which the pellet originated (now S15). This 5x concentrated P15 fraction and all other fractions were aliquoted and stored at -80°C.

In vitro RNA synthesis assay

These assays were performed inside the BSL-3 facility, since infectious CHIKV remained present in the lysates and fractions used. Standard 30- μ l reactions contained either 25 μ l PNS, or 25 μ l S15, or 5 μ l P15 (5x concentrated) mixed with 20 μ l dilution buffer, 30 mM

HEPES (pH 7.5), 220 mM sucrose, 7 mM KCl, 2.5 mM DTT, 3 mM magnesium acetate, 2 μ g/ ml ActD, 0.2 U/µl of RiboLock, 20 mM creatine phosphate, 10 U/ml creatine phosphokinase, 10 μ Ci (0.12 μ M) of [α -³²P]-CTP (Perkin Elmer), 1 mM ATP, and either 0.01 or 0.2 mM of GTP, UTP, and CTP. Unless otherwise stated, reactions were performed for 60 min at 30°C and terminated by the addition of 60 μ l of 5% lithium dodecyl sulfate, 0.1 M Tris-HCl pH 8.0, 0.5 M LiCl, 10 mM EDTA, 5 mM DTT, and 0.1 mg/ml proteinase K. After a 15-min incubation at 42 °C, unincorporated label was removed using RNase-free Micro Bio-spin 30 columns (Bio-Rad), and total RNA was isolated and analyzed as described above.

Detection of ³²P-labeled reaction products with capture probes

In vitro transcribed RNAs (1 μ g) representing the first 1348 nucleotides of the CHIKV negative strand, the 1100 3'-terminal nt of the CHIKV genome, or nt 1-2042 of the unrelated equine arteritis virus (negative control), were immobilized to Hybond N+ membrane (GE Healthcare) as described [20]. The membrane with the immobilized probes was incubated with half of the ³²P-labeled RNA isolated from a 30- μ l *in vitro* reaction and washing and detection of bound material were done as described [20]. ³²P-labeled *in vitro* transcript that contained 5' and 3' UTR of CHIKV or a transcript with the complementary sequence were used as positive controls and to assess the specificity of the immobilized probes.

RNase treatments

RNA isolated from *in vitro* RTC reactions was treated with a RNase A/T1 mixture under high salt conditions to degrade single-stranded RNA, or with RNase III to degrade double-stranded RNA as described [25].

Targeted RNA cleavage assay

For targeted cleavage, RNA isolated from *in vitro* reactions was heated to 96°C for 4 min, followed by hybridization overnight at 45°C in 20 μ l of 40 mM PIPES pH 6.5, 0.4 M NaCl, 1mM EDTA, and 80% deionized formamide with DNA probe AS25 (5'-GATAACTGCGGCCAATACTTAT-3') that is complementary to nt 9548 – 9569 of the CHIKV genome. RNase H (7.5 U; Fermentas) in a volume of 150 μ l 10 mM PIPES pH 6.5, 0.1 M NaCl, 0.25 mM EDTA, and 10 mM DTT was added to the hybridized RNA. After an incubation for 60 min at 37°C, 225 μ l of RNase inactivation solution (Ambion RPA kit) was added, followed by incubation for 30 min at -20°C. This treatment results in the targeted cleavage of a 2,242 nucleotide fragment of the 3' end of genomic and sgRNA, The digested RNA was precipitated in the presence of GlycoBlue (Ambion) at 4°C by centrifugation for 30 min at 15,000 x g. After resuspension in 5 μ l of 1 mM sodium citrate, RNA was analyzed by denaturing agarose gel electrophoresis.

SDS-PAGE and western blot analysis

SDS-PAGE and Western blotting using a Trans-Blot Turbo instrument and fluorescent detection of antibodies with a Typhoon 9410 scanner were done as described [24].

Acknowledgements

This work was supported by the European Union Seventh Framework Programme (FP7/2007–2013) under SILVER grant agreement nr. 260644 and the Marie Curie Initial Training Network EUVIRNA (grant agreement 264286). We thank prof. Andres Merits (Tartu, Estonia) and dr. Gorben Pijlman (Wageningen, The Netherlands) for generously sharing their CHIKV antisera.

References

- 1. Burt, F.J., et al., Chikungunya: a re-emerging virus. Lancet, 2012. 379(9816): p. 662-71.
- Gibney, K.B., et al., Chikungunya fever in the United States: a fifteen year review of cases. Clin Infect Dis, 2011. 52(5): p. e121-6.
- Rezza, G., et al., Infection with chikungunya virus in Italy: an outbreak in a temperate region. Lancet, 2007. 370(9602): p. 1840-6.
- 4. Grandadam, M., et al., Chikungunya virus, southeastern France. Emerg Infect Dis, 2011. 17(5): p. 910-3.
- Frolova, E.I., et al., Functional Sindbis virus replicative complexes are formed at the plasma membrane. J Virol, 2010. 84(22): p. 11679-95.
- Weaver, S.C., Arrival of chikungunya virus in the new world: prospects for spread and impact on public health. PLoS Negl Trop Dis, 2014. 8(6): p. e2921.
- 7. Leparc-Goffart, I., et al., Chikungunya in the Americas. Lancet, 2014. 383(9916): p. 514.
- Fischer, M., et al., Notes from the field: chikungunya virus spreads in the americas Caribbean and South america, 2013-2014. MMWR Morb Mortal Wkly Rep, 2014. 63(22): p. 500-1.
- Spuul, P., et al., Phosphatidylinositol 3-kinase-, actin-, and microtubule-dependent transport of Semliki Forest Virus replication complexes from the plasma membrane to modified lysosomes. J Virol, 2010. 84(15): p. 7543-57.
- 10. Froshauer, S., J. Kartenbeck, and A. Helenius, Alphavirus RNA replicase is located on the cytoplasmic surface of endosomes and lysosomes. J Cell Biol, 1988. **107**(6 Pt 1): p. 2075-86.
- 11. Kujala, P., et al., Biogenesis of the Semliki Forest virus RNA replication complex. J Virol, 2001. 75(8): p. 3873-84.
- 12. Baltimore, D., et al., Poliovirus-induced RNA polymerase and the effects of virus-specific inhibitors on its production. Proc Natl Acad Sci U S A, 1963. **49**: p. 843-9.
- 13. Lai, V.C., et al., In vitro RNA replication directed by replicase complexes isolated from the subgenomic replicon cells of hepatitis C virus. J Virol, 2003. 77(3): p. 2295-300.
- 14. Yang, W. and M. Huang, *Studying HCV RNA synthesis in vitro with replication complexes*. Methods Mol Biol, 2009. **510**: p. 177-84.
- 15. Hardy, R.W., et al., Hepatitis C virus RNA synthesis in a cell-free system isolated from replicon-containing hepatoma cells. J Virol, 2003. 77(3): p. 2029-37.
- Ali, N., K.D. Tardif, and A. Siddiqui, Cell-free replication of the hepatitis C virus subgenomic replicon. J Virol, 2002. 76(23): p. 12001-7.
- 17. Grun, J.B. and M.A. Brinton, Characterization of West Nile virus RNA-dependent RNA polymerase and cellular terminal adenylyl and uridylyl transferases in cell-free extracts. J Virol, 1986. **60**(3): p. 1113-24.
- Chu, P.W. and E.G. Westaway, Replication strategy of Kunjin virus: evidence for recycling role of replicative form RNA as template in semiconservative and asymmetric replication. Virology, 1985. 140(1): p. 68-79.
- You, S. and R. Padmanabhan, A novel in vitro replication system for Dengue virus. Initiation of RNA synthesis at the 3'-end of exogenous viral RNA templates requires 5'- and 3'-terminal complementary sequence motifs of the viral RNA. J Biol Chem, 1999. 274(47): p. 33714-22.
- van Hemert, M.J., et al., SARS-coronavirus replication/transcription complexes are membrane-protected and need a host factor for activity in vitro. PLoS Pathog, 2008. 4(5): p. e1000054.
- Barton, D.J., S.G. Sawicki, and D.L. Sawicki, Solubilization and immunoprecipitation of alphavirus replication complexes. J Virol, 1991. 65(3): p. 1496-506.
- 22. Wielgosz, M.M. and H.V. Huang, A novel viral RNA species in Sindbis virus-infected cells. J Virol, 1997. 71(12): p. 9108-17.
- 23. Traut, T.W., Physiological concentrations of purines and pyrimidines. Mol Cell Biochem, 1994. 140(1): p. 1-22.
- Scholte, F.E., et al., Characterization of synthetic Chikungunya viruses based on the consensus sequence of recent E1-226V isolates. PLoS One, 2013. 8(8): p. e71047.
- van Hemert, M.J., et al., The in vitro RNA synthesizing activity of the isolated arterivirus replication/ transcription complex is dependent on a host factor. J Biol Chem, 2008. 283(24): p. 16525-36.

- Levin, J.G. and R.M. Friedman, Analysis of arbovirus ribonucleic acid forms by polyacrylamide gel electrophoresis. J Virol, 1971. 7(4): p. 504-14.
- Bruton, C.J. and S.I. Kennedy, Semliki Forest virus intracellular RNA: properties of the multi-stranded RNA species and kinetics of positive and negative strand synthesis. J Gen Virol, 1975. 28(1): p. 111-27.

26.

- Crotty, S., C. Cameron, and R. Andino, *Ribavirin's antiviral mechanism of action: lethal mutagenesis?* J Mol Med (Berl), 2002. 80(2): p. 86-95.
- Beaucourt, S. and M. Vignuzzi, Ribavirin: a drug active against many viruses with multiple effects on virus replication and propagation. Molecular basis of ribavirin resistance. Curr Opin Virol, 2014. 8C: p. 10-15.
- Metz, S.W., et al., Effective chikungunya virus-like particle vaccine produced in insect cells. PLoS Negl Trop Dis, 2013. 7(3): p. e2124.

Chapter 3

Suramin inhibits chikungunya virus replication through multiple mechanisms

Irina C. Albulescu¹, Marcella van Hoolwerff¹, Laura Wolters¹, Elisabetta Bottaro², Claudio Nastruzzi², Shih Chi Yang³, Shwu-Chen Tsay³, Jih Ru Hwu³, Eric J. Snijder¹, and Martijn J. van Hemert^{1*}

¹Molecular Virology Laboratory, Department of Medical Microbiology, Leiden University Medical Center, Leiden, The Netherlands. ²Department of Life Sciences and Biotechnology, University of Ferrara, Italy. ³Department of Chemistry and Frontier Research Center on Fundamental and Applied Sciences of Matters, National Tsing Hua University, Hsinchu, Taiwan.

*Corresponding author.

Published in Antiviral Research, 2015 Sep; 121:39-46.

Abstract

Chikungunya virus (CHIKV) is a mosquito-borne alphavirus that causes severe and often persistent arthritis. In recent years, millions of people have been infected with this virus for which registered antivirals are still lacking. Using our recently established *in vitro* assay, we discovered that the approved anti-parasitic drug suramin inhibits CHIKV RNA synthesis (IC₅₀ of ~5 μ M). The compound inhibited replication of various CHIKV isolates in cell culture with an EC₅₀ of ~80 μ M (CC₅₀ >5 mM) and was also active against Sindbis virus and Semliki Forest virus. *In vitro* studies hinted that suramin interferes with (re)initiation of RNA synthesis, whereas time-of-addition studies suggested it to also interfere with a post-attachment early step in infection, possibly entry. CHIKV (nsP4) mutants resistant against favipiravir or ribavirin, which target the viral RNA polymerase, did not exhibit cross-resistance to suramin, suggesting a different mode of action. The assessment of the activity of a variety of suramin-related compounds in cell culture and the *in vitro* assay for RNA synthesis provided more insight into the moieties required for antiviral activity. The antiviral effect of suramin-containing liposomes was also analyzed. Its approved status makes it worthwhile to explore the use of suramin to prevent and/or treat CHIKV infections.

1. Introduction

Chikungunya virus (CHIKV) is a mosquito-borne arthritogenic alphavirus that has infected millions of people since its re-emergence in 2005. In November 2013, CHIKV emerged in the Caribbean [1, 2], starting an outbreak that has thus far resulted in over 1.2 million cases in the Americas (http://www.paho.org/hq/index.php?Itemid=40931).

CHIKV replication occurs in the cytoplasm on modified endosomal membranes and is driven by replication and transcription complexes (RTCs) that contain CHIKV nonstructural proteins (nsP) nsP1-4, of which nsP4 is the RNA-dependent RNA polymerase (RdRp). Early in infection negative-stranded RNA (RNA) complementary to the viral genome is synthesized, which serves as template for the production of genomic and subgenomic RNA (sgRNA). The genome serves as mRNA for the production of nsPs and the sgRNA is translated into the structural proteins that are required for the biogenesis of new virions.

Despite intensified research efforts over the past years and the identification of a variety of compounds with anti-CHIKV activity in preclinical studies [3], there are still no registered drugs on the market for treating CHIKV infections. Suramin is a symmetrical sulfonated naphthylurea compound that was approved for the treatment of parasitic infections in 1921, but its anti-cancer and antiviral potential were discovered only 60 years later (reviewed in [4-6]). It was shown that suramin had anti-reverse transcriptase activity against tumor-inducing viruses [7] and it was actually the first documented HIV reverse transcriptase inhibitor that was tested in human patients [8], but the compound's side effects outweighed the clinical benefit due to the required long term treatment [9]. A later study revealed that suramin's anti-HIV activity *in vivo* was actually due to its inhibitory effect on the interaction between the viral gp120 and the CD4 receptor [10]. Suramin has also been shown to block

the binding or early steps of infection of several DNA and RNA viruses, like herpes simplex virus type-1 [11], cytomegalovirus [12], human hepatitis B virus [13], hepatitis delta virus [14], hepatitis C virus [15], dengue virus [16], several bunyaviruses [17-20], norovirus-like particles [21] and enterovirus 71 [22], for which the antiviral activity of suramin was also confirmed in an animal model [23]. In recent *in vitro* studies suramin was identified as a hepatis C virus and dengue virus helicase inhibitor [24, 25] and also as a norovirus RdRp inhibitor by virtual screening and biochemical assays with purified enzymes [26, 27]. In the present study we assessed the effect of suramin on CHIKV RNA synthesis using our recently established *in vitro* assay that relies on RTCs isolated from infected cells [28]. We found that suramin inhibits both CHIKV RNA synthesis *in vitro* as well as an early step in CHIKV infection of cultured cells. In addition to describing the inhibition of CHIKV replication through two independent mechanisms, we provide more insight into the moieties required for suramin's antiviral activity.

2. Material and Methods

2.1. Cell lines, viruses and virus titration.

Vero E6 and BHK-21 cell culture and infectious clone-derived CHIKV LS3 and strain ITA07-RA1 have been described previously [29]. CHIKV STM35 is an infectious clone-derived virus based on the sequence of a clinical isolate from the island of St. Martin (manuscript in preparation). CHIKV M5 is a reverse-engineered LS3-derived (nsP4) mutant virus that is resistant to favipiravir [30] and CHIKV C483Y is identical to LS3 except for a C483Y mutation in nsP4 that renders it resistant to ribavirin [31]. Sindbis virus (SINV) strain HR and Semliki Forest virus (SFV) strain SFV4 were used. Virus titers were determined by plaque assay on Vero E6 cells as described [29]. All experiments with CHIKV were performed in a Leiden University Medical Center biosafety level 3 facility.

2.2. Compounds

Suramin was from Santa Cruz and Sigma and 3'dUTP from TriLink. Suramin-related compounds were synthesized at the National Tsing Hua University in Taiwan and their synthesis and spectroscopic data will be reported separately (manuscript in preparation). All compounds were dissolved in water. Suramin-containing liposomes were prepared as previously described [32].

2.3. Cytopathic effect (CPE) protection assay

CPE protection assays with Vero E6 cells and the CellTiter 96° Aqueous Non-Radioactive Cell Proliferation kit (Promega) were performed as described [29].

2.4. In vitro RNA synthesis assay

In vitro assays for viral RNA synthesis, based on the incorporation of ³²P-CTP into viral RNA, were performed as described [28] using RTCs isolated from VeroE6 cells infected with CHIKV LS3, SINV or SFV4 or BHK-21 cells transfected with CHIKV replicon RNA (see 2.6).

2.5. CHIKV protein and RNA analysis

RNA isolation from infected cells, denaturing agarose gel electrophoresis, detection of ³²P-RNA or viral RNA by hybridization with (strand-) specific probes have been described previously [28, 33]. CHIKV genome copy numbers were determined by internally-controlled TaqMan multiplex RT-qPCR as described [34]. Detection of CHIKV proteins by SDS-PAGE and Western blotting was done using procedures and antisera that were described previously [29, 34].

2.6. Transfection of cells with CHIKV replicon RNA

Freshly trypsinized BHK-21 cells were transfected by electroporation using 4 x 10⁶ cells in 0.4 mL PBS and 4 μ g of *in vitro* transcribed CHIKV replicon RNA [35] per 4 mm cuvette (Bio-Rad). After two pulses with an Eurogentec Easyjet Plus instrument set at 850 V and 25 μ F, cells were transferred to T-75 flasks with pre-warmed medium, followed by a 10-h incubation at 37 °C.

2.7 Statistical analysis

Graph-Pad Prism 5.01 was used for EC_{50} , IC_{50} and CC_{50} determination by non-linear regression and for statistical analysis performed with one-way ANOVA with Dunnett's (figure 2C) or Bonferroni's (figure 6B) multiple comparison test.

3. Results and discussion

3.1. Suramin inhibits RNA synthesis of CHIKV and other alphaviruses in vitro

As suramin was previously shown to inhibit the *in vitro* activity of a number of viral polymerases, including that of noroviruses [26, 27], we set out to study its effect on CHIKV RNA synthesis using our recently established *in vitro* assay that is based on the RNA-synthesizing activity of RTCs isolated from CHIKV-infected cells. This assay measures the incorporation of $[\alpha]^{32}$ P-CTP into viral RNA, which was severely impaired by suramin in a dose-dependent manner, with an IC₅₀ of approximately 5 μ M (Fig. 1A, supplemental figure S1A). Suramin also inhibited the *in vitro* activity of RTCs derived from SINV- (Fig. 1B) or SFV-infected cells (Fig. 1C), suggesting that it is a broad-spectrum inhibitor of alphavirus RNA synthesis. A small fraction of the RNA-synthesizing activity appeared refractory to the inhibitory effect of suramin, as some residual incorporation of ³²P-CTP remained even in the presence of 500 μ M of the compound.



Figure 1. Effect of suramin on alphavirus RNA synthesis in vitro.

In vitro RNA synthesis assays with RTCs isolated from Vero E6 cells infected with CHIKV (A), SINV (B) or SFV (C) were performed in the presence of the suramin concentrations indicated above the lanes. RNA was extracted and the ³²P-labeled reaction products were analyzed by denaturing agarose gel electrophoresis and phosphorimaging. A lysate from mock-infected cells was used as a negative control and 18S ribosomal RNA, detected by hybridization, was used as loading control.

3.2. Suramin inhibits the replication of CHIKV and other alphaviruses in cell culture

To determine the antiviral efficacy of suramin in cell culture, Vero E6 cells were infected with different CHIKV strains and treated with serial dilutions of the compound in a CPE protection assay. Viability assays on uninfected cells were performed in parallel to determine the CC_{50} . The EC_{50} values for infectious clone-derived CHIKV LS3, a natural isolate from Italy (ITA07-RA1) and a Caribbean CHIKV strain (STM35) were 75-80 μ M (Table 1, supplemental figure S2). The EC_{50} values are ~15 times higher than the IC_{50} values, maybe due to inefficient cellular uptake or poor availability of the compound. Suramin also inhibited the replication of SINV and SFV in cell culture (Table 1). The CC_{50} of suramin was higher than 5 mM, but the compound had a cytotoxic effect at high concentrations, as viability dropped to 65% at 5 mM, the highest concentration of suramin tested. This results in a selectivity index (SI) of >60 for CHIKV and SFV. For comparison, with the same CHIKV strain we found SI values of >32 for ribavirin >7 for T-705. In a plaque reduction assay, in which suramin was only present for 1 h during infection, the concentration that reduced the number of CHIKV plaques by 50% was determined to be 80 μ M (data not shown).

Table 1. Antiviral activity of suramin against various alphaviruses in cell culture. EC_{50} values were determined in CPE reduction assays and the average and standard deviation of 2 independent experiments, performed in quadruplicate are listed. (* in an independent experiment with higher concentrations suramin was found to be toxic, but the CC_{50} was still above 5 mM)

Virus	EC ₅₀ (μM)	CC ₅₀ (mM)
CHIKV LS3	79 ± 11.6	> 1*
CHIKV ITA07-RA1	76 ± 7	> 1*
CHIKV STM35	79 ± 12.9	> 1*
SINV	141 ± 18.3	> 0.4*
SFV	40 ± 10	> 0.4*

3.3. Suramin reduces CHIKV RNA and protein levels and infectious progeny titers

A dose-response experiment was performed to analyze the antiviral effect of suramin in a single replication cycle. CHIKV-infected Vero E6 cells were grown in the presence of various suramin concentrations (up to 500 μ M) from -6 to 12 h p.i., when they were lysed and analyzed. Suramin reduced the accumulation of nsP1 and capsid protein in a dose-dependent manner, to hardly detectable levels in cells treated with 500 μ M suramin (Fig. 2A). The accumulation of CHIKV RNA and positive-stranded RNA (+RNA) was also severely impaired at concentrations of 125 μ M suramin or higher (Fig. 2B). The production of infectious CHIKV was strongly inhibited, leading to a 4-log reduction when 500 μ M of the compound was present (Fig. 2C). The observed reduction of RNA levels (and consequently +RNA, nonstructural and structural proteins and infectious virus) in this single-cycle analysis suggests that suramin affects an early step in the CHIKV replication cycle.



Figure 2. Effect of suramin on CHIKV replication (A) Western blot analysis of nsP1 and capsid protein levels in CHIKV-infected Vero E6 cells (MOI 1) that were treated with suramin at the concentrations indicated above the lanes and analyzed at 12 h p.i. Actin was used as loading control. (B) CHIKV RNA and +RNA were detected in total RNA samples from CHIKV-infected cells treated with suramin at the concentrations indicated above the lanes and analyzed at 12 h p.i. by hybridization with specific probes. 18S ribosomal RNA, detected with a probe was used as loading control. (C) Infectious CHIKV titers at 20 h p.i. in the culture medium of cells treated with various concentrations of suramin were determined by plaque assay. The bars represent the average (\pm stdev) of two independent experiments with plaque assays performed in duplicate; ***p < 0.005.

3.4. Suramin also inhibits an early step of the CHIKV replication cycle

To determine which step of CHIKV replication is inhibited, we performed a time-ofaddition experiment in which cells were treated with 500 μ M suramin. Suramin was added at 30 or 10 minutes prior to infection or at 0, 5, 10, 20 or 30 minutes after infection, and remained present up to 60 min p.i., when the inoculum was removed, cells were washed 5 times with warm PBS and incubated in medium without suramin (Fig. 3A). In addition, cells were infected in the absence of suramin, and then treated with 500 μ M suramin from 1 - 7 h p.i. (Fig. 3A, sample 8). At 7 h p.i. cells were lysed and CHIKV replication was assessed by analyzing CHIKV –RNA levels (Fig. 3A). When suramin was added very early, not later than 20 minutes p.i., it strongly reduced CHIKV replication, as indicated by the ~85 -75% reduction of RNA levels compared to those in untreated infected cells. Addition of suramin later than 30 min p.i., and even treatment from 1 - 7 h p.i., was much less effective, leading to a reduction in RNA levels of only ~20% (Fig. 3A, samples 7 & 8). These results suggest that -besides its effect on RNA synthesis- suramin also inhibits an early step of the CHIKV replicative cycle, possibly attachment or entry.



Figure 3. Effect of suramin on early steps of the CHIKV replication cycle (A) Vero E6 cells were infected with CHIKV (MOI 5) and were left untreated (Ctrl) or were treated with 500 μ M suramin during the intervals schematically indicated for each sample. At 60 min p.i., cells were washed extensively and incubated in medium without suramin (sample 1-7) or with 500 μ M suramin (sample 8) for an additional 6 h. At 7 h p.i. CHIKV –RNA levels were determined by hybridization with a specific probe and the normalized quantities (% of untreated control) corrected for variations in loading, based on the 18S rRNA signal, are indicated under the lanes. (B) 10⁵ PFU of CHIKV were incubated for 30 min in medium without (Ctrl) or with 500 μ M suramin or with 70% ethanol, followed by determination of the infectious virus titer by plaque assay. (C) CHIKV (MOI 5) was allowed to bind for 30 min at 4 °C to confluent monolayers of Vero E6 cells in 12-well clusters in the presence of various high concentrations of suramin. After extensive washing with ice-cold PBS, the amount of bound CHIKV was determined by measuring the number of genome copies per well using an internally controlled multiplex RT-qPCR.

To test whether suramin has a negative (virucidal) effect on the infectivity of virions, 10^5 PFU of CHIKV were incubated with 500 µM suramin or 70% ethanol (positive control for virucidal activity) for 30 min and the remaining infectivity was analyzed by plaque assay. Compared to the untreated control (Ctrl), suramin treatment did not reduce the infectious titer, while ethanol completely abolished infectivity (Fig. 3B). This demonstrated that suramin is not virucidal and has no irreversible negative effects on the virus. To assess whether suramin blocks attachment of CHIKV to cells, Vero E6 cells were incubated with CHIKV (MOI 5) at 4°C (to block entry by endocytosis) for 30 min in medium with various concentrations of suramin, after which the cells were washed 5 times with ice-cold PBS. The amount of cell-associated virus was quantified by RT-qPCR analysis of total RNA collected immediately after the last washing step (Fig. 3C). Suramin treatment did not affect the amount of bound virus and therefore does not appear to interfere with CHIKV attachment, but likely interferes with a later, post-attachment step such as entry, fusion of the viral envelope with the endosomal membrane and/or the release of nucleocapsids into the cytoplasm.

3.5. Suramin also inhibits CHIKV RNA synthesis in cell culture

To assess whether suramin also inhibits CHIKV RNA synthesis in infected cell culture, we analyzed the kinetics of the accumulation of CHIKV genomic RNA following treatment with various high doses of suramin added 1 h after infection. Figure 4 shows that post-infection treatment changed the kinetics of RNA synthesis in cell culture, leading to ~4-fold reduction in the number of CHIKV genome copies per cell at 7 h p.i.

To validate the effect of suramin on RNA synthesis in cell culture, independent of its effect on entry, we electroporated BHK-21 cells with CHIKV replicon RNA and treated these cells with different concentrations of suramin. Suramin inhibited RNA synthesis and expression of the eGFP reporter gene, which is dependent on transcription of the sgRNA (data not shown), suggesting that besides its effect on entry, suramin also inhibits RNA synthesis in cell culture. However, the latter effect appeared to be much weaker compared to the impact on the early step (entry), which might be explained by poor uptake or intracellular availability of suramin.



Figure 4. Effect of suramin on the kinetics of CHIKV RNA accumulation in cell culture.

Vero E6 cells were infected with CHIKV (MOI 3) and at 1 h p.i. the inoculum was removed, cells were washed extensively with warm PBS, followed by incubation in medium with 0, 0.5, 1 or 2 mM suramin. Intracellular RNA was isolated at 3, 5 and 7 h p.i. and the CHIKV genome copy numbers per cell were determined by RT-qPCR.

3.6. Mutations that confer resistance to favipiravir or ribavirin do not provide cross-resistance to suramin.

We determined the suramin sensitivity of nsP4 mutant CHIKV M5, which is resistant to favipiravir [30], and of CHIKV C483Y, which has a C483Y mutation in nsP4 rendering it resistant to ribavirin [31] (supplemental figure S3A). CHIKV M5 and C483Y exhibited an ~2-fold resistance to favipiravir (supplemental figure S3B) and ribavirin (supplemental figure S3C), respectively. In a CPE protection assay with suramin and using BHK-21 cells (to allow parallel experiments with ribavirin, which does not work in Vero E6 cells) EC₅₀ values of 27 and 48 µM were found for CHIKV M5 and CHIKV C483Y, respectively, compared to 41 μ M for the parental CHIKV LS3. In a CPE protection assay with Vero E6 cells EC50 values of 72 and 61 µM were found for CHIKV M5 and CHIKV C483Y, respectively, suggesting that these mutants are equally or even somewhat more sensitive to suramin than the parental CHIKV LS3 (EC₅₀ 79 µM). The effect of suramin in the CPE protection assay is likely mainly due to its inhibition of the early step of CHIKV replication. Therefore, we also analyzed more specifically the effect of suramin on the kinetics of CHIKV RNA accumulation for wt CHIKV LS3 and the favipiravir- and ribavirin-resistant mutants (Fig. 5). As for the wt virus, the RNA synthesis of both mutants was inhibited by suramin. The lack of cross-resistance suggests that suramin acts on RNA synthesis (RdRp) through a different mechanism.



Figure 5. Effect of suramin on the kinetics of CHIKV RNA accumulation of wt CHIKV and two mutants that are resistant to ribavirin and favipiravir. Vero E6 cells were infected with CHIKV LS3, CHIKV M5 or CHIKV C483Y at an MOI of 3 and at 1 h p.i. the inoculum was removed, cells were extensively washed with warm PBS, followed by incubation in medium with 0, 0.5, 1 or 2 mM suramin. Intracellular RNA was isolated at 3, 5 and 7 h p.i. and the CHIKV genome copy numbers per cell were determined by RT-qPCR.

3.7. Suramin appears to inhibit (re)initiation of CHIKV RNA synthesis

To gain more insight into the mechanism by which suramin inhibits CHIKV RNA synthesis, *in vitro* assays with RTCs isolated from CHIKV replicon-transfected cells were employed. Also in this biosafe system, suramin inhibited RNA synthesis, with an IC₅₀ of 6.7 μ M (Fig. 6A, supplemental figure 1B). The inhibitory effect of nucleoside analogs can be reversed by adding an excess of NTPs [28], as can be seen for 3'dUTP in Fig. 6B. The inhibitory effect of suramin could not be reversed by an excess of NTPs, suggesting the compound does not compete with NTPs.

As can be seen in figures 1 and 6B, even at very high suramin doses some incorporation of 32 P-CTP into viral RNA remained. We hypothesized that this might be caused by complexes already involved in RNA synthesis (interacting with the template) that are insensitive to suramin, which would then mainly inhibit (re)initiating RTCs. To test our hypothesis, we allowed reactions to proceed for 15 min in the absence of 32 P-CTP (so products will not be detected), and in the presence or absence of 500 µM of suramin or the nucleoside analog 3'dUTP as a control. After 15 min, 32 P-CTP was added and the reactions were allowed to proceed for 60 min (Fig. 6C, condition 1). Under this condition suramin completely blocked the synthesis of radiolabeled RNA, suggesting it was able to inhibit (re)initiating RdRps during the first 15 min of the reaction, during which the "suramin-resistant RTCs" finished their products that are non-radioactive (and therefore are not detected). Merely preincubating RTCs with suramin for 15 min before starting the *in vitro* reaction did not abolish the residual activity (Fig. 6C, condition 2), demonstrating that the effect (condition 1) is not due to just giving the compound time to access the RTC.





(A) Inhibition of the *in vitro* RNA-synthesizing activity of RTCs isolated from CHIKV replicon-transfected cells by suramin. The nucleoside analog 3'dUTP was used as a control. (B) The inhibitory effect of 50 μ M 3'dUTP or 32 μ M suramin in a standard *in vitro* reaction and in a reaction supplemented with 200 μ M NTPs. Reaction products were quantified and normalized to untreated control reactions (100%). (C) RNA synthesizing activity in a 60-min reaction that followed a 15-min pretreatment with 32 μ M suramin or 50 μ M 3'dUTP under conditions that sustain (condition 1) or do not sustain (condition 2) RNA synthesis. ³²P-CTP was absent during the first 15 min, but was present during the following 60 min. For details see section 3.7. The bars represent the average (± stdev) of two independent experiments; ***p < 0.005, **p < 0.01, *p < 0.05.

3.8. Effect of suramin containing liposomes on CHIKV replication

Due to its charged groups suramin poorly crosses the cell membrane. In an attempt to improve suramin delivery into the cell we tested various cationic liposome formulations containing the negatively charged suramin for their efficacy to inhibit CHIKV replication in CPE protection assays (Table 2, supplemental figure S4). This approach is expected to decrease drug-related toxicity, enhance cellular uptake, and might lead to higher

accumulation in macrophage-rich (CHIKV-infected) tissues when used *in vivo*. Control liposomes without suramin exhibited a relatively high cytotoxicity, while suramin-containing liposomes were less cytotoxic, with CC_{50} values of 50-100 μ M. The enhanced toxicity of the control liposomes was previously reported for similar cationic liposomes extensively investigated and employed as non-viral gene delivery systems. It is well known that cationic lipids (i.e. DDAB and DDAC included in the liposomes) are "membrane active" molecules, which interfere with membrane function and affect the integrity of the cell or subcellular compartments, leading to toxicity. The formation of ionic complexes between the positively charged lipids and the negatively charged suramin leads to a reduction of the toxicity of the cationic liposomes. Formulation #PC3-Cl1-sur.2 inhibited CHIKV replication with an EC₅₀ of ~62 μ M, which is slightly better than suramin dissolved in water (79 μ M). The #PC3-Cl1-sur.2 formulation is an interesting starting point for further optimization to improve the efficacy of suramin. The liposome-mediated direct delivery into cells might also help in studying specifically the effect of suramin on viral RNA synthesis, separate from the compound's effect on the early stage of infection.

Formulation iden- tification name	PC content (mM)	DDAC content (mM)	DDAB content (mM)	Suramin (mM)	EC ₅₀ (μΜ)	СС ₅₀ (µМ)
#PC3-Cl1-sur0	3.0	1.0	-	-	ND	4
#PC3-Br1-sur0	3.0	-	1.0	-	ND	7
#PC3-Cl1-sur.2	3.0	1.0	-	0.2	62	~100
#PC3-Br1-sur.2	3.0	-	1.0	0.2	~ 100	~ 100
#PC9-Cl1-sur0	9.0	1.0	-	-	ND	7
#PC9-Br1-sur0	9.0	-	1.0	-	ND	35
#PC9-Cl1-sur.2	9.0	1.0	-	0.2	ND	~ 50
#PC9-Br1-sur.2	9.0	-	1.0	0.2	ND	~ 50

Table 2. Antiviral and cytotoxic effects of suramin-containing and empty control liposomes,determined by CPE protection assay with Vero E6 cells.

ND: EC_{50} not determined due to low CC_{50} . PC: phosphatidylcholine, DDAC: distearyldimethylammonium chloride, DDAB: dimethyl-dioctadecylammonium bromide.

3.9. Essential moieties in the structure of suramin

Suramin is a symmetric molecule (Fig. 7A; 1a in Table 3) with in the center a urea (NH–CO–NH) functional group as the "neck". Suramin also contains two benzene rings with amide linkers on each side as the "arms" and possesses two naphthalene rings as the



Α

Suramin structure

Figure 7. Effect of suramin-related molecules on CHIKV RNA synthesis.

(A) Structure of suramin. (B) Effect of the suramin-related compounds indicated above the lanes (structures are depicted in table 3) on CHIKV RNA synthesis *in vitro*. See legend of Fig. 1 for details.

"palms," and six sulfonate groups as the "fingers". Table 3 lists its structure (1a) and those of ten related compounds (1–5), which have fewer sulfonate fingers, shorter arms, only one side, or no neck in comparison with suramin. These compounds were tested for their ability to inhibit CHIKV RNA synthesis *in vitro* (Fig. 7B; Table 3) and CHIKV replication in cell culture in a CPE protection assay. Examination of the biological activities of compounds 1–5 shown in Table 3 indicates that CHIKV RNA synthesis was inhibited by compounds 1a, 1b, 5a and 5b. Unsymmetrical compounds 2a–d and 3a–c, which had only one arm, were inactive regardless of its length. Suramin (1a) with six sulfonate groups exhibited greater anti-CHIKV activity (EC_{50} 80 µM) than tetrasulfonate 1b (EC_{50} 200 µM) in cell culture. This preliminary structure-activity analysis gives a first indication of the moieties that are important for suramin's activity and suggest that the two antiviral activities can be separated and optimized independently. The synthesis of additional related compounds is currently in progress and further analysis of structures and biological activities is required to design a rational route for optimization of suramin-like molecules into effective antiviral drugs.
compound structure	label	x =	y =	finger
SO ₃ Na	a	NH	Н	2
	b	NO ₂	Н	2
	c	NH 2	Me	2
¥`x	d	NO ₂	Me	2
SO ₃ Na	a	NH 2	—	2
NaO ₃ S HN O	b	NO ₂	—	2
$\qquad \qquad $	с	NHC(=S)OEt	_	2
NaO ₃ S SO ₃ Na NaO ₃ S HN H SO ₃ Na NH SO ₃ Na	_	_	_	4
NaO ₃ S SO ₃ Na NaO ₃ S HN O O NH SO ₃ Na	a	_	Н	4
	Ь	_	Me	4
NaO ₃ S X NaO ₃ S HN O NH SO ₃ Na	a (suramin)	SO Na 3	_	6
	b	Н	_	4

Table 3. Structures of suramin-related compounds and their effect on CHIKV replicationin cell culture and RNA synthesis *in vitro*.

palm	arm	neck	Effect on RNA synthesis	EC ₅₀ (μM)	СС ₅₀ (µМ)
1	short $\times 1$	0	-	739	>800
1	short $\times 1$	0	-	>800	>800
1	short $\times 1$	0	-	>800	>800
1	short \times 1	0	-	>800	>800
1	$\log \times 1$	0	-	420	>800
1	$\log \times 1$	0	-	>800	>800
1	$\log \times 1$	0	-	>800	>800
2	0	1	-	>800	>800
2	short × 2	1	+	403	>800
2	short $\times 2$	1	+	>800	>800
2	long × 2	1	+	79	>800
2	$long \times 2$	1	+	210	>800

4. Conclusion

In this study we show that the anti-parasitic drug suramin inhibits the replication of CHIKV and other alphaviruses. We discovered that while *in vitro* suramin is a potent inhibitor of RNA synthesis, in cell culture the compound mainly inhibits an earlier, post-attachment step of the CHIKV replicative cycle, likely viral entry. The inhibition of an early step in infection (receptor binding or entry or uncoating) has also been reported for a variety of other viruses [11-13, 15, 16, 18, 22]. Suramin appears to inhibit (re)initiation of CHIKV RNA synthesis, maybe by interfering with binding of the template RNA. This would be in line with earlier *in vitro* studies, reporting that suramin inhibits various RNA-binding enzymes like viral polymerases [7, 27, 36] and helicases [24, 25]. Several suramin-related compounds were analyzed, and though these compounds were not more effective, they provided insight into the (different) structural elements that are important for both inhibitory activities of suramin observed *in vitro* and in cell culture.

Clinical trials that evaluated the efficacy of suramin for treating patients chronically infected with HIV or HBV [9, 37] revealed serious side-effects during the required long term treatment. As a result and also because more promising drugs like 2',3'-dideoxy-3'- azidothymidine (AZT) became available [38], further clinical development was halted. The treatment of CHIKV infections would likely not require prolonged treatment and, therefore, we feel it is still worthwhile to obtain more insight into the modes of action of suramin and to explore its therapeutic potential.

Acknowledgements

This work was supported by the EU-FP7 EUVIRNA (#264286) and SILVER (#260644) grants, as well as by the Ministry of Science and Technology of R.O.C. (# NSC 103-2923-I-008-001 and MOST 103-2113-M-007-018-MY3) and the National Central University, Taiwan (# 103G603-14). We are grateful to the LUMC biosafety officer dr. Gijsbert van Willigen for his support in keeping our BSL-3 facility operational and to dr. Gilles Querat (Aix Marseille University) for supplying CHIKV RNA isolated during the outbreak at St. Martin.

References

- Weaver, S.C., Arrival of chikungunya virus in the new world: prospects for spread and impact on public health. PLoS Negl Trop Dis, 2014. 8(6): p. e2921.
- Weaver, S.C. and M. Lecuit, Chikungunya virus and the global spread of a mosquito-borne disease. N Engl J Med, 2015. 372(13): p. 1231-9.
- Thiberville, S.D., *et al.*, Chikungunya fever: epidemiology, clinical syndrome, pathogenesis and therapy. Antiviral Res, 2013. 99(3): p. 345-70.
- 4. Voogd, T.E., *et al.*, Recent research on the biological activity of suramin. Pharmacol Rev, 1993. 45(2): p. 177-203.
- Liu, N. and S. Zhuang, Tissue protective and anti-fibrotic actions of suramin: new uses of an old drug. Curr Clin Pharmacol, 2011. 6(2): p. 137-42.
- De Clercq, E., Curious Discoveries in Antiviral Drug Development: The Role of Serendipity. Med Res Rev, 2015. 35(4): p. 698-719.
- De Clercq, E., Suramin: a potent inhibitor of the reverse transcriptase of RNA tumor viruses. Cancer Lett, 1979. 8(1): p. 9-22.
- Broder, S., *et al.*, Effects of suramin on HTLV-III/LAV infection presenting as Kaposi's sarcoma or AIDSrelated complex: clinical pharmacology and suppression of virus replication *in vivo*. Lancet, 1985. 2(8456): p. 627-30.
- Kaplan, L.D., *et al.*, Lack of response to suramin in patients with AIDS and AIDS-related complex. Am J Med, 1987. 82(3 Spec No): p. 615-20.
- Schols, D., *et al.*, Dextran sulfate and other polyanionic anti-HIV compounds specifically interact with the viral gp120 glycoprotein expressed by T-cells persistently infected with HIV-1. Virology, 1990. 175(2): p. 556-61.
- 11. Aguilar, J.S., M. Rice, and E.K. Wagner, The polysulfonated compound suramin blocks adsorption and lateral difusion of herpes simplex virus type-1 in vero cells. Virology, 1999. 258(1): p. 141-51.
- 12. Baba, M., *et al.*, Selective inhibition of human cytomegalovirus replication by naphthalenedisulfonic acid derivatives. Antiviral Res, 1993. 20(3): p. 223-33.
- Schulze, A., P. Gripon, and S. Urban, Hepatitis B virus infection initiates with a large surface proteindependent binding to heparan sulfate proteoglycans. Hepatology, 2007. 46(6): p. 1759-68.
- Petcu, D.J., *et al.*, Suramin inhibits *in vitro* infection by duck hepatitis B virus, Rous sarcoma virus, and hepatitis delta virus. Virology, 1988. 167(2): p. 385-92.
- Garson, J.A., *et al.*, Suramin blocks hepatitis C binding to human hepatoma cells *in vitro*. J Med Virol, 1999. 57(3): p. 238-42.
- 16. Chen, Y., *et al.*, Dengue virus infectivity depends on envelope protein binding to target cell heparan sulfate. Nat Med, 1997. 3(8): p. 866-71.
- Crance, J.M., *et al.*, Inhibition of sandfly fever Sicilian virus (Phlebovirus) replication *in vitro* by antiviral compounds. Res Virol, 1997. 148(5): p. 353-65.
- Iqbal, M., H. Flick-Smith, and J.W. McCauley, Interactions of bovine viral diarrhoea virus glycoprotein E(rns) with cell surface glycosaminoglycans. J Gen Virol, 2000. 81(Pt 2): p. 451-9.
- Ellenbecker, M., J.M. Lanchy, and J.S. Lodmell, Inhibition of Rift Valley fever virus replication and perturbation of nucleocapsid-RNA interactions by suramin. Antimicrob Agents Chemother, 2014. 58(12): p. 7405-15.
- 20. Jiao, L., *et al.*, Structure of severe fever with thrombocytopenia syndrome virus nucleocapsid protein in complex with suramin reveals therapeutic potential. J Virol, 2013. 87(12): p. 6829-39.
- 21. Tamura, M., *et al.*, Genogroup II noroviruses efficiently bind to heparan sulfate proteoglycan associated with the cellular membrane. J Virol, 2004. 78(8): p. 3817-26.
- 22. Wang, Y., et al., Suramin inhibits EV71 infection. Antiviral Res, 2014. 103: p. 1-6.

- Ren, P., *et al.*, The approved pediatric drug suramin identified as a clinical candidate for the treatment of EV71 infection-suramin inhibits EV71 infection *in vitro* and *in vivo*. Emerg Microbes Infect, 2014. 3(9): p. e62.
- 24. Mukherjee, S., *et al.*, Identification and analysis of hepatitis C virus NS3 helicase inhibitors using nucleic acid binding assays. Nucleic Acids Res, 2012. 40(17): p. 8607-21.
- Basavannacharya, C. and S.G. Vasudevan, Suramin inhibits helicase activity of NS3 protein of dengue virus in a fluorescence-based high throughput assay format. Biochem Biophys Res Commun, 2014. 453(3): p. 539-44.
- Mastrangelo, E., *et al.*, Structure-based inhibition of Norovirus RNA-dependent RNA polymerases. J Mol Biol, 2012. 419(3-4): p. 198-210.
- Tarantino, D., et al., Naphthalene-sulfonate inhibitors of human norovirus RNA-dependent RNApolymerase. Antiviral Res, 2014. 102: p. 23-8.
- 28. Albulescu, I.C., *et al.*, An *in vitro* assay to study chikungunya virus RNA synthesis and the mode of action of inhibitors. J Gen Virol, 2014.
- 29. Scholte, F.E., *et al.*, Characterization of synthetic Chikungunya viruses based on the consensus sequence of recent E1-226V isolates. PLoS One, 2013. 8(8): p. e71047.
- 30. Delang, L., *et al.*, Mutations in the chikungunya virus non-structural proteins cause resistance to favipiravir (T-705), a broad-spectrum antiviral. J Antimicrob Chemother, 2014. 69(10): p. 2770-84.
- Coffey, L.L., *et al.*, Arbovirus high fidelity variant loses fitness in mosquitoes and mice. Proc Natl Acad Sci U S A, 2011. 108(38): p. 16038-43.
- 32. Mastrangelo, E., *et al.*, Delivery of suramin as an antiviral agent through liposomal systems. ChemMedChem, 2014. 9(5): p. 933-9.
- van Hemert, M.J., *et al.*, SARS-coronavirus replication/transcription complexes are membrane-protected and need a host factor for activity *in vitro*. PLoS Pathog, 2008. 4(5): p. e1000054.
- 34. Scholte, F.E., *et al.*, Stress Granule Components G3BP1 and G3BP2 Play a Proviral Role Early in Chikungunya Virus Replication. J Virol, 2015. 89(8): p. 4457-69.
- Fros, J.J., *et al.*, Chikungunya virus nonstructural protein 2 inhibits type I/II interferon-stimulated JAK-STAT signaling. J Virol, 2010. 84(20): p. 10877-87.
- Ono, K., H. Nakane, and M. Fukushima, Differential inhibition of various deoxyribonucleic and ribonucleic acid polymerases by suramin. Eur J Biochem, 1988. 172(2): p. 349-53.
- Loke, R.H., *et al.*, Suramin treatment for chronic active hepatitis B--toxic and ineffective. J Med Virol, 1987. 21(1): p. 97-9.
- De Clercq, E., Antiviral drug discovery: ten more compounds, and ten more stories (part B). Med Res Rev, 2009. 29(4): p. 571-610.



Figure S1. Quantification of the effect of suramin on alphavirus RNA synthesis *in vitro*. *In vitro* RNA synthesis assays were performed with RTCs, either isolated from CHIKV-infected Vero E6 cells (A) or from BHK-21 cells transfected with a CHIKV replicon (B), in the presence of increasing concentrations of suramin. RNA synthesis, i.e. the amount of incorporated ³²P-CTP, was quantified by determining the volume of the bands that are visible in Fig. 1A and Fig. 6A using the Quantity One* software. The values (sum of genome and subgenomic RNA) were normalized to the untreated control sample (100%) and were corrected for variations in RNA isolation efficiency and loading using 18S ribosomal RNA signal (detected by hybridization). The corrected normalized values (amount of RNA synthesized) obtained at each suramin concentration were plotted against the log of the suramin concentration and the IC₅₀ value was determined by non-linear regression using GraphPad Prism 5.



Figure S2. Antiviral activity of suramin in cell culture. Results of CPE protection assays using three different CHIKV strains, SINV and SFV. EC_{50} values for suramin (mentioned in table 1) were determined by non-linear regression using GraphPad Prism. The average and SD were calculated based on 2 independent experiments, performed in quadruplicate.



Figure S3. Effect of suramin on the replication of wt CHIKV LS3 and two mutants resistant to favipiravir and ribavirin in BHK-21 cells. CPE reduction assays were performed in BHK-21 cells, to make it possible to do parallel experiments with ribavirin, as ribavirin is converted to its active triphosphate form only very inefficiently in Vero-E6 cells. The cells were infected with CHIKV LS3 (wt), CHIKV M5 (favipiravir resistant) or CHIKV C483Y (ribavirin resistant) and treated with various concentrations of suramin (A), favipiravir (B) or ribavirin (C).



Figure S4. Effect of suramin-liposome preparation #PC3-Cl1-sur.2 and empty control liposomes on cell viability and CHIKV replication. Vero E6 cells were treated with suramin containing liposome formulation #PC3-Cl1-sur.2 (sur2) or 'empty' control liposomes #PC3-Cl1-sur.0 (sur0) and their effect on viability (uninfected cells) and CHIKV replication (by CPE reduction assay) was determined.

Chapter 4

Suramin inhibits Zika virus replication through multiple mechanisms

Irina C. Albulescu, Kristina Kovacikova, Ali Tas, Eric J. Snijder, and Martijn J. van Hemert*

Molecular Virology Laboratory, Department of Medical Microbiology, Leiden University Medical Center, Leiden, The Netherlands.

*Corresponding author.

Published in Antiviral Research, 2017 Jul; 143:230-236.

Abstract

Zika virus (ZIKV) is a mosquito-borne flavivirus that mostly causes asymptomatic infections or mild disease characterized by low-grade fever, rash, conjunctivitis, and malaise. However, the recent massive ZIKV epidemics in the Americas have also linked ZIKV infection to fetal malformations like microcephaly and Guillain-Barré syndrome in adults, and have uncovered previously unrecognized routes of vertical and sexual transmission. Here we describe inhibition of ZIKV replication by suramin, originally an anti-parasitic drug, which was more recently shown to inhibit multiple viruses. In cell culture-based assays, using reduction of cytopathic effect as read-out, suramin had an EC₅₀ of ~40 μ M and a selectivity index of 48. In single replication cycle experiments, suramin treatment also caused a strong dose-dependent decrease in intracellular ZIKV RNA levels and a >3-log reduction in infectious progeny titers. Time-of-addition experiments revealed that suramin inhibits a very early step of the replication cycle as well as the release of infectious progeny. Only during the first two hours of infection suramin treatment strongly reduced the fraction of cells that became infected with ZIKV, suggesting the drug affects virus binding/entry. Binding experiments at 4°C using ³⁵S-labeled ZIKV demonstrated that suramin interferes with attachment to host cells. When suramin treatment was initiated post-entry, viral RNA synthesis was unaffected, while both the release of genomes and the infectivity of ZIKV were reduced. This suggests the compound also affects virion biogenesis, possibly by interfering with glycosylation and the maturation of ZIKV during its traffic through the secretory pathway. The inhibitory effect of suramin on ZIKV attachment and virion biogenesis and its broad-spectrum activity warrant further evaluation of this compound as a potential therapeutic.

Highlights

Suramin inhibits Zika virus replication in cell culture Suramin interferes with Zika virus attachment to host cells Suramin also affects release of infectious Zika virus

1. Introduction

Zika virus (ZIKV) is a flavivirus that was first isolated in 1947 in Uganda [1] and is primarily transmitted by Aedes mosquitos. Up to 2007, it received little attention, as only sporadic outbreaks were reported and these were not linked to serious disease. ZIKV has gained attention after the 2007 outbreak on Yap island and the 2013 epidemic in French Polynesia. Research efforts and public concern were further sparked by the massive epidemic in Brazil and many other South- and Central-American countries, which uncovered a link between ZIKV infection and serious foetal neurodevelopmental defects and neurological complications in adults.

About 80% of ZIKV infections are asymptomatic. In combination with ZIKV persistence in semen and the potential for sexual transmission, this poses a risk to women that are pregnant or trying to conceive. Symptomatic ZIKV infections are commonly characterized by low-grade fever, skin rash, conjunctivitis, and general malaise. However, the enormous scale of the recent epidemics has also revealed uncommon but more serious consequences, like the Guillan-Barré syndrome and foetal neurodevelopmental defects, like microcephaly. These could be due to ZIKV's ability to infect and persist in several immune-privileged tissues, like the central nervous system, placenta and testis [2].

ZIKV has a 10.7 kb ssRNA genome of positive polarity, which is 5' capped and lacks a poly(A) tail. The replicative cycle begins with binding to receptor and co-receptors, like AXL and TIM1 [3, 4], followed by entry through receptor-mediated endocytosis, uncoating and translation of the genome into a single polyprotein. The latter is processed by cellular and viral (NS3) proteases to yield the structural proteins C, prM and E and 7 nonstructural proteins, which are responsible for RNA replication in association with modified endoplasmic reticulum (ER) membranes and interactions with the host, including counteracting innate immune responses. Newly synthesized genomes, together with C, prM and E, are assembled into immature virions that bud into the ER lumen. These pass through the secretory pathway during which prM cleavage by the host protease furin leads to maturation, before particles are released into the extracellular space.

There are no registered vaccines against ZIKV and efforts to identify inhibitors of ZIKV replication have been initiated only recently, for example in the form of two large-scale compound screens that yielded surprisingly few common hits [5-9]. Considering the generally low success rate in antiviral drug development, it is crucial to identify a large number of ZIKV inhibitors in cell culture and elucidate their mode of action, in order to have sufficient lead compounds with the potential to advance further towards clinical development.

Suramin is an approved anti-parasitic drug that also blocks the replication of a variety of viruses (see [10] and references therein), including arboviruses like chikungunya virus (CHIKV) [10-12] and dengue virus [13]. Animal experiments have indicated that suramin can be used for preventing and treating enterovirus-71 and CHIKV infections [14, 15].

In this study we show that suramin effectively blocks ZIKV replication by interfering with attachment, while the compound also affects a later step, as it reduces the release of infectious progeny. Our results warrant further evaluation of suramin as an anti-ZIKV compound. Its broad spectrum of activity, limited toxicity and approved status offer interesting perspectives for repurposing this anti-parasitic drug as an antiviral.

2. Materials and Methods

2.1. Cells, viruses and compounds

Vero cells (ATCC CCL-81) were grown in Dulbecco's modified Eagle's medium (DMEM), supplemented with 8% fetal calf serum (FCS), 100 IU/mL of penicillin and 100 μ g/mL of streptomycin at 37°C in 5% CO₂. ZIKV SL1602 was isolated in Leiden, the Netherlands, in 2016 from an infected traveler returning from Suriname (van Boheemen et al, submitted). Vero cells were seeded (150,000 cells or 75,000 cells/well, respectively) in 12-well or 24-well clusters 24 h prior to infection with ZIKV (MOI of 3) in EMEM with 25 mM HEPES, 2% FCS, 2 mM L-glutamine, and antibiotics (EMEM/2%FCS). After 2 h, the inoculum was removed, cells were washed 3 times with PBS after which they were maintained in EMEM/2%FCS. All experiments with ZIKV were performed in our BSL-3 facility. Suramin (Sigma) was dissolved in MilliQ to yield a 25 mM stock and mycophenolic acid (Sigma) was dissolved in ethanol to yield a 10 mM solution.

2.2. Cytopathic effect (CPE) reduction assay

CPE reduction assays with Vero cells were performed by seeding cells at a density of 5,000 cells/well in 96-well clusters, 24 h before compound treatment and infection. The next day, 1.5-fold or 2-fold serial dilutions of the compound were added to the cells, followed by 500 PFU/well of ZIKV in a total volume of 150 µL. Each concentration was tested in quadruplicate and each assay plate contained the following controls: no cells, uninfected&untreated cells, infected&untreated cells and infected&solvent-treated cells. After 4 days, 30 µL/well of CellTiter 96^{*} Aqueous Non-Radioactive Cell Proliferation reagent (Promega) was added. After a 3-h incubation, reactions were stopped and virus was inactivated by adding 30 µL of 37% formaldehyde, followed by measuring the absorption at 490 nm. Viability assays on uninfected cells were performed in parallel to determine the CC_{50} . Data were normalized to untreated uninfected cells and EC_{50} and CC_{50} values were calculated with Graph-Pad Prism 7 (see section 2.8).

2.3. ZIKV titration and plaque reduction neutralization titer (PRNT) assay

Plaque assays were performed as described (Van Boheemen *et al*, submitted). Briefly, after a 1-min centrifugation at 13,000xg, virus-containing cell culture supernatants were 10-fold serially diluted in medium and 500-µL volumes were adsorbed for 2 h on confluent Vero cell monolayers in 6-well clusters. The inoculum was removed and an overlay of 1.2% Avicel RC-581 (FMC BioPolymer) in DMEM, 2% FCS, 25 mM HEPES, and antibiotics, was added. After a 5-day incubation, cells were fixed with formaldehyde, stained with crystal violet and plaques were counted. For PRNT, $3.2x10^4$ PFU of ZIKV were incubated with various dilutions of heat-inactivated patient serum (against ZIKV SL1602) or a control serum for 1h at 37°C. Afterwards 250-µL volumes of 10-fold serial dilutions were adsorbed for 2 h on confluent Vero cell monolayers in 12-well clusters. An overlay was placed on top and plaques were visualized after 5 days as described for the plaque assay.

2.4. ZIKV RNA analysis

RNA was isolated from ZIKV-infected cells using TriPure isolation reagent (Roche) according to the manufacturer's instructions. ZIKV RNA was isolated from the medium of infected cells using the Qiamp Viral mini kit (Qiagen) with AVL lysis buffer spiked with equine arteritis virus (EAV) as internal control, as described previously [16]. ZIKV RNA copy numbers were determined by an in-house internally-controlled TaqMan multiplex RT-qPCR, using PGK1 as a reference gene, as described (Manuscript in preparation). Briefly, the detection of ZIKV RNA was done with the primer pair Fw 5'-AATGGCAGTCAGTGGAGATG/ Rv 5'-ACTCTTGTGTGTCCTTCCTAAC and a FAM-labelled probe (5'-6FAM-ATAGGTTTGCACATGCCCTCAGGT-3'-BHQ_1) using the TaqMan Fast Virus 1-Step Master Mix (Applied Biosystems) and the CFX384 Touch[™] Real-Time PCR Detection System. Known quantities of *in vitro* transcribed RNA were used to generate calibration curves for absolute quantification of copy numbers. Each sample was analyzed in triplicate.

2.5. Immunofluorescence assay

Vero cells (50,000 per well) were seeded on coverslips in 24-well clusters. The next day they were infected with ZIKV or mock-infected and treated with various concentrations of suramin. At 26 h p.i. cells were fixed with 3% paraformaldehyde in PBS, and processed for immunofluorescence microscopy as previously described [16]. ZIKV-specific proteins were visualized with a 1:1000 dilution of the SL1602 patient antiserum, and double-stranded RNA was stained with mouse monoclonal antibody J2 (English & Scientific Consulting). Detection of primary antibodies was done with 1:500 dilutions of goat anti-human Alexa488 and donkey anti-mouse Cy3.

2.6. Production of ³⁵S-labeled ZIKV and binding assays

Confluent Vero cells (75 cm²) were infected with ZIKV at an MOI of 0.05 or mock infected and incubated for 3 days. Medium was replaced with protein-labeling medium (Met- and Cys-free DMEM) containing 88 μ Ci of ³⁵S-labelled Met and Cys (0.0176 μ Ci/ μ L) and cells were incubated for 15 h. The culture supernatants were collected and cellular debris was removed by centrifugation for 10 min at 233 x g. Unincorporated label was removed by pelleting virions through a cushion of 20% sucrose in TESV buffer (20 mM Tris-HCl pH 7.4, 1 mM EDTA, 100 mM NaCl) by ultracentrifugation for 2 h at 45,000 rpm and 4°C using a Beckmann SW55 rotor. After removal of the supernatant and sucrose layer, the virus pellet was washed with TESV buffer, and resuspended in 0.6 mL of TESV. Aliquots of labelled virus were stored at -80°C. For binding assays, Vero cells seeded in 24-well clusters were incubated with $4 \mu L$ (3.2×10^4 PFU) of radioactive ZIKV for 2 h at 4°C in a total volume of 125 μ L. After removal of the inoculum and washing three times with PBS, cells were lysed in 200 μ L of 100 mM Tris-HCl pH 6.8, 8% SDS (w/v), 40 mM DTT, and 40% glycerol to quantify cell-bound radioactivity using a Beckman-Coulter LS6500 Multi-Purpose scintillation counter and Ultima Gold^m scintillation liquid.

2.8. Statistical analysis

GraphPad Prism 7 was used for EC_{50} , IC_{50} and CC_{50} calculations by non-linear regression and for statistical analyses by one-way ANOVA with Dunnett's multiple comparison test.

3. Results

3.1. Suramin inhibits ZIKV in CPE reduction assays

To determine the anti-ZIKV activity of compounds, we have set up CPE reduction assays with Vero cells, which were treated with serial dilutions of compound, followed by mock-infection or ZIKV-infection at an MOI of 0.05. After a 4-day incubation, colorimetric viability assays were performed to quantify the protective effect of a compound against ZIKV-induced CPE. Publications on compounds with demonstrated anti-ZIKV activity were not available at the time we set up our assay (3/2016) and therefore we included mycophenolic acid (MPA), known to inhibit a broad-spectrum of viruses, as a positive control. MPA had an EC₅₀ of 0.42 μ M and a CC₅₀ of over 250 μ M in our assay and its antiviral effect has also been demonstrated by others ([6, 8]. Suramin protected Vero cells from ZIKV-induced cell death with an EC₅₀ value of 39.8 μ M (supplemental Fig. S1). The CC50 of suramin was 1.9 mM, resulting in a selectivity index (SI) of 48.

3.2. Suramin reduces ZIKV RNA accumulation and infectious progeny titers in cell culture To confirm the antiviral effect of suramin treatment observed in the CPE reduction assays, a dose-response experiment was performed with ZIKV-infected Vero cells (MOI 3) that were treated with 25 to 400 μ M of suramin (or 2.5 μ M MPA as positive control) from 2 h prior to infection up to 44 h p.i. Suramin treatment caused a convincing dose-dependent reduction in intracellular ZIKV RNA levels (Fig. 1A), yielding a 3-log decrease in ZIKV genome copies after treatment with 400 μ M of suramin. An even stronger effect was observed on the production of infectious progeny (Fig.1B). No infectious particles could be detected at the highest suramin dose tested, although the presence of 40 μ M residual suramin in the 10-1 dilution of the samples used for titration by plaque assay may have increased the limit of detection to 100 PFU/mL for samples treated with 400 μ M suramin. Nonetheless, even

4

at 50 μ M, suramin treatment caused a more than 2-log reduction in progeny virus titers. MPA treatment only modestly inhibited RNA synthesis and virus production, yielding a reduction of about 1-log in both assays.



Figure 1. Effect of suramin on the accumulation of intracellular ZIKV RNA and infectious progeny titers. A. ZIKV-infected Vero cells were treated with 25 to 400 μ M suramin from 2 hours prior to infection till 44 h p.i. Untreated and mock-infected cells were included as controls and the inhibitory effect of suramin was compared to that of 2.5 μ M MPA. Total RNA was isolated from cells (n=3) at 44 h p.i. and intracellular ZIKV RNA copy numbers were determined by RT-qPCR using a standard curve of *in vitro* transcribed ZIKV RNA for absolute quantification. B. ZIKV titers in the medium of suramin-treated cells (n=3) were determined by plaque assay. Significant differences are indicated by * (***, p<0.0001).

3.3. Suramin inhibits the attachment of ZIKV to host cells

A time-of-addition experiment was performed to determine which step(s) of the ZIKV replication cycle are inhibited by suramin. The compound was added to Vero cells to a final concentration of 200 µM either two hours before infection, at the time of infection, or at various time points post infection, after which the compound remained present until sample collection at the end of the experiment (Fig. 2). Suramin had the strongest inhibitory effect on the accumulation of intracellular ZIKV RNA when added prior to or together with the virus (Fig. 2, samples -2 and 0), resulting in a ~100-fold reduction in the amount of intracellular ZIKV RNA. When the compound was added 30 min or more after adding the virus, the inhibitory effect on the accumulation of intracellular ZIKV RNA diminished rapidly. Immunofluorescent staining of ZIKV-infected Vero cells treated with 25-200 µM suramin from 2 h before till 2 h after addition of the virus revealed that the compound caused a dose-dependent reduction in the number of infected cells reaching 95% at 200 μ M (Fig. 3A). The remaining ZIKV-positive cells appeared not to be protected by suramin treatment as their signal intensity in immunofluorescence microscopy resembled that of untreated cells (see inset of Fig. 3A). These remaining ZIKV-positive cells are most likely responsible for the residual production of infectious virus that was observed when cells were treated with 200 µM suramin from -2 to 2 h p.i. (Fig. 3B). Nonetheless, we concluded that suramin treatment during the early steps of the replication cycle (binding/entry) only, still resulted in an up to 100-fold and dose-dependent reduction of progeny virus titers (at 26 h p.i.).



Figure 2. Effect of suramin on ZIKV replication in time-of-addition assay ZIKV-infected Vero cells were treated with 200 μ M of suramin starting 2 h before infection until 26 h p.i. Untreated (Ut) and mock-infected cells were included as controls. Total RNA was isolated from cells at 26 h p.i. and intracellular ZIKV RNA copy numbers were determined by RT-qPCR (n=3). Significant differences compared to untreated infected cells are indicated by * (*, p<0.05; **, p<0.01; ***, p<0.001).

To test whether suramin affects ZIKV attachment to host cells, we have produced ³⁵S-labeled ZIKV for use in virus binding assays. The radioactive ZIKV was incubated with Vero cells for 2 h at 4°C, which should only allow binding, while preventing endocytosis/entry. After washing, cells were lysed and the amount of cell-bound radioactivity was determined by scintillation counting. Vero cell-associated radiolabel was readily detected when ³⁵S-labeled ZIKV was used in this assay, while hardly any radiolabel was bound when Vero cells were incubated with a virus-free control sample that had been prepared by ³⁵S-labeling of mock-infected cells (Fig. 3C). To determine whether inhibition of binding could be measured in this assay, ³⁵S-labeled ZIKV was incubated with a neutralizing antiserum from a ZIKV-infected patient prior to the binding assay (Fig. 3C). This serum completely neutralized the virus at 1:5 (and higher) dilutions in plaque reduction neutralization assays (Fig. 3C, inset). When the binding assay was performed with virus that was neutralized with a 1:5 antiserum dilution, a 50% reduction in cell-bound radiolabel was observed. It remains unclear whether the remaining amount of radioactivity resulted from cell-bound radiolabeled virus or from radiolabeled host proteins that might be absent in the ³⁵S-labeled control sample obtained from mockinfected cells as the latter had not undergone the ZIKV-induced CPE. Nonetheless, the assay allowed us to measure decreases in ZIKV binding. When various suramin concentrations were tested in this assay, we observed a dose-dependent inhibition of ³⁵S-ZIKV binding of up to 50% (Fig. 3D), the same residual level of labeling measured when using pre-neutralized virus (Fig. 3C). Taken together these results indicate that suramin treatment affects binding of ZIKV to the host cell.





A. ZIKV-infected Vero cells were treated with 25 to 200 µM of suramin, starting 2 hours prior to the addition of virus till 2 h p.i. At 26 h p.i. cells were fixed and analyzed by immunofluorescence microscopy. Quantification of the percentage of infected cells was performed by determining the number of dsRNA (red) and ZIKV protein (green) positive cells in two images acquired per sample. Results are presented as the number of infected cells (%) relative to the untreated control. The inset shows typical micrographs of untreated cells and cells treated with 200 µM of suramin only during the early steps of infection. B. ZIKV titers in the medium of Vero cells that were treated with 25 to 200 µM of suramin from -2 to 2 h p.i. Supernatants were harvested at 26 h p.i. from two independent infections performed in duplicate, and ZIKV titers were determined by plaque assay (2 replicates per supernatant). C. ZIKV binding assay. 35S-labeled ZIKV was incubated with Vero cells for 2 h at 4°C. A virusfree control sample that had been prepared by ³⁵S-labeling of mock-infected cells was included as a negative control (Mock). ³⁵S-labeled ZIKV that was neutralized (ZIKV/Neu) with a patient immune serum (see inset for PRNT results) was included as a control for reduced binding. After extensive washing the remaining cell-bound radioactivity was quantified by scintillation counting (n=2). D. Effect of suramin on ZIKV binding. ³⁵S-labeled ZIKV was incubated for 2 h at 4°C with Vero cells without suramin or in the presence of 12.5 to 100 µM suramin. After extensive washing the remaining cell-bound radioactivity was quantified by scintillation counting (n=2). Significant differences compared to untreated infected cells are indicated by * (**, p<0.01; ***, p<0.001).

3.4. Suramin treatment also reduces the release of infectious progeny

Analysis of the infectious virus titers in the supernatant of samples from a time-of-addition experiment revealed that suramin not only affected an early step of the replication cycle, but also reduced progeny titers when added at later stages (Fig. 4A). When suramin treatment was started at 2 h post entry, it had no significant effect on intracellular viral RNA levels,

suggesting it does not interfere with RNA replication (Fig. 2A), while a ~100-fold reduction of infectious progeny released into the medium was observed (Fig. 4A). To confirm that suramin also has a late effect on the release or maturation of viral particles, we treated ZIKV-infected cells with 25-200 μ M of suramin from 14 to 26 h p.i. As expected, the number of infected cells (based on immunofluorescence microscopy) did not change (results not shown) by this late suramin treatment, while there was a dose-dependent decrease of up to 35-fold in the amount of infectious progeny released (Fig. 4B).

To further corroborate this effect on virion biogenesis, a time-of-addition experiment was performed, in which cells where infected and subsequently treated with 200 μ M suramin starting from 2, 14, 18, or 22 h p.i. until the medium was harvested at 26 h p.i. A consistent ~10-fold reduction in the amount of released genome copies was observed, whether suramin treatment was initiated at 2 h p.i. or as late as 22 h p.i. (Fig. 4C, black bars). However a stronger, ~100-fold reduction in released infectious progeny titers was observed when suramin treatment was initiated at 2 h p.i. (Fig. 4C, gray bars). The latter effect was less pronounced (~10-fold) when suramin treatment was initiated at 2 h p.i. (Fig. 4C, gray bars). The latter effect was less pronounced (~10-fold) when suramin treatment was initiated late in the replication cycle (18 or 22 h p.i.). These results suggest that suramin affects both the total number of released virions (genome copies) and their infectivity (PFU).





A. Time-of-addition-assay. ZIKV-infected Vero cells were treated with 200 μ M of suramin from -2 to 26 h p.i. Untreated (Ut) and mock-infected cells were included as controls. At 26 h p.i. supernatants were harvested in triplicate and virus titers were determined by plaque assay. B. ZIKV-infected Vero cells were treated with 25 to 200 μ M suramin from 14 to 26 h p.i. Untreated and mock-infected cells were included as controls. At 26 h p.i., supernatants were harvested from two independent infections performed in duplicate and analyzed by plaque assay (n=4). C. ZIKV-infected Vero cells were treated with 200 μ M of suramin starting at 2, 14, 18, and 22 h p.i. Untreated (Ut) and mock-infected cells were included as controls. At 26 h p.i., supernatants were harvested and virus titers were analyzed by plaque assay (n=3). Viral RNA was isolated from the supernatants and copy numbers were determined by RT-qPCR (n=3). Significant differences compared to untreated infected cells are indicated by * (**, p<0.01; ***, p<0.001).

4. Discussion

Here we demonstrate that the approved drug suramin has anti-ZIKV activity by interfering with viral attachment, as well as the release of infectious progeny from ZIKV-infected cells. In CPE reduction assays suramin had an EC_{50} value of 39.8 μ M and a SI of 48. In this assay, MPA had an EC_{50} of 0.42 μ M, similar to values reported recently by others [6, 8]. In low-

MOI CPE reduction assays MPA appeared to inhibit ZIKV more efficiently than suramin, but in the context of a high-MOI infection, treatment with 2.5 μ M MPA led to a mere 10-fold reduction in the accumulation of intracellular ZIKV RNA and infectious ZIKV titers in the medium (Fig. 1). In contrast, treatment of cells infected at a high MOI with 50 μ M of suramin, while also reducing intracellular ZIKV RNA accumulation by 10-fold, yielded a ~300-fold reduction in the amount of infectious progeny (Fig. 1B). MPA's limited effect on high-MOI ZIKV infections, its immunosuppressive properties and its contraindicated use during pregnancy makes it an unattractive lead to follow.

Time-of-addition experiments and binding experiments with radioactive ³⁵S-labeled ZIKV at 4°C suggested that suramin affects viral attachment to the host cell.

The inhibition of attachment might be due to suramin's effect on virions, as was previously reported for DENV [13] and enterovirus A71 [17], for which suramin was proposed to have a 'neutralizing effect' that blocks the interactions of these viruses with cellular heparan sulfate, and the P-selectin ligand, respectively. Suramin was suggested to interact with surface-exposed positively charged residues in the DENV E protein that are part of the glycosaminoglycan-binding (GAG-binding) sites, thereby interfering with attachment to the host cell [13]. It was recently demonstrated that binding to GAGs likely also plays a role in ZIKV attachment [18].

Besides interacting with the virion, suramin could also have an effect on host cell (co) receptors, as it was shown to abolish the interaction of several cellular receptors with their ligands [19-21]. Binding of suramin to cellular (glyco)proteins, like the proposed ZIKV receptors AXL [3, 22] or glycoprotein TIM1 [23] could interfere with ZIKV attachment. Suramin treatment also inhibited the release of infectious progeny from ZIKV-infected cells, not only by lowering the number of genomes released, but also by reducing the specific infectivity (Fig. 4).

This post-entry effect might be due to binding of the negatively charged suramin to one of ZIKV's RNA-binding proteins, as suramin was shown to inhibit Dengue and hepatitis C virus helicase activity [24, 25], norovirus RNA-dependent RNA polymerase activity [26, 27], reverse transcriptase activity of various retroviruses [28], including HIV [29], and the activity of alphavirus replication complexes [10]. Since we did not observe inhibition of ZIKV RNA synthesis in infected cells, an effect on NS5 appears unlikely. Suramin might inhibit the helicase activity of ZIKV NS3 or might affect packaging by binding to positively charged residues on the capsid protein. This would result in accumulation of genomes in the cell, while virus release would be diminished. The E protein, prM, NS1, NS2A, NS4A or/and NS4B [130] or furin could also be targets of suramin, with consequences for virion assembly, glycosylation of ZIKV proteins and/or maturation of the virion, lowering the infectivity of released virions. ZIKV prM, E, and NS1 proteins contain

potential N-glycosylation sites [31, 32] and glycosylation of the E proteins of ZIKV and West Nile Virus is important for virion maturation and infectivity [33, 34]. Suramin is a highly charged molecule that, complexed with serum albumin, can be taken up by cells through endocytosis. The concentration of albumin determines whether suramin mainly accumulates in the lysosomal compartment or in the Golgi apparatus, mitochondria and nucleolus [35]. The latter might cause a generalized negative effect on the secretory pathway that would also affect virion biogenesis.

Studies on suramin-related compounds revealed that the six sulfonate groups as well as the symmetry of the molecular structure are crucial elements for suramin's anti-CHIKV activity [10]. Addition of sulfonate groups and a higher level of branching increased the compound's anti-enterovirus activity [17], suggesting the molecule can be further optimized.

Because of the increased and geographically expanding incidence, unnoticed infections in combination with sexual transmission, and potentially severe consequences of mother-to-child transmission, ZIKV infection continues to be a major public health concern, for which therapeutics are urgently needed. Suramin has been approved for use in humans and has been used safely and successfully in animal models for treating several virus infections. Future studies should evaluate the efficacy of suramin on ZIKV replication in animal models and determine whether suramin or related compounds have the potential to advance into further clinical development for treatment of ZIKV infection.

Acknowledgments

This work was supported by the European Union's Horizon 2020 Research and Innovation Programme under ZIKAlliance Grant Agreement no. 734548 (MJvH, AT, ICA) and under the Marie Skłodowska-Curie ETN 'ANTIVIRALS' Grant Agreement no. 642434 (KK). Funding sources were not involved in study design, data collection, analysis and interpretation, writing of the report and the decision to submit the manuscript for publication.

Supplemental figures



Figure S1. Protective effect of suramin on viability of ZIKV-infected Vero cells in CPE reduction assay (black squares). The effect of suramin on the viability of uninfected cells (grey circles) was measured to determine the CC_{so} of suramin.

References

- Dick, G.W., S.F. Kitchen, and A.J. Haddow, Zika virus. I. Isolations and serological specificity. Trans R Soc Trop Med Hyg, 1952. 46(5): p. 509-20.
- Miner, J.J. and M.S. Diamond, Zika Virus Pathogenesis and Tissue Tropism. Cell Host Microbe, 2017. 21(2): p. 134-142.
- Meertens, L., et al., Axl Mediates ZIKA Virus Entry in Human Glial Cells and Modulates Innate Immune Responses. Cell Rep, 2017. 18(2): p. 324-333.
- 4. Liu, S., *et al.*, AXL-Mediated Productive Infection of Human Endothelial Cells by Zika Virus. Circ Res, 2016. 119(11): p. 1183-1189.
- Dallmeier, K. and J. Neyts, Zika and Other Emerging Viruses: Aiming at the Right Target. Cell Host Microbe, 2016. 20(4): p. 420-422.
- Adcock, R.S., *et al.*, Evaluation of anti-Zika virus activities of broad-spectrum antivirals and NIH clinical collection compounds using a cell-based, high-throughput screen assay. Antiviral Res, 2017. 138: p. 47-56.
- Xu, M., *et al.*, Identification of small-molecule inhibitors of Zika virus infection and induced neural cell death via a drug repurposing screen. Nat Med, 2016. 22(10): p. 1101-1107.
- Barrows, N.J., *et al.*, A Screen of FDA-Approved Drugs for Inhibitors of Zika Virus Infection. Cell Host Microbe, 2016. 20(2): p. 259-70.
- Cheng, F., J.L. Murray, and D.H. Rubin, Drug Repurposing: New Treatments for Zika Virus Infection? Trends Mol Med, 2016. 22(11): p. 919-921.
- 10. Albulescu, I.C., *et al.*, Suramin inhibits chikungunya virus replication through multiple mechanisms. Antiviral Res, 2015. 121: p. 39-46.
- 11. Ho, Y.J., *et al.*, Suramin Inhibits Chikungunya Virus Entry and Transmission. PLoS One, 2015. 10(7): p. e0133511.
- 12. Henss, L., *et al.*, Suramin is a potent inhibitor of Chikungunya and Ebola virus cell entry. Virol J, 2016. 13: p. 149.
- 13. Chen, Y., *et al.*, Dengue virus infectivity depends on envelope protein binding to target cell heparan sulfate. Nat Med, 1997. 3(8): p. 866-71.
- Ren, P., *et al.*, The approved pediatric drug suramin identified as a clinical candidate for the treatment of EV71 infection-suramin inhibits EV71 infection *in vitro* and *in vivo*. Emerg Microbes Infect, 2014. 3(9): p. e62.
- 15. Kuo, S.C., *et al.*, Suramin treatment reduces chikungunya pathogenesis in mice. Antiviral Res, 2016. 134: p. 89-96.
- 16. Scholte, F.E., *et al.*, Characterization of synthetic Chikungunya viruses based on the consensus sequence of recent E1-226V isolates. PLoS One, 2013. 8(8): p. e71047.
- Nishimura, Y., *et al.*, The Suramin Derivative NF449 Interacts with the 5-fold Vertex of the Enterovirus A71 Capsid to Prevent Virus Attachment to PSGL-1 and Heparan Sulfate. PLoS Pathog, 2015. 11(10): p. e1005184.
- 18. Kim, S.Y., et al., Interaction of Zika Virus Envelope Protein with Glycosaminoglycans. Biochemistry, 2017.
- Wu, Z.S., *et al.*, Suramin blocks interaction between human FGF1 and FGFR2 D2 domain and reduces downstream signaling activity. Biochem Biophys Res Commun, 2016. 477(4): p. 861-7.
- Coffey, R.J., Jr., *et al.*, Suramin inhibition of growth factor receptor binding and mitogenicity in AKR-2B cells. J Cell Physiol, 1987. 132(1): p. 143-8.
- 21. Waltenberger, J., *et al.*, Suramin is a potent inhibitor of vascular endothelial growth factor. A contribution to the molecular basis of its antiangiogenic action. J Mol Cell Cardiol, 1996. 28(7): p. 1523-9.
- 22. Richard, A.S., *et al.*, AXL-dependent infection of human fetal endothelial cells distinguishes Zika virus from other pathogenic flaviviruses. Proc Natl Acad Sci U S A, 2017.
- Tabata, T., *et al.*, Zika Virus Targets Different Primary Human Placental Cells, Suggesting Two Routes for Vertical Transmission. Cell Host Microbe, 2016. 20(2): p. 155-66.

- Basavannacharya, C. and S.G. Vasudevan, Suramin inhibits helicase activity of NS3 protein of dengue virus in a fluorescence-based high throughput assay format. Biochem Biophys Res Commun, 2014. 453(3): p. 539-44.
- Mukherjee, S., *et al.*, Identification and analysis of hepatitis C virus NS3 helicase inhibitors using nucleic acid binding assays. Nucleic Acids Res, 2012. 40(17): p. 8607-21.
- Mastrangelo, E., *et al.*, Structure-based inhibition of Norovirus RNA-dependent RNA polymerases. J Mol Biol, 2012. 419(3-4): p. 198-210.
- 27. Tarantino, D., *et al.*, Naphthalene-sulfonate inhibitors of human norovirus RNA-dependent RNA-polymerase. Antiviral Res, 2014. 102: p. 23-8.
- 28. De Clercq, E., Suramin: a potent inhibitor of the reverse transcriptase of RNA tumor viruses. Cancer Lett, 1979. 8(1): p. 9-22.
- Chandra, P., A. Vogel, and T. Gerber, Inhibitors of retroviral DNA polymerase: their implication in the treatment of AIDS. Cancer Res, 1985. 45(9 Suppl): p. 4677s-4684s.
- 30. Garcia-Blanco, M.A., *et al.*, Flavivirus RNA transactions from viral entry to genome replication. Antiviral Res, 2016. 134: p. 244-249.
- Kuno, G. and G.J. Chang, Full-length sequencing and genomic characterization of Bagaza, Kedougou, and Zika viruses. Arch Virol, 2007. 152(4): p. 687-96.
- 32. Berthet, N., *et al.*, Molecular characterization of three Zika flaviviruses obtained from sylvatic mosquitoes in the Central African Republic. Vector Borne Zoonotic Dis, 2014. 14(12): p. 862-5.
- 33. Li, J., *et al.*, The glycosylation site in the envelope protein of West Nile virus (Sarafend) plays an important role in replication and maturation processes. J Gen Virol, 2006. 87(Pt 3): p. 613-22.
- Mossenta, M., et al., Role of N-glycosylation on Zika virus E protein secretion, viral assembly and infectivity. Biochem Biophys Res Commun, 2017.
- Baghdiguian, S., *et al.*, Intracellular localisation of suramin, an anticancer drug, in human colon adenocarcinoma cells: a study by quantitative autoradiography. Eur J Cancer, 1996. 32A(3): p. 525-32.

Chapter 5

Suramin inhibits chikungunya virus replication by interacting with virions and blocking the early steps of infection

Irina C. Albulescu^{aş}, Leonie White-Scholten^{a*}, Ali Tas^a, Tabitha E. Hoornwegb^{\$}, Salvatore Ferla^c, Kristina Kovacikova^a, Jolanda M. Smit^b, Andrea Brancale^c, Eric J. Snijder^a, Martijn J. van Hemert^a•

^aDept. of Medical Microbiology, Leiden University Medical Center, Leiden, NL. ^bDept. of Medical Microbiology, University Medical Center Groningen, University of Groningen, Groningen, NL. ^cSchool of Pharmacy and Pharmaceutical Sciences, Cardiff University, Cardiff, UK. *Present address* [§]Utrecht University, Faculty of Veterinary Medicine, Utrecht, NL. [°]HollandBIO, Den Haag, NL.

'Corresponding author.

Manuscript for submission to Antiviral Research

Abstract

Chikungunya virus (CHIKV) is a mosquito-transmitted alphavirus that can cause a debilitating disease that is primarily characterized by persistent joint pain. CHIKV has been emerging globally, while neither a vaccine nor antiviral medication is available. Suramin, a drug used to treat parasitic infections, was previously shown to inhibit CHIKV replication. In this study we aimed to obtain more detailed insight into its mechanism of action. We found that suramin interacts with virions and can inhibit virus binding to cells. It also appeared to inhibit post-attachment steps of the infection process, likely by preventing conformational changes of the envelope glycoproteins required for fusion and the progression of infection. Following selection and genotyping of suramin-resistant mutants, the substitutions N5R and H18Q in the E2 glycoprotein were reverse engineered in order to understand their role. Indeed, suramin-resistant viruses carrying these two E2 mutations appeared to be able to overcome the post-attachment inhibitory effect of suramin. Conversely, a virus with a G82R mutation in E2, which renders the virus dependent on the interaction with heparan sulfate for its entry, was more sensitive to suramin than wild-type virus. Using molecular modelling studies, we predicted the potential suramin binding sites on the mature spikes of the chikungunya virion. We conclude that suramin interferes with CHIKV entry by interacting with the E2 envelope protein, which can inhibit attachment and interfere with conformational changes required for fusion.

Keywords: anti-CHIKV treatment, suramin, E2 envelope protein, attachment, fusion, drug repurposing

1. Introduction

Chikungunya virus (CHIKV) is a re-emerging alphavirus that is transmitted by *Aedes* sp. mosquitoes and has caused several large outbreaks in the past 15 years. Before 2013 CHIKV was circulating mainly in Africa and Asia [1], but following its introduction into the Caribbean it has now become endemic in Latin America as well [2]. Acute CHIKV infection is associated with fever, rash, muscle pain, and general malaise. Furthermore the virus often causes a debilitating joint pain that can last for months to years [3]. Prophylactic or therapeutic treatment for CHIKV infections is still not available on the market, and vector control measures do not provide the ultimate solution [4]. Although progress is being made in CHIKV vaccine and antiviral drug development [3, 5-7], we would still be poorly prepared in the face of a new CHIKV epidemic.

We and others have previously shown that the antiparasitic drug suramin inhibits CHIKV replication by targeting an early step in the viral replication cycle [8-10]. Moreover, the compound also inhibits CHIKV replication in a mouse model [11]. These findings show the potential for the (off-label) therapeutic use of suramin for the treatment of chronic chikungunya fever, and possibly also for its prophylactic use during severe CHIKV outbreaks. CHIKV has a 11.8-kb single-stranded (ss) RNA genome of positive polarity, which is capped, poly-adenylated and packaged into an icosahedral nucleocapsid that is surrounded by an envelope containing 80 projections (spikes), each consisting of three E1-E2 (envelope proteins) heterodimers [12]. The E2 protein is involved in the interaction with the host cell and therefore is an important determinant of pathogenicity, cellular tropism [13, 14], and immunogenicity [15, 16]. The viral replication cycle begins with attachment of the virion to the cell surface through interactions with glycosaminoglycans, such as heparan sulfate [17]. Subsequent binding to a receptor, like the recently identified Mxra8 protein [18] will lead to uptake of the virion via receptor-mediated endocytosis [19]. Endosomal acidification causes structural rearrangements in the virion that induce E1 protein-mediated fusion of the viral envelope with the endosomal membrane [20]. This leads to nucleocapsid release and its disassembly in the cytoplasm to liberate the RNA genome. Subsequently, the genome is translated into a polyprotein that is processed into the non-structural proteins nsP1 to 4 that (together with host factors) assemble into membrane-associated replication and transcription complexes (RTCs). The structural proteins (capsid, E3, E2, 6k/TF and E1) are expressed from a subgenomic RNA in the form of a second polyprotein. After autoproteolytic release of the capsid protein, the remainder of the structural polyprotein traffics through the secretory pathway, during which it is cleaved by host cell proteases and undergoes posttranslational modifications like glycosylation. Ultimately, the mature envelope proteins will reach the plasma membrane. Here, interaction between nucleocapsids and the cytoplasmic sides of the envelope proteins are essential for the budding process and the formation of new virus particles [21].

The work presented here provides insight into the mode-of-action of suramin, through specific analysis of virus binding and fusion, and by selecting and characterizing suramin-resistant CHIKV variants, which contained N5R and H18Q mutations in the envelope protein E2. Moreover, we found that a virus with the G82R mutation in E2, which renders the virus dependent on heparan sulfate binding for infectivity and was implicated in attenuation of vaccine strain 181/25 [6], was more sensitive to suramin than wild-type CHIKV. Molecular docking studies provided more insight into suramin's inhibitory activity, since the compound was predicted to bind to virus particles at positions that could interfere with conformational changes in the envelope proteins that need to occur during entry.

2. Materials and Methods

2.1. Cells, compounds, and viruses.

Vero E6 and BHK21 cells were grown in DMEM or BHK medium, respectively, supplemented with 8% fetal calf serum (FCS) and penicillin/streptomycin. BS-C-1 cells were cultured as previously described [22]. Suramin, chloroquine and ammonium chloride were purchased from Sigma and ³H-suramin from Hartmann Analytic. CHIKV LS3 (KC149887) and Semliki Forest virus strain SFV4 (KP699763.1) were launched from full-length cDNA clones. All studies with live CHIKV were performed in biosafety cabinets in BSL-3 facilities.

2.2. Preparation of ³⁵S-labeled viruses and purification of virus stocks.

³⁵S-labeled CHIKV and SFV were produced in Vero E6 cells as described before [23]. To remove non-incorporated label, the cell culture supernatant was subjected to pelleting through a sucrose cushion by ultracentrifugation in a SW41 Ti rotor (at ~200,000g for 2h). Virus pellets were resuspended in 1xTESV buffer (20mM Tris-HCl pH 7.4, 1mM EDTA, 100mM NaCl), before aliquoting and storage at -80°C.

2.3. Virus attachment assay. To study virus binding

³⁵S-CHIKV or ³⁵S-SFV samples (1x10⁴ CPM) were incubated with Vero E6 cells for 1h at 4°C, in the presence or absence of suramin, followed by washing and lysis in 4x dyefree Laemmli sample buffer (LSB). The amount of bound radioactivity in samples was quantified by liquid scintillation counting with a Beckman-Coulter LS6500 Multi-Purpose scintillation counter and Ultima Gold[™] scintillation liquid. The binding of fluorescently labeled CHIKV (DiD-CHIKV) to BS-C-1 cells was assessed by fluorescence microscopy, as previously described [22].

2.4. Bulk fusion assay.

Pyrene-labelled CHIKV and liposomes containing phosphatidylcholine (PC), phosphatidylethanolamine (PE), sphingomyelin and cholesterol in a molar ratio of 1:1:1:1.5 were used in a bulk fusion assay as previously described [19].

2.5. ³H-suramin-virus binding assay.

Purified virus particles prepared in 1xTESV buffer (containing only trace amounts of protein from the culture medium) were incubated with 0.5x10⁶ CPM of ³H-suramin, for 1h at 37°C. The unbound suramin was removed by gel-filtration using P30-Microbiospin columns (BioRad) according to the manufacturer's instructions; virions in the flow-through were lysed in dye-free LSB and bound radioactive suramin was quantified by scintillation counting as described under 2.3.

2.6. Reverse genetics.

The mutations listed in Table 1 were introduced into the pMALS2L (G588A, A979G, G980A, T5645C) and pMALS2R (A8554G, C8595A) plasmids [24] using QuickChange sitedirected mutagenesis (Stratagene, USA). Subsequently, single or combined mutations were transferred to the CHIKV LS3 plasmid by using unique BamHI, XmaI, AgeI, XhoI and SfiI restriction sites. The reverse-engineered mutant CHIKV mutants were launched via *in vitro* transcription (Epicentre) and RNA transfection of BHK-21 cells as described previously [24]. After 24 to 48 h, when extensive cytopathic effects (CPE) had occurred, the virus-containing supernatants were harvested and used to produce passage 1 virus stocks on Vero E6 cells, which were used in subsequent experiments. These virus stocks were verified by Sanger sequencing of the full genome to confirm the presence of the introduced mutations and absence of other mutations. Only in the case of G980A (in nsP1) we observed rapid reversion to the wild-type genotype.

2.7. RNA isolation and RT-qPCR.

Total cellular RNA was isolated by lysing the cells in LiDS/LET as previously described [24] or in TRIpure reagent according to the manufacturer's (Invitrogen) instructions. To measure genome copy numbers an internally controlled TaqMan quantitative RT-PCR assay was used [24].

2.8. Cytopathic effect (CPE) reduction assay.

CPE reduction assays were basically performed as previously described [8], except that an MOI of 0.05 was used and the incubation period was 72h for CHIKV mutants S4.1, S4.3, S5 or 96h for wt CHIKV and mutants S2.1, S2.3, S3, S7, S8 and S9, or 120h for mutants S2.2 and G82R. After performing a colorimetric viability assay, absorption was measured at 450 nm using an EnVision Multilabel Plate Reader (PerkinElmer, US).

2.9. Plaque number reduction assay.

To study virus entry, Vero E6 cells were incubated with approximately 100 PFU of CHIKV for 1h at 37°C in the presence of a range of suramin concentrations. After removing the inoculum, the cell monolayer was washed and overlay medium containing 1.2% Avicel RC-581 (FMC BioPolymer) in DMEM, 2% FCS, 25mM HEPES, and penicillin/streptomycin was added. After three days, the cell monolayers were fixed with 3% formaldehyde in PBS solution and plaques were stained and counted. To study attachment, suramin treatment and virus uptake were done for 1h at 4°C, and 1000 PFU were used in order to detect approx. 100 plaques in the untreated wells, as low temperature diminished virus binding.

2.10. Molecular modelling.

Molecular Operating Environment (MOE) 2018.10 [25] and Maestro [26] software was used. The CHIKV E2-E1 glycoprotein heterodimer (PDB ID 3N42) and trimeric complex (PDB ID 3J2W) were preprocessed using the Schrödinger Protein Preparation Wizard by assigning bond orders, adding hydrogens, and performing a restrained energy minimization of the added hydrogens using the OPLS_2005 force field. The missing residues of E2 (1-6) were manually introduced and the downstream docking processes are described in more detail in the Supplemental information. The electrostatic potential surface was obtained using the Surfaces and Maps tool in MOE after splitting the molecule in multiple chains. Figures were prepared with MOE.

2.11. Statistics.

GraphPad Prism 8 was used as previously described [8, 23] for EC_{50} determination by non-linear regression. The statistical analysis was performed with one-way ANOVA using Dunnett's multiple comparison test (Fig. 1 A, B, C).

3. Results and discussion

3.1. Suramin inhibits viral attachment and fusion by interacting with the chikungunya virion Several experiments were carried out to analyze the impact of suramin on the early events of infection, i.e. virus attachment, internalization, and fusion. ³⁵S or DiD-labeled CHIKV [22] were used in virus binding assays, in the presence or absence of suramin, at 4°C. Active endocytosis does not occur at temperatures below 18°C [27] and therefore suramin is expected to remain in the extracellular environment. Virus attachment to the cell surface probably involves electrostatic interactions with the GAGs or other plasma membrane factors (receptors, adhesion molecules) and infections can be synchronized by placing cells at 4°C. By directly measuring the amount of radioactively- or fluorescently-labeled virus in the presence of increasing concentrations of compound, we found that suramin inhibits CHIKV attachment (Fig. 1A and B) in two distinct cell lines and with two different experimental readouts. Binding of ³⁵S-labelled SFV was even more strongly inhibited by suramin (supplemental figure S1A), which is in line with suramin's previously observed lower EC₅₀ for SFV compared to CHIKV [8]. In addition, suramin (30 µM) completely inhibited pyrene-labeled CHIKV from fusing with liposomal membranes at pH ~5.5 in the absence of host proteins, as shown in figure 1C [19]. Because suramin is known to bind to positive charges on the surface of proteins, we suspect that suramin binds directly to the envelope proteins, thereby preventing the conformational changes that are required for fusion. In contrast to Ho et al., who studied the effect of suramin on surface-expressed envelope proteins in insect cells that were subsequently triggered to fuse at low pH, we studied the effect of suramin on membrane fusion in the context of whole virions. To confirm the interaction of suramin with virions, we incubated purified CHIKV (lacking serum proteins as these are known to strongly bind suramin; [28] with ³H-labeled suramin. Compared to a control supernatant from mock-infected cells (which bound 3000 CPM of suramin) purified CHIKV bound over 30,000 CPM of ³H-suramin (Fig. 1D), confirming that the compound does interact with virus particles. ³H-suramin also interacted with SFV (Fig. S1C), and more specifically with the (native) E proteins on the surface of intact virions, since treatment with proteinase K or heat denaturation severely decreased the quantity of bound radiolabeled suramin. For other viruses like enterovirus A71 (EV-A71) and human immunodeficiency virus 1 (HIV-1) it was also found that suramin blocked their access to cellular receptors by directly interacting with virions [29, 30].



Figure 1. The effect of suramin on the early events of CHIKV infection. (A) The binding of ³⁵S-labelled CHIKV to Vero E6 cells in the presence or absence of suramin was determined at 4°C by scintillation counting of remaining radioactivity in cellular lysates obtained after extensive washing (average +/- SD; n=3). (B) Binding of DiD-labeled CHIKV to suramin-treated BS-C-1 cells, analyzed by fluorescent microscopy, in the presence of increasing concentrations. (C) Fusion of pyrene-labeled CHIKV in a bulk fusion assay with liposomes, triggered by lowering the pH, in the presence of increasing suramin concentrations (n=5 and 3, for untreated and treated samples, respectively). (D) Binding of ³H-labeled suramin to CHIKV (purified virus was used to exlude interference by serum proteins). The control used in this assay was culture medium from uninfected cells that was treated the same way as when purifying virus (n=3). The data represent the means ± the SD and significant differences are indicated with * (****p < 0.001, ***p < 0.005, **p < 0.01, *p < 0.05 and ns as not significant).

Surprisingly, when SFV particles were pre-attached to cells (at 4°C, in the absence of compound) and treated with suramin at 37°C, only a modest ~20% inhibition in virus uptake was observed, regardless of the suramin concentrations tested (Fig. S1B). Hence it is possible that at physiological temperature, pre-attached viruses cannot be displaced by the compound or the entry process might be too fast to be inhibited by suramin. When evaluating these direct binding studies, it should be noted that it remains unclear what fraction of the bound virions will actually lead to a productive infection.

The inhibitory effect of suramin on the binding of virus particles to the cell surface could be due to its direct interaction with virions and/or with cellular receptors.

The direct binding assays with radiolabeled or fluorescently labeled viral particles clearly showed that suramin inhibits binding/fusion of both CHIKV and SFV (Fig. 1 and S1). This is in contrast with our earlier study [8], in which we concluded that suramin inhibits an early step, but not attachment. In that study, we relied on an RT-qPCR assay to quantify cell-bound CHIKV. However, this is an indirect measurement and the majority of the detected RNA molecules do not represent infectious particles, as we and others found that genome copy to PFU ratios of commonly used virus stocks are generally over 1000:1 [31, 32]. At least in part, this is due to the fact that virus stocks are generally harvested when extensive cytopathic effect (CPE) has occurred, which leads to the release of viral RNA not associated with infectious virus particles, e.g. in the form of naked RNA, nucleoprotein complexes or nucleocapsids. Therefore, we no longer consider RT-qPCR an appropriate assay to study CHIKV binding when standard, non-purified CHIKV specimens are used.

To evaluate whether the effect of suramin on CHIKV binding could be reliably measured by an improved RT-qPCR-based assay, we have used a PEG-precipitated virus stock to improve the CHIKV RNA to PFU ratio. After titration on Vero E6 cells, the ratio ranged from 7632:1 in the non-purified stock to 68:1 in the PEG-precipitated stock. Clearly, even after this procedure, the number of CHIKV RNA molecules still strongly outnumbers the number of infectious particles. Subsequently, we repeated the qPCR-based measurement of virus binding at 4°C in the presence and absence of suramin (Fig. S2), for both nonpurified and PEG-precipitated virus stocks. While for the former there was an unexpected increase in detected CHIKV RNA in the presence of suramin, the PEG-precipitated stock showed a slight decrease, confirming that the quality of the virus stock strongly influences the experimental outcome and conclusions.

3.2. Suramin-resistant CHIKV variants acquired mutations in the envelope protein E2

Repeated passaging of CHIKV in the presence of increasing suramin concentrations that do not fully inhibit replication (from 25 up to 300 μ M) yielded variants that could grow to titers above 105 PFU/mL in the presence of 150 μ M suramin. By passage 5 (P5) and passage 7, drug concentrations of 150 and 300 μ M, respectively, were tolerated, concentrations that reduced wild-type virus titers by at least 2 logs [8]. Genotyping of the P5 virus revealed the

presence of three nonsynonymous mutations in the CHIKV nsP1, nsP3 and E2 proteins. Two passages later (P7), the suramin-resistant variant had acquired several additional mutations in the same 3 proteins (Table 1).

Mutation in	At P5 (150 µM suramin)		At P7 (300 μM suramin)		
	Nucleotide substitution	Amino acid mutation	Nucleotide substitution	Amino acid mutation	
nsP1	G588A	R171Q	G588A A979G G980A	R171Q T301K G302R	
nsP3	U5645C	opal524R			
E2	A8554G	N5R	A8554G C8595A	N5R H18Q	

Table 1. Mutations in CHIKV resulting from serial passaging in the presence of increasing (suboptimal) concentrations of suramin.

To pinpoint which of these mutations is/are responsible for the suramin-resistant phenotype, they were all reverse engineered into our CHIKV full-length cDNA clone [24], either alone or in combination. For each of the reverse-engineered viruses, the plaque phenotype and sensitivity to suramin were determined and for several mutants the growth kinetics were also compared.

Viruses with the R171Q and opal524R mutations (CHIKV mutants S4.1, S4.3 and S5) were found to produce larger plaques on Vero E6 cells, suggesting they could be linked to cell culture adaptation. The apparent 'suramin resistance' (a maximum EC_{50} increase of 1.5-fold) of some of these mutants was more likely due to their accelerated growth and increased CPE, rather than to a specific resistance to the compound (Fig. 3A). Both R171Q and opal524R had been previously reported in CHIKV isolates such as MADOPY1, StBI and StVE (GenBank accession numbers KP003808.1, KP003811.1, and KP003810.1 respectively) which all had been passaged in cell culture prior to sequencing [33]. The mutation opal524R in the nsP3-coding region was reported also by Mounce et al. in the context of resistance selection against the compound DFMO [34], alongside with other nsP mutations. In the case of CHIKV and other alphaviruses (SFV, ONNV), evolutionary pressures have maintained both variants (stop and arginine codon) as it offers a fitness advantage when switching between vertebrate and invertebrate hosts [35-37]. The combination of the R171Q and opal524R mutations was also found independently in other CHIKV isolates (Kovacikova et al, manuscript in preparation), and it seems that these merely reflect adaptation to repeated passaging in mammalian cells.

In contrast, the T301K mutation in nsP1, causing a small-plaque phenotype, and the two mutations in E2 (N5R and H18Q) that did not alter the plaque phenotype (Fig. 2A), have to our knowledge not been previously identified in other isolates. However, the T301 residue was found to be changed to an I in the CHIKV 181/25 vaccine strain, that was attenuated by repeated

passaging in MRC-5 cells, and was accompanied by mutations T12I and G82R in E2, C42F in 6K and A404V in E1 [38]. It was later proven that T301I was not related to the attenuated phenotype of the vaccine strain, but that the two E2 mutations were responsible [39].

Reverse engineering of the N5R and H18Q mutations in E2 produced CHIKV variants with an increased tolerance to suramin (Fig. 2A), as an increase in EC_{50} was observed when both mutations were present (S5 and S9 in Fig. 2A). This suggested that the two E2 mutations cause the suramin-resistant phenotype, identifying E2 as the target of the compound.



Figure 2. Characterization and suramin sensitivity of reverse engineered CHIKV variants. (A) Mutations identified in suramin-resistant CHIKV mutants (Table 1) were reverse-engineered (individually or in combinations) into infectious cDNA clone CHIKV LS3. Plaque morphology (in the absence of suramin) and EC_{50} (mean, n=4) for suramin are shown for each of the recombinant viruses. The values were determined from at least two independent experiments. For S2.2, the EC_{50} could not be determined because in this assay the mutant did not cause CPE with the same kinetics as the other strains. (B), (C) Replication kinetics of CHIKV mutants S4, S5, S9 and wt virus were compared during infection of Vero E6 cells in the absence (B) or presence of 0.2 mM suramin (C). At several time-points p.i., culture supernatants were harvested and infectious virus titers were determined by plaque assay (n=2). (D) Side-by-side comparison of the 36h p.i. titers of various mutants and wt virus grown in the absence (N.T., not treated) or presence of 0.2 mM suramin (n=2). All experiments were performed in Vero E6 cells and the data represent mean ± the SD.

The replication kinetics of relevant mutants, S4.3 (containing mutations in nsPs only), S9 (containing only the E2 mutations), and S5 (containing all mutations) were slightly faster than that of the wt control, reaching slightly higher titers by 24 h p.i., but similar titers by 36 h p.i. (Fig. 2B). Despite its faster replication, S4 is as sensitive to suramin treatment as the wt virus, with a 3-log reduction in titers upon treatment with 200 μ M suramin. The S9 and S5 mutants, on the other hand, exhibited a titer reduction of only 1 log (Fig. 2C and D). These findings indicate that the nsP mutations (the S4 variant) were not involved in suramin resistance and that the E2 mutations alone were responsible for the resistant phenotype.

3.3. The N5R and H18Q mutations in E2 enhance CHIKV entry

To understand how the E2 mutations affect CHIKV infectivity in the presence of suramin, we analyzed the early steps (attachment, uptake, fusion) of infection, using a plaque number reduction assay in which virus and compound were present only during the first hour. We also reverse engineered a virus with a G82R mutation in E2, as this mutation was previously shown to render CHIKV fully dependent on HS binding for infection and is responsible for the attenuated phenotype of CHIKV vaccine strain 181/25 [13, 40]. Among all E2 mutant viruses tested, the G82R mutant was extremely sensitive to suramin, as its uptake and infectivity were almost abolished in the presence of 50 μ M suramin, whereas this concentration merely caused a 40% reduction for wt virus (Fig. 3A). The S7 and S9 mutants were indeed more resistant to suramin than wt virus, as their entry was less inhibited by the presence of 50 μ M suramin. In the presence of 200 μ M suramin, the uptake of wt virus, mutant S7 and mutant S8 was reduced to 40% of that of untreated cells, while the uptake of mutant S9 remained at $\sim 60\%$, confirming that the presence of both E2 mutations leads to an increased resistance. The binding of mutant S9 at 4°C, however, was inhibited to the same extent by suramin as that of wt virus, suggesting the N5R and H18Q mutations do not offer resistance to the compound during the attachment step (Fig. 3B). This finding was corroborated by a time-of-addition assay where the pre-treatment of cells with suramin inhibited wt and mutant S9 to the same extent. However, if treatment was started at the moment of infection or later, the S9 mutant replicated better than the wt virus (supplemental figure S3). Therefore, it seems that the two mutations allow the virus to overcome the inhibitory effect of suramin at a post-attachment step, such as fusion.



Figure 3. The effect of E2 mutations on suramin-resistance and the early steps of infection. (A) Virus uptake and infectivity were determined based on a PRNT-like assay. Approx. 100 PFU of wt CHIKV and mutants S7, S8, S9 and G82R were incubated with Vero E6 cells for 1h in the presence or absence of increasing suramin concentrations. Afterwards the inoculum was removed, the monolayers were washed with PBS and overlay medium was added. After a 3-day incubation, the cells were fixed and plaques were stained and counted. (B) The plaque number reduction assay was used to analyze synchronized attachment of wt CHIKV and variant S9 at 4° C, in the presence and absence of suramin. After binding for 1h in the cold, the inoculum and suramin were removed and replaced with overlay medium without suramin, and the rest of the procedure was performed as described under (A). The data represent the means \pm the SD (n=3).

3.4. Molecular modelling predicts suramin to bind between two adjacent E2 proteins in a mature spike

The observations on the involvement of E2 mutations in suramin resistance prompted us to explore the interaction between the viral envelope protein and compound in more detail, by using a molecular docking approach. We employed a molecular docking approach using suramin and the E2 protein structure on its own, in the form of the E1-E2 heterodimer or as present within a mature CHIKV spike. Our initial attempts of docking suramin to an isolated E2 protein structure or to the E1-E2 heterodimer (PDB ID 3N42) revealed that the compound could interact with a region lacking a clear secondary structure (a groove between domains A and C and linked to the flexible N-terminal part of E2). The N5R mutation that was implicated in suramin resistance maps to this flexible region in the N-terminal domain of E2. Based on this predicted binding, the introduction of negative charges at position 6 and 160 (F6D, T160D) were expected to repel the binding of suramin to that area. To test whether this was indeed the case, these mutations were reverse engineered. The recombinant viruses had a plaque size similar to the wt CHIKV, but in CPE reduction assays they were more or equally sensitive to suramin, with EC_{50} of 54.5 μ M (for F6D) and 16.7 µM (for T160D), compared to 55.5 µM of wt CHIKV. Previously Ho et al. have also docked suramin to the E1-E2 heterodimer and predicted that suramin would bind in a region between the two proteins [9].

In a more recent publication, a structure for the mature CHIKV spike was obtained by modeling the crystal structures of the E1 and E2 proteins into the cryo-EM image of a CHIKV VLP [41]. Due to this novel information, we now consider the earlier docking

studies to be less representative for the natural situation, since in the context of a virion the suggested suramin binding sites would not be exposed in mature spikes, which are formed by E1-E2 heterotrimers. Therefore, to refine our understanding of the possible electrostatic interactions between suramin and the surface of CHIKV, we employed a molecular docking approach based on the more relevant model of the envelope heterotrimer.

In this model, suramin (depicted in yellow) interacts with a very flexible loop in the N-terminal region of one E2 molecule, while it extends towards the middle of domain A of an adjacent E2 (Fig. 4A). Moreover, several suramin molecules are predicted to bind the same spike. Because domain A is involved in receptor recognition, as well as being the target of neutralizing antibodies, it has a pivotal role in the viral replicative cycle [14, 16, 18, 42].

By analyzing the charge distribution on the surface of the CHIKV spike, it became clear that the N-terminal loop of E2 harbors positive charges (K at positions 3 and 10). By acquiring the mutation N5R, there was a clear increase in the positive charges (differences indicated with black rectangles in Fig. 4C and D), probably leading to a stronger interaction with suramin and perhaps pulling it away from another area of the spike. Counterintuitively, because one would expect that a resistance mutation would prevent the interaction with suramin, we speculate that the N5R mutation actually attracts the compound thereby changing the binding mode. Perhaps the N5R mutation directs suramin away from the center of domain A, which is known to be involved in receptor recognition and is also the target of neutralizing antibodies [15, 18].

According to our molecular modelling prediction, the H18Q mutation, which is located in a region across from the E1 fusion loop, might enhance the effect of the N5R mutation by stabilizing the flexible N-terminal loop to achieve a better interaction with the compound or could cause other unpredicted structural changes in the heterotrimer leading to a decreased affinity for suramin. Additionally, H18Q could also facilitate fusion by aiding in the conformational rearrangements required for exposure of the fusion loop. Combined, the N5R and H18Q mutations might change the binding geometry of suramin, perhaps sidetracking it from the core of the E1/E2 heterodimer/spike and/or facilitating fusion (Fig. 4C).

Our hypothesis that suramin is attracted to the center of domain A is further supported by the observation that the CHIKV variant that is completely dependent on heparan sulfate interactions for infectivity [13, 40] has a G82R mutation in E2. In CPE-based assays, the G82R mutant is more sensitive to suramin ($EC_{50} < 15 \mu$ M) than wt CHIKV. Interestingly, residue G82 is located at the center of the spike, where the A domains of the three E2 subunits are found (highlighted in red in Fig. 4A), and maps to the area that interacts with MXRA8,
a receptor which was recently found to promote CHIKV entry [18]. The introduction of a positive charge at position 82 of E2 could increase the affinity for suramin at the center of the spike and block cell attachment.

These results suggest that suramin has more impact on the infection when it is attracted to the center of the spike, while more distant positive charges could direct it away from this important region, allowing attachment and interaction with specific receptors, which could explain why the observed the suramin-resistance mutations emerged.



Figure 4. Molecular docking of suramin to a mature CHIKV spike. (A) The top view of a full wt E1-E2 heterotrimer (PDB ID 3J2W). The E2 proteins are represented as blue ribbon, the E1 as purple ribbon and the fusion loop as orange ribbon; the N5 and H18 residues are represented with carbon atoms in blue and green, respectively, and residue G82 with carbon atoms in red, belong to E2. Suramin is represented in yellow (3 molecules, carbon atoms and molecular surface). In the right black rectangle there is a clearer view of the spike core. (B) Electrostatic potential (Coulombic surface coloring) of the heterotrimer of wt CHIKV. The black rectangle marks the N-terminal domain of one E2 protein, where the positive charges are found. (C) Electrostatic potential (Coulombic surface coloring) of the heterotrimer of the N5R/H18Q mutant, CHIKV S9. The black rectangle highlights the N-terminal domain of E2 showing an increase in positive charges (blue molecular surface). For presentation purposes, the transmembrane and C-terminal segments of the E1 and E2, which interact with capsid proteins seen in (A), were removed.

4. Conclusions

The aim of our study was to understand how suramin inhibits the early steps of the replicative cycle of CHIKV and other alphaviruses. We have shown that suramin can interact with CHIKV *in vitro*, and inhibits attachment of the virus to the host cell. Moreover, it could prevent conformational rearrangements in the viral spike glycoproteins that are required for fusion. We were able to select suramin-resistant CHIKV variants and demonstrated that the N5R and H18Q mutations in E2 were responsible for resistance. These mutations did not offer the virus a major advantage during the binding to cells in the presence of suramin. The benefit of these mutations appears to play a role in overcoming suramin's inhibitory effects during later stages of entry, perhaps allowing the suramin-bound (mutant) spikes to undergo conformational changes required for fusion and progression of the infection. Although CHIKV is able to acquire resistance mutations to the compound, suramin is still an interesting drug candidate as the level of resistance is rather low and required repeated passaging. Additionally, suramin protected human primary dermal fibroblasts from CHIKV-induced CPE with an EC₅₀ of approx. 95 μ M , proving its efficiency in a more relevant cell model for arbovirus infection.

Regarding its use in humans, suramin could be explored as prophylactic in the context of an outbreak, since it is a compound with one of the longest known half-lives in humans [43]. In previous clinical studies, concerning the treatment of AIDS and certain types of cancer, multiple and serious side effects were attributed to the long-term use of suramin [44]. However, this concerned seriously ill patients and long-term treatment. Such an extended regimen is not required for the treatment of parasitic infections for which suramin has proven to be effective and much better tolerated. Therefore, we believe that also for the short-term treatment or prevention of chikungunya virus infections, suramin would be an interesting drug to evaluate.

5. Acknowledgements

We thank Laura A. Wolters for technical assistance and Dr. Igor Sidorov for bioinformatics advice. We would also like to thank Dr. Adriaan H. de Wilde for fruitful discussions and feedback. SF and AB acknowledge support from the Life Science Research Network Wales grant no. NRNPGSep14008, an initiative funded through the Welsh Government's Ser Cymru program. ICA, EJS, MJvH acknowledge the support of the EU-FP7 EUVIRNA grant (No. 264286).

Appendix 1 - Supplemental materials and methods

Molecular modeling – in depth description of section 2.10

All molecular docking studies were performed on a Viglen Genie Intel®CoreTM i7-3770 vPro CPU@ 3.40 GHz x 8 running Ubuntu 14.04. Molecular Operating Environment (MOE) 2018.10 [25] and Maestro [26] were used as molecular modelling software. The CHIKV E2-E1 glycoprotein heterodimer and trimeric complex structures were downloaded from the PDB data bank (http://www.rcsb.org/; PDB code 3N42 and 3J2W). The proteins were preprocessed using the Schrödinger Protein Preparation Wizard by assigning bond orders, adding hydrogens and performing a restrained energy minimization of the added hydrogens using the OPLS 2005 force field. The missing N-terminal residues at the start of the E2 protein (residues 1-6) were manually introduced and energy minimized. Suramin structure was built with MOE and then prepared using the Maestro LigPrep tool by energy minimizing the structures (OPLS_2005 force filed), generating possible ionization states at pH 7±2, generating tautomers and low-energy ring conformers. SiteMap tool in Maestro was used to individuate a potential binding area in proximity of the important mutations, N5R and H18Q. The selected area was used as binding site for the molecular docking studies. For the trimeric complex, three different, one for each heterodimer, 36 Å docking grids (inner-box 10 Å and outer-box 46 Å) were prepared using as centroid a threonine residue localized at the spatial center between the N5 and H18 of two adjacent E2 subunits. In the case of the isolated E2 unit, the same binding area for grid generation was defined selecting as centroid a dummy atom, manually positioned at equal distance from N5 and H18. Molecular docking studies were performed using Glide SP precision keeping the default parameters and setting 5 as number of output poses per input ligand to include in the solution. The output database was saved as mol2 file. The docking results were visually inspected for their ability to bind the active site. The electrostatic potential surface was obtained using the Surfaces and Maps toll in MOE after splitting the molecule in multiple chains. Figures were prepared with MOE.

Appendix 2 - Supplemental figures S1, S2, S3



Figure S1. Effect of suramin on the early events of SFV infection. (A) Binding of ³⁵S-labelled SFV to Vero E6 cells, at 4°C, in the presence of 4-fold serial dilutions of suramin. (B) after a 30-min pre-attachment of virus at 4°C in the absence of the compound, the unbound particles were removed by washing with PBS. Subsequently, medium with suramin was added to the cells followed by incubation at 37°C for 30 min. Cell-associated radiolabeled virus was quantified by cell lysis and liquid scintillation counting. (C) ³H-suramin was incubated with purified SFV (in the absence of serum proteins) followed by treatment with proteinase K for 15 min at 37°C or by heat denaturation for 5 min at 95°C, followed by quantification of suramin binding as described for panel B. The average +/-SD is shown; n=2.



Figure S2. Effect of suramin on CHIKV uptake analyzed by RT-qPCR. The effect of treatment with 0.5 mM suramin during viral uptake (MOI=1) using either a non-purified, standard CHIKV stock (Stand. stock) or a PEG-precipitated stock of the same virus (PEG prec.). After 1h p.i. at 4°C the compound and inoculum (which had been added simultaneously) were washed away with PBS and the CHIKV RNA copy numbers in the cell lysates were analyzed by RT-qPCR (n=2).



Figure S3. The effect of E2 mutations on suramin-resistance analyzed through a time-of addition assay. Replication of wt and S9 in the presence of 400 μ M suramin, added at various time points prior and after the start of the infection, was determined by measuring the titer of infectious CHIKV at 10 h p.i.

References

- Weaver, S.C. and M. Lecuit, Chikungunya virus and the global spread of a mosquito-borne disease. N Engl J Med, 2015. 372(13): p. 1231-9.
- Weaver, S.C. and N.L. Forrester, Chikungunya: Evolutionary history and recent epidemic spread. Antiviral Res, 2015. 120: p. 32-9.
- Silva, L.A. and T.S. Dermody, Chikungunya virus: epidemiology, replication, disease mechanisms, and prospective intervention strategies. J Clin Invest, 2017. 127(3): p. 737-749.
- von Seidlein, L., A.S. Kekule, and D. Strickman, Novel Vector Control Approaches: The Future for Prevention of Zika Virus Transmission? PLoS Med, 2017. 14(1): p. e1002219.
- Tharmarajah, K., S. Mahalingam, and A. Zaid, Chikungunya: vaccines and therapeutics. F1000Res, 2017.
 6: p. 2114.
- Silva, J.V.J., Jr., *et al.*, Current status, challenges and perspectives in the development of vaccines against yellow fever, dengue, Zika and chikungunya viruses. Acta Trop, 2018. 182: p. 257-263.
- Ching, K.C., F.P.N. L, and C.L.L. Chai, A compendium of small molecule direct-acting and host-targeting inhibitors as therapies against alphaviruses. J Antimicrob Chemother, 2017. 72(11): p. 2973-2989.
- 8. Albulescu, I.C., *et al.*, Suramin inhibits chikungunya virus replication through multiple mechanisms. Antiviral Res, 2015. 121: p. 39-46.
- 9. Ho, Y.J., *et al.*, Suramin Inhibits Chikungunya Virus Entry and Transmission. PLoS One, 2015. 10(7): p. e0133511.
- 10. Henss, L., *et al.*, Suramin is a potent inhibitor of Chikungunya and Ebola virus cell entry. Virol J, 2016. 13: p. 149.
- 11. Kuo, S.C., *et al.*, Suramin treatment reduces chikungunya pathogenesis in mice. Antiviral Res, 2016. 134: p. 89-96.
- 12. Voss, J.E., *et al.*, Glycoprotein organization of Chikungunya virus particles revealed by X-ray crystallography. Nature, 2010. 468(7324): p. 709-12.
- Silva, L.A., *et al.*, A single-amino-acid polymorphism in Chikungunya virus E2 glycoprotein influences glycosaminoglycan utilization. J Virol, 2014. 88(5): p. 2385-97.
- Weger-Lucarelli, J., *et al.*, Dissecting the Role of E2 Protein Domains in Alphavirus Pathogenicity. J Virol, 2015. 90(5): p. 2418-33.
- 15. Fox, J.M., *et al.*, Broadly Neutralizing Alphavirus Antibodies Bind an Epitope on E2 and Inhibit Entry and Egress. Cell, 2015. 163(5): p. 1095-1107.
- Jin, J., *et al.*, Neutralizing Monoclonal Antibodies Block Chikungunya Virus Entry and Release by Targeting an Epitope Critical to Viral Pathogenesis. Cell Rep, 2015. 13(11): p. 2553-2564.
- Tanaka, A., *et al.*, Genome-Wide Screening Uncovers the Significance of N-Sulfation of Heparan Sulfate as a Host Cell Factor for Chikungunya Virus Infection. J Virol, 2017. 91(13).
- Zhang, R., *et al.*, Mxra8 is a receptor for multiple arthritogenic alphaviruses. Nature, 2018. 557(7706): p. 570-574.
- van Duijl-Richter, M.K., *et al.*, Early Events in Chikungunya Virus Infection-From Virus Cell Binding to Membrane Fusion. Viruses, 2015. 7(7): p. 3647-74.
- 20. Li, L., *et al.*, Structural changes of envelope proteins during alphavirus fusion. Nature, 2010. 468(7324): p. 705-8.
- Kuhn, R.J., Fields' Virology (Chapter 22 Togaviride), in Fields' Virology, U.L.W.W. D. M. Knipe & P. M. Howley. Philadelphia, Editor. 2013. p. 629-650.
- 22. Hoornweg, T.E., *et al.*, Dynamics of Chikungunya Virus Cell Entry Unraveled by Single-Virus Tracking in Living Cells. J Virol, 2016. 90(9): p. 4745-4756.
- 23. Albulescu, I.C., *et al.*, Suramin inhibits Zika virus replication by interfering with virus attachment and release of infectious particles. Antiviral Res, 2017. 143: p. 230-236.
- 24. Scholte, F.E., *et al.*, Characterization of synthetic Chikungunya viruses based on the consensus sequence of recent E1-226V isolates. PLoS One, 2013. 8(8): p. e71047.

- 25. Chemical Computing Group, I. Molecular Operating Environment (MOE 2018.10). 2018; Available from: http://www.chemcomp.com.
- 26. Schrödinger, L. Schrödinger Release 2017-1. 2017; Available from: https://www.schrodinger.com/ citations/#Maestro.
- 27. Tomoda, H., Y. Kishimoto, and Y.C. Lee, Temperature effect on endocytosis and exocytosis by rabbit alveolar macrophages. J Biol Chem, 1989. 264(26): p. 15445-50.
- 28. Hawking, F., Suramin: with special reference to onchocerciasis. Adv Pharmacol Chemother, 1978. 15: p. 289-322.
- 29. Ren, P., *et al.*, Suramin interacts with the positively charged region surrounding the 5-fold axis of the EV-A71 capsid and inhibits multiple enterovirus A. Sci Rep, 2017. 7: p. 42902.
- Yahi, N., *et al.*, Suramin inhibits binding of the V3 region of HIV-1 envelope glycoprotein gp120 to galactosylceramide, the receptor for HIV-1 gp120 on human colon epithelial cells. J Biol Chem, 1994. 269(39): p. 24349-53.
- Coffey, L.L. and M. Vignuzzi, Host alternation of chikungunya virus increases fitness while restricting population diversity and adaptability to novel selective pressures. J Virol, 2011. 85(2): p. 1025-35.
- Ashbrook, A.W., *et al.*, Residue 82 of the Chikungunya virus E2 attachment protein modulates viral dissemination and arthritis in mice. J Virol, 2014. 88(21): p. 12180-92.
- 33. Moyen, N., *et al.*, First reported chikungunya fever outbreak in the republic of Congo, 2011. PLoS One, 2014. 9(12): p. e115938.
- Mounce, B.C., *et al.*, Chikungunya Virus Overcomes Polyamine Depletion by Mutation of nsP1 and the Opal Stop Codon To Confer Enhanced Replication and Fitness. J Virol, 2017. 91(15).
- Rupp, J.C., *et al.*, Alphavirus RNA synthesis and non-structural protein functions. J Gen Virol, 2015. 96(9): p. 2483-500.
- Myles, K.M., *et al.*, Effects of an opal termination codon preceding the nsP4 gene sequence in the O'Nyong-Nyong virus genome on Anopheles gambiae infectivity. J Virol, 2006. 80(10): p. 4992-7.
- 37. Jones, J.E., *et al.*, Disruption of the Opal Stop Codon Attenuates Chikungunya Virus-Induced Arthritis and Pathology. MBio, 2017. 8(6).
- Levitt, N.H., *et al.*, Development of an attenuated strain of chikungunya virus for use in vaccine production. Vaccine, 1986. 4(3): p. 157-62.
- Gorchakov, R., *et al.*, Attenuation of Chikungunya virus vaccine strain 181/clone 25 is determined by two amino acid substitutions in the E2 envelope glycoprotein. J Virol, 2012. 86(11): p. 6084-96.
- 40. Gardner, C.L., *et al.*, Deliberate attenuation of chikungunya virus by adaptation to heparan sulfatedependent infectivity: a model for rational arboviral vaccine design. PLoS Negl Trop Dis, 2014. 8(2): p. e2719.
- 41. Sun, S., *et al.*, Structural analyses at pseudo atomic resolution of Chikungunya virus and antibodies show mechanisms of neutralization. Elife, 2013. 2: p. e00435.
- 42. Weber, C., *et al.*, Identification of Functional Determinants in the Chikungunya Virus E2 Protein. PLoS Negl Trop Dis, 2017. 11(1): p. e0005318.
- World Health Organization. Control and surveillance of human African trypanosomiasis. World Health Organ Tech Rep Ser 2013 984]; 1-237]. Available from: http://www.ncbi.nlm.nih.gov/pubmed/24552089.
- 44. van Genderen, P.J.J., Side Effects of Drugs Annual. 2005. p. 346-355.

Chapter 6

General discussion

CHIKV is a serious human pathogen for which antiviral drugs are still not available. The research described in this thesis aimed to advance our knowledge on CHIKV replication and to contribute to the development of much-needed inhibitors of CHIKV infection. Following the development of an *in vitro* assay to study CHIKV replication, this tool was used to study the mode of action (MoA) of antiviral compounds (chapter 2) and suramin was identified as a potent inhibitor of viral RNA synthesis. However, we discovered that in cell culture, suramin's antiviral activity was mainly due to inhibition of CHIKV binding/ entry, and to a lesser extent virus release (chapters 3 and 5). Suramin was also found to inhibit binding/entry and virion biogenesis of Zika virus (ZIKV), a recently emerged flavivirus that caused massive epidemics and serious manifestations, such as malformations in newborns and Guilain-Barre syndrome in adults (chapter 4). Due to its ability to form electrostatic interactions with positive charges on proteins, suramin may block the contact between virions and their (co)receptors, by interacting with either virus or receptor, or with both. In chapter 5, using radioactively-labelled suramin, it was clearly shown that the compound interacts with CHIKV and SFV particles, more specifically with their envelope proteins. Additionally, suramin could interfere with cell attachment and/or the structural changes required for fusion. Suramin-resistant CHIKV variants were selected, which contained mutations in the E2 envelope protein (involved in receptor interactions), supporting the idea that suramin blocks the early steps of the infectious cycle. Below, these findings are further discussed in the context of findings related to other viruses that are inhibited by suramin.

1. An in vitro system for CHIKV replication and its applications

The goal of this research project was to gain more insight into the replication mechanism of CHIKV and to develop assays for identifying inhibitors of CHIKV RNA synthesis.

To develop *in vitro* assays for CHIKV RNA synthesis, two approaches were followed. The first concerned the reconstruction of active complexes using recombinant nsP4, the viral RNA polymerase, expressed in and purified from bacteria, and a preparation of nsP123 isolated from mammalian cells. The second approach concerned the (semi)purification of membrane-associated viral replication and transcription complexes (RTCs) from CHIKV-infected cells. The activity of such complexes was tested in assays that measure the incorporation of radiolabeled CTP into viral RNA products.

The first approach seemed promising as it had been described for Sindbis virus (SINV) nsP4. Initially Tomar and collaborators were able to purify only D97nsP4 (the core catalytic domain), because the full-length protein was prone to degradation. However, the truncated protein only showed *in vitro* terminal adenylyl-transferase (TATase) activity (1). The N-terminal domain of nsP4 is very important for interactions with the other nsPs to control RNA synthesis (1, 2), but the purification of full-length nsP4 is challenging because of its instability, which is caused by the N-terminal Tyr residue that renders it a target for rapid

proteasomal degradation via the N-end rule pathway (3). These issues could potentially be avoided by expressing nsP4 N-terminally linked to another protein that can be removed after the purification step. Using 6xHis-SUMO as the N-terminal tag, Rubach *et al.* were successful in purifying full-length SINV nsP4 in significant amounts, without reporting any proteolytic degradation (4). They were able to show that nsP4 retained the TATase activity, and additionally had *in vitro* -RNA synthesizing activity, when supplemented with a BHK-21 cell membrane fraction containing the polyprotein nsP123 (in a form that could not be cleaved internally).

We have expressed nsP4 of CHIKV, in a recombinant form preceded by 6xHis-SUMO. This protein could be purified and also displayed terminal transferase activity in an *in vitro* assay (conference abstract (5)). Unfortunately, the expression and purification procedures did not yield sufficient amounts of pure CHIKV nsP4 for extensive characterization in enzymatic assays (or crystallization studies), due to massive degradation and insolubility. In the future, other nsP4 expression systems should be explored, e.g. baculovirus-driven expression in insect cells, which could be considered a more natural situation for the expression of arbovirus proteins, than bacteria. More recently, a truncated version of CHIKV nsP4 (nsP4-D118) was purified from bacteria, but the characterization of its *in vitro* TATase activity and detergent tolerance did not provide new information with respect to what had been described for SINV 13 years earlier (6).

While these earlier studies (1, 4) were very useful for optimizing the purification strategies for alphavirus polymerases and understanding the minimal requirements for their *in vitro* activity, they did raise some major concerns, since they relied on T7 RNA polymerase driven expression. As pointed out by Lehmann *et al.*, T7 RNA polymerase can be a notorious contaminant in this type of assays, which can lead to false-positive results (7). Although the active site mutant nsP4 was also tested, controls using 'empty bacteria', only expressing the T7 RNA polymerase were lacking in the papers cited above.

Due to the unsolved technical issues and problems with the 'reconstitution system' based on pure nsP4 and nsP123, we decided to (semi)purify active RTCs from CHIKV-infected cells. This strategy had already been successfully applied in our laboratory for several nidoviruses (8, 9). An *in vitro* system for studying the RNA synthesis of SINV had also been developed, but the experimental design heavily relied on the T7 RNA polymerase-driven expressing only the T7 RNA polymerase were lacking (10).

Chapter 2 describes in detail the purification of CHIKV RTCs from infected mammalian cells, the optimization of an *in vitro* replication assay (IVRA) to study RNA replication and its application to evaluate (direct acting) inhibitors.

CHIKV RTCs were harvested at a timepoint (6 h p.i.), when they were most active in +RNA synthesis (genomic and subgenomic RNA). Newly synthesized RNA was detected by the incorporation of ³²P-CTP and reaction products were stable in this system, being protected from the activity of cytoplasmic nucleases perhaps due to association with cellular membranes (11), capsid proteins or polysomes. This was concluded after comparing the half-life of the RNA already present in the isolated RTCs with an RNA transcript introduced in the system, which was rapidly degraded. In addition, it has also been shown that newly introduced RNA templates are not used by pre-assembled SFV replication complexes, possibly due to their sequestration in spherules (12).

Surprisingly, in addition to the CHIKV genome and subgenomic RNA, another +ssRNA molecule was consistently detected (see figure 1 below), both in infected cell lysates as well as after synthesis of CHIKV RNA in the IVRA. This newly (re)discovered RNA species, termed RNAII, corresponded to the 5'-proximal ~7.5kb of the genome, up to the subgenomic promoter region (Psg). An RNA similar to the one we found for CHIKV has been described for SINV in 1997 and was named RNAII, as it is part of the replicative form II (RFII) (13). Earlier publications from the 1970's on SFV and SINV also mentioned the presence of other ssRNA besides the genome and subgenomic mRNA (14, 15). When we examined figures in publications from the late 1980's (16), we could also observe RNAII, although it was misidentified in the text. RNAII is likely visible in several other older publications, but often was ignored or mislabeled. A recent example is a publication on the importance of non-structural polyprotein processing and nsPs for SINV pathogenesis, in which the authors present pictures of gels (Fig 3 C, D) in which RNAII is clearly present (separated from the genomic and subgenomic RNA), but not indicated (17). In addition, in the same paper, another SINV-specific RNA can be noticed below the sgRNA, which becomes more abundant later in infection. We have observed a similar RNA in CHIKVinfected cells (Chapter 2, fig 1a), but this species is not detected as a product of the IVRA, although this might be due to its small size and limited incorporation of radiolabel (below the detection level).

The role of RNAII in alphavirus replication has not been investigated in detail. It could be merely a byproduct from complexes that were engaged in genome synthesis and became blocked/stalled when reaching a region of active transcription at the subgenomic promoter region. From an evolutionary point of view, this assumption does not appear to make sense, considering how well-regulated the rest of the replication cycle is. It also remains to be determined if RNAII is capped and/or polyadenylated, and whether it may function as an mRNA for nsP synthesis.

I would favor the idea that RNAII is produced in order to drive the predominant synthesis of sgRNA at later time points in infection, by forming a dsRNA region in the preceding 5' part, hence directing the RTCs to only transcribe the single-stranded region of the negative sense RNA. Of course, this raises many questions worthy of investigation. For example, is

RNAII present in comparable amounts as -RNA? Which type of RTC is responsible for its synthesis? Could it perhaps be the short-lived nsP1-nsP23-nsP4 complex? Another open question is whether RNAII is exported from the spherules that contain the RTCs, since all other +RNAs are released into the cytosol. Whether the synthesis of RNA II is indeed connected to the production of sgRNA could be explored using seco-pregnane steroids, which have been shown to specifically block sgRNA synthesis for SINV and other viruses (18).

In order to investigate the nucleotide requirements of several CHIKV fidelity mutants, another research group has applied an *in vitro* assay somewhat similar to ours, which relied on ³²P-UTP incorporation (19). However, their assay optimization is not described and genomic RNA synthesis was hardly visible. This is making it difficult to reliably assess whether genomic and subgenomic RNAs are truly produced de novo in this system, and that the incorporation did not merely result from 'end labeling' of pre-existing RNA molecules. The fact that these authors did not observe RNAII raises further concerns about RTC activity and reaction products. Because of the discrepancies between the results obtained with the two assays, it is important to make a comparison concerning how they were designed to understand where the dissimilarities might stem from. Stapleford et al. harvested RTCs at the moment when - in our hands - they are hardly active anymore. Prior to setting up the IVRA, we first used metabolic labeling to determine when RTCs were most active and harvested at that particular moment. Also, why the authors used an MOI of only 1 for infection and then harvested the RTCs at 16 h p.i. is not clear. We used high MOIs to achieve a synchronized single-cycle infection and thus maximize the quantity of active complexes. The choice of $[\alpha^{32}P]$ -UTP as the radiolabeled nucleotide in the IVRA of Stapleford *et al.* is also questionable, since this can produce false-positive results due to UTP incorporation driven by host terminal transferases (8), which is why we opted to use radiolabeled CTP in our assays. Lastly, it is unclear why the authors chose to separate the isolated RNA in non-denaturing agarose gels, as it gave poor and uncertain results, due to the absence of loading controls. Though the work of Stapleford et al. extends beyond their in vitro CHIKV replication assay, it is regrettable that they did not consult the recent literature concerning this topic.

The observations concerning RNAII underline that the replication of alphaviruses is more complex than presented in most articles and text books. Therefore, further investigation is required as the potential role of the additional RNAs in the replication cycle, pathogenesis and dissemination in the insect and mammalian host remains unclear.



Fiure 1. Schematic representation of CHIKV replication, updated to include RNAII.

Besides the opportunity to study more fundamental aspects of CHIKV RNA synthesis in depth, our IVRA can also be used as a tool to evaluate CHIKV inhibitors, e.g. to determine whether they are active in their present form and do not require further metabolic processing (assuming this step would be performed extracellularly or in living cells). Additionally, the IVRA could be used to determine if compounds (e.g. identified by cell-based screening) directly affect viral RNA synthesis, by targeting one or more proteins of the RTC or its co-opted host factors. The assay would also allow for screening of compounds that are cytotoxic or poorly taken up by cells, obviously followed by medicinal chemistry-driven efforts to improve their activity and obtain derivatives suitable for use in cells, and ultimately humans.

2. Suramin inhibits the activity of CHIKV RTCs in vitro

Using our IVRA, the first non-nucleosidic compound identified as an inhibitor of CHIKV RNA synthesis was suramin, a compound marketed for the treatment of parasitic infections. It also inhibited the activity of SINV and SFV RTCs, as well as the activity of RTCs isolated from cells transfected with CHIKV RNA replicon-s, with a similar IC₅₀ as found for the complexes obtained from infected cells (see Chapter 3). The major advantage of using an IVRA based on replicon RTCs is the absence of virus particles, which thus offered a biosafe alternative for studying the structure-activity relationship of several suramin-related compounds. Based on this assay, we could conclude that the symmetry of the compound and the presence of a high number of sulfate groups (negative charges) are a pre-requisite for inhibition of CHIKV replication. Interestingly, using the replicon

RTCs-based IVRA, we could also show that suramin did not compete with NTPs, and that its inhibitory mechanism appeared to be the blocking of re-initiation of RNA synthesis (on negative strand templates).

Initially, suramin was identified in a molecular docking screen for compounds that bind to the active site of the norovirus RdRp (20), which represents a conserved region among the polymerases of RNA viruses. This was the main reason why it was chosen for testing in the IVRA, and its inhibitory effect on RNA synthesis, presumably by blocking the RdRp of CHIKV (nsP4), was confirmed (Chapter 3). Because suramin can interact with many positively-charged (RNA-binding) protein surfaces, many other viral or host proteins could (also) be the target of suramin (summarized in the introduction of Chapter 3 and discussed in more details below, at point 5). In the IVRA, an (additional) effect on nsP2, which has NTPase, 5'-RNA triphosphatase and helicase activity cannot be excluded, as suramin was also shown to inhibit HCV NS3 helicase activity (21). The NTPase and 5' RNA phosphatase activities of CHIKV nsP2 depend on a pH of 7 or higher and the presence of Mg²⁺ and NTPs (22), conditions which are met in our IVRA and would therefore support these activities. Thus, it is highly possible that suramin targets both nsP4 and nsP2, while inhibiting CHIKV RNA synthesis.

Furthermore, it is important to exclude that suramin binds and inhibits the activity of creatine phosphokinase (CPK), which is part of the crucial energy regenerating system (relying on creatine phosphate and creatine phosphokinase). Using a separate assay, to monitor only the activity of CPK in the presence of suramin (a typical IVRA reaction, but without the RTCs), we could confirm that suramin does not inhibit the synthesis of ATP by the energy regenerating system (results not shown). Hence, the CP/CPK system is not among the targets of suramin in the IVRA.

Results obtained using IVRAs should obviously be confirmed in cultured cells or *in vivo* (if it is known). For example, certain nucleoside analogs, even if supplied in their active form (i.e. tri-phosphorylated), may not show an inhibitory effect on RNA synthesis in an IVRA, because they exert a dual effect in living cells on both host and virus. For example, ribavirin triphosphate can inhibit virus replication indirectly by reducing GTP pools (23, 24), but also by being incorporated into the newly produced RNA and causing lethal mutagenesis (25).

3. In cell culture, suramin inhibits multiple steps of the CHIKV replication cycle

To confirm the inhibitory activity of suramin that was found in the IVRA, we analyzed its effect in cell-based assays for CHIKV infection. Using a dose-response assay we confirmed that suramin inhibits CHIKV replication in a dose-dependent manner, as reduced RNA and protein levels were observed, and a decrease in infectious progeny was observed. These observations were validated in parallel by another group, using CHIKV strain S27 in

BHK21 cells (26). Their work supports part of our results presented in chapters 3 and 5, and therefore strengthens our conclusions. The following paragraphs summarize our results from Chapters 3 and 5 and compare them with the findings of Ho *et al.*

In Chapter 3, suramin was shown to have an antiviral effect in CPE reduction assays with several CHIKV isolates and related alphaviruses SINV and SFV. Ho *et al.* have performed a similar analysis with several CHIKV strains (S27 and three Asian strains from Singapore, Indonesia, and Malaysia) on BHK-21, U2OS, and MRC-5 cells. Their EC₅₀ estimations were much lower than ours, which might be due to the strains used in their study. However, the discrepancy is more likely caused by their viability assay which is based on crystal violet staining and OD measurement, which in our experience is less accurate than colorimetric cell viability assays like the MTS assay. Surprisingly, the authors did use such a commercial WST-1 viability assay to determine the CC₅₀ of suramin, but for unclear reasons did not use the same assay in the CPE reduction assay to determine the EC₅₀, EC₅₀ and CC₅₀ determination should be performed in parallel and with the same assay. to avoid discrepancies and inconsistencies.

To identify the step of the replication cycle that is targeted by suramin, time of compound addition experiments were performed. Despite some differences in experimental setup and readout, both our experiments (Chapter 3) and those by Ho et al. demonstrated that suramin blocks CHIKV replication more efficiently when it is used prior to infection or at the moment the virus is added to the cells. These results imply that suramin might interfere with virus binding/entry, e.g. by blocking the interaction with receptor or co-receptor (by binding to the sites on the virus surface required for receptor recognition/interaction, or vice versa, by saturating the virus-binding sites on the receptors). In addition, Ho et al. have shown that suramin treatment at 2 or 6 h p.i. did not affect intracellular levels of CHIKV RNA, while the extracellular levels were 1 log lower, suggesting that suramin also interferes with the release of newly formed particles. In contrast to the effect on RNA synthesis found in the IVRA, suramin mainly inhibits an early step of the replication cycle in cell-based assays. We could demonstrate some effect on RNA synthesis also in infected cells, but this would normally be overshadowed by the early effect on binding/entry. One complicating factor is that the large size and negative charges of suramin likely will hamper its cellular uptake (27). Because of the anticipated low intracellular concentration, suramin will most likely not have a direct effect on the activities of the nsPs in the infected cell. In infected cells the effect on RNA synthesis is limited, compared to the major effect suramin has on the entry step - either by inactivating the virus, blocking its attachment to the cells surface or fusion with the host membranes or several of these processes.

We demonstrated that ³H-suramin binds to intact CHIKV and SFV particles, more specifically to the proteins on the envelope surface (chapter 5). We have shown that this binding does not have a virucidal effect and this was confirmed by Ho *et al.* Therefore, we

presume suramin does not bind to virus particles irreversibly, and most likely the compound will dissociate in media supplemented with FCS, as serum albumin was shown to have a high affinity for suramin (28). Using a direct approach based on radiolabeled or fluorescent CHIKV, we demonstrated that suramin inhibits virus attachment to cells (Chapter 5), in assays performed at 4°C, a condition that would prevent endocytosis. Our findings were supported by those of Ho *et al.*, who have used a PRNT-based assay. Consequently, virus attachment is at least one of the steps that is targeted by suramin. In Chapter 5, a bulk fusion assay was employed, which revealed that suramin also inhibits fusion with liposomal membranes. This might be because the compound blocks the structural changes required for the two membranes to fuse. Our experimental setup allowed a more direct, specific measurement of fusion, compared to the insect cell-based expression of structural proteins used by Ho *et al.*, which demonstrated that suramin appeared to inhibit low pH-induced cell fusion (similarly to a neutralizing antibody). In conclusion, the interaction of suramin with CHIKV appears to interfere with attachment as well as fusion.

Ho *et al.* also described a minor inhibitory effect of suramin on CHIKV release, and they have shown that high suramin concentrations block the release of virus, without affecting intracellular viral RNA levels. These authors claimed that virus budding was affected, leading to reduced extracellular transmission. An alternative explanation would be that the newly released particles are less infectious, perhaps by suramin inhibiting the proper maturation of structural proteins during their post-translational trafficking along the secretory pathway. In support of this idea, the activity of several lysosomal enzymes is inhibited by suramin (29, 30), which was also shown to accumulate in other low-pH cellular compartments with a low pH, such as the trans-Golgi network (31, 32).

We have used molecular modeling to identify potential suramin binding sites on the virion surface. In Chapter 5, we predicted that one heterotrimeric CHIKV surface projection could bind several suramin molecules towards its top. The ligand would stretch from the N-terminal disordered region of one E2 protein towards the middle of domain A of an adjacent E2 molecule. In this manner, one arm of suramin, with its negative charges, blocks the region of domain A that interacts with glycosaminoglycans (GAGs) or specific receptors. This region was recently found to be the target of two neutralizing antibodies (NAbs) that block virus entry and egress, and for which escape mutations at residues W64 and G95 were found (33). The W64G substitution in E2 offered an escape from neutralization by both NAbs, but left the virus with an attenuated phenotype in mice.

Ho and collaborators have also used molecular docking in an attempt to predict suramin binding sites, but they have modelled suramin on a single E1-E2 heterodimer and consequently their results predict suramin to bind to a region between the two proteins that is not exposed to the environment in the actual trimeric surface projection/spike. Therefore, it is unlikely that the surface they suggest to be the suramin-binding site will have much relevance for the inhibition of virus entry.

In Chapter 3, we concluded that suramin did not block virus attachment to the cell surface, while in Chapter 5, the results obtained with the radiolabeled or fluorescent CHIKV clearly demonstrated inhibition of attachment. This apparent discrepancy was due to the fact that we initially used an indirect measurement, the quantification of viral RNA by RT-qPCR to measure bound virus particles, whereas in Chapter 5 we relied on direct measurements of labelled viruses. The problem with the RT-qPCR-based method can be attributed to the commonly used virus stocks, which have genome copy to PFU ratios that are over 1000:1, due to the moment of harvesting when extensive CPE has occurred and intracellular viral RNA has likely been released into the medium as well (our own observations and those of (34, 35)). Besides providing a more direct measure for the binding of intact, envelopelabelled viruses, the ³⁵S-virus preparations are further purified to remove unincorporated label and other contaminants (naked RNA, nucleocapsids). Thus, we assume that the lack of effect found in Chapter 3, was due to this technical issue (i.e. the bulk of detected RNA did not represent infectious particles) that masked the inhibition of binding by suramin. We therefore think the experiments with labeled virions represent the actual situation more accurately and concluded that suramin does inhibit virus attachment. This was corroborated by additional RT-qPCR based experiments with improved purification of virus stocks, which however still had RNA copy:PFU ratios of ~80:1, but did reveal a modest inhibitory effect of suramin.

While binding experiments with radiolabeled CHIKV at 4°C clearly demonstrated that suramin inhibited attachment, we obtained more puzzling results when we studied the effect of suramin on virus binding and uptake at 37°C (results shown in the Appendix, left side). At this higher temperature, we found an increased amount of radiolabeled envelope proteins at 1 h p.i. in lysates of cells treated with suramin compared to untreated cells. Because the cells are metabolically inactive at 4°C, we suspect that suramin then blocks the electrostatic interactions between CHIKV and GAGs, the negatively charged co-receptors/ attachment factors, which would lead to release of virus during the washing steps. Under physiological conditions (37°C), when endocytosis occurs, treatment with suramin might not inhibit attachment so strongly, because the viral attachment dynamics are much faster at this higher temperature. At 37°C suramin might have an inhibitory effect later in the infectious cycle, for example on the fusion step in the endosome.

In infected cells, the envelope proteins of radiolabeled virions display a clear degradation pattern when total cell lysates are compared at 1 and 3 h p.i. This might be due to endosomal degradation of virion-associated envelope proteins or degradation of post-fusion envelope

proteins that ended up in membranes that would further progress through the endosomal pathway. However, these proteins remained stable in suramin-treated samples. There are four possible explanations for these observations:

CHIKV particles remained attached to the plasma membrane and did not enter via endocytosis, thus remaining resistant to degradation in the endo-lysosomal pathway.

CHIKV particles reached the endosomes, but the envelope proteins were not degraded because suramin inhibited the responsible proteases or blocked endosomal maturation.

CHIKV particles were taken up by endosomes, but their conformation was 'frozen' by suramin, preventing membrane fusion and rendering the envelope proteins resistant to degradation.

With the bold assumption that, once in the endosome, CHIKV envelope projections require enzymatic cleavage in order to expose the fusion loop (as is the case for the spike proteins of coronaviruses), suramin might block the endosomal enzyme required for the fusion step.

4. Mutations in the CHIKV E2 protein lead to suramin resistance

To determine the target of suramin, we have selected for suramin-resistant CHIKV variants. The reverse genetics studies described in Chapter 5 demonstrated that the mutations N5R and H18Q in the E2 protein were both responsible for suramin resistance. These mutations are in regions of E2 that are not highly conserved between viruses of the SFV clade. To our knowledge, these were the first reported mutations that cause some resistance to suramin for any virus. The N5R mutation is located in a flexible loop with a nearby positive charge (K3), and the R could potentially orient itself towards areas/ligands with negative charges. The H18 residue is not surface exposed at all and is thought to be involved in E2-E1 contacts in the p62/E1 crystal, together with the preceding amino acid, L16 (36).

Individually, the suramin-resistance mutations offer little resistance, but when combined, the S9 virus performs better in the presence of suramin in CPE and PRNT-like assays, and it also shows improved replication kinetics. The N5R and H18Q mutations of S9 did not offer the virus an advantage during the attachment step, but apparently offered an advantage at a later stage of entry, perhaps during fusion of the viral envelope with the host membrane.

Besides the E2 mutations that were shown to be responsible for suramin-resistance, mutations in several nsPs were also detected after repeated passaging of CHIKV in the presence of the compound. These mutations, R171Q and T301K in nsP1 and the opal-R in nsP3, as discussed in chapter 5, seem to be merely non-specific cell culture adaptations. In a study concerning SINV, repeated passaging of the virus in BHK-21 cells gave rise to adaptive mutations, mainly in E2 (S1R, D70K and S114R), which rendered the virus attachment dependent on heparan sulfate (HS) for infection (37). However, in an assay with liposomes containing lipid-conjugated heparin there was no difference in the low pH-induced fusion activity of wt virus and the adapted heparin-binding virus (38). These

findings indicate that whether SINV strains interact with HS or not at neutral pH, they are all capable of fusion with membranes under acidic conditions and perhaps this could be a characteristic of alphaviruses.

While analyzing mutations in CHIKV E2 that were previously reported in the literature, we discovered that the G82R mutation in E2 makes the virus dependent on HS (39). G82R, the most important factor that causes attenuation of CHIKV 181/25 (40), is the same mutation that was independently selected for in MRC-5 cells treated with the antiviral compound arbidol, which is marketed as a broad-spectrum antiviral drug for the treatment of respiratory infections (41). This was likely not noted before as the authors referred to this mutation as G407R, based on the amino acid numbering of the full-length structural polyprotein instead of E2 alone. We were unable to reproduce the results of Delogu et al. in our CPE-reduction assays, and found that arbidol was quite toxic and provided little protection against CHIKV infection (unpublished results). Therefore, we suspect that the outcome of the resistance-passaging of Delogu et al. merely reflected adaptation of CHIKV to more efficient infection of MRC-5 cells (even in the presence of arbidol). In our plaque assays, CHIKV E2-G82R had a small-plaque phenotype in Vero E6 cells, while virus titers were not affected. This combination might be explained by a restricted cell-to-cell spread (maybe HS expression is less abundant in Vero E6 cells). Strikingly, in MRC-5 cells the E2-G82R mutant virus had a large-plaque phenotype, suggesting enhanced cell-to-cell spread (results not shown). G82R, but also wt CHIKV reached higher titers in MRC-5 cells than on Vero E6 cells and it would be interesting to study whether this is due to differences in the GAG abundance/expression patterns. As mentioned above, adaptation of CHIKV to HS-binding in cell culture is expected to direct the selection of mutations that increase the number of positively charged residues at the surface of the E2 protein (37, 40). This is corroborated by the observation that the G82R mutation attenuates CHIKV because it renders virus infectivity dependent on HS (39, 42). Due to their location, it is unlikely that the N5R and H18Q mutations that cause suramin-resistance, are a result of cell culture adaptation, since neither of these mutations is located in the region targeted by neutralizing antibodies (33) or directly involved in receptor-binding (43).

To understand if any of the mutations acquired by passaging CHIKV in the presence of suramin offers an advantage in other cell types (treated or not with suramin), we compared various reverse-engineered mutants in CPE-based assays (results not shown). Wild-type virus and all tested mutants produced extensive CPE in Vero E6 cells, except for the T301K virus. In the presence of suramin, only the mutant viruses with both E2 mutations replicated and caused CPE in Vero E6 cells. The E2 G82R mutation caused an extreme sensitivity to suramin and it was the only mutant that exhibited sensitivity to suramin in all tested cell lines (Vero E6, HeLa, MRC-5, BHK21), implying that suramin interferes with its attachment to the cell surface via HS. Interestingly, the G82R and N5R mutants were the only two variants that caused CPE in HeLa cells (in the absence of suramin),

suggesting that also the N5R mutation has an effect on interactions with HS or another GAG, and thereby modulates infectivity in a cell type-dependent way. In MRC-5 cells, most CHIKV variants caused extensive CPE and did not respond to suramin treatment, with the exception of the G28R mutant. It seems that CHIKV wt and the S9 mutant with the two suramin-resistance mutations in E2 share a similar HS-independent entry mechanism and a complete resistance to suramin treatment in MRC-5 cells. Consequently, the selected suramin-resistant mutations seem to be specific for the situation in Vero E6 cells treated with suramin – and might represent an escape mechanism that allows faster entry in the presence of suramin in this cell type.

Suramin was originally developed to treat Trypanosoma infections, but its exact MoA has not been clarified (44). Nevertheless, T. brucei strains with increased resistance to suramin could be selected, but only in haemolymphatic stage parasites and not in the case of the procyclic forms produced in insects (45). Using RNAi target sequencing (RIT-seq), followed by RNAi screens, Alsford et al. were able to shed more light on how the antitrypanosomal activity of drugs (suramin included) was actually induced (46). In the case of suramin multiple targets surfaced, linked to its uptake or inhibitory activity, and eight were selected for further investigation. Knockdown of the endomembrane protein MFST (major facilitator superfamily transporter) the lysosomal cathepsin-L like protease (Cat-L) lead to a clear increase in EC_{50} for suramin. Other identified proteins that affected the sensitivity towards suramin were: a bloodstream stage-specific invariant surface glycoprotein (ISG75), lysosomal proteins (CBP1peptidases, p67 and Golgi/lysosomal protein-1, GLP-1), several spermidine and N-acetylglucosamine (NAG) biosynthetic enzymes, and all subunits of the adaptin complex (AP) 1, which is involved in clathrin-mediated endocytosis. The proposed mechanism-of-action for suramin is summarized in Figure 2. Although it had been shown previously that suramin resistance was linked to downregulation of endocytosis, another study demonstrated it was connected to antigenic variation of trypanosomal surface glycoproteins (out of 2000 types, only one is expressed), as suramin treatment led to the emergence of a *T. brucei* form with a surface glycoprotein (VSGsur) that caused resistance to suramin treatment (47).



Figure 2. Schematic summary of RIT-seq determinants of suramin, and other drugs, efficacy in *T. brucei*. Depicted in red and green are proteins and metabolites linked to drug activity. For suramin, the proposed course of events would be binding to ISG75, and subsequent accumulation in lysosomes via the flagellar pocket (FP) and endosomes. From there MFST could transport suramin into the cytosol. (Modified from (46)).

5. Suramin has broad-spectrum antiviral activity

Another important pathogen that is sensitive to suramin is ZIKV, a flavivirus endemic to Africa, that re-emerged and at the beginning of 2015 caused a massive epidemic in South America with devastating neurodevelopmental outcomes for newborns from infected mothers. Briefly, as described in Chapter 4 of this thesis, we have shown that ZIKV SL1602 (a clinical isolate) is sensitive to suramin using several assays adapted from our CHIKV studies. For ZIKV and CHIKV the compound seems to have a similar MoA, affecting virus entry steps as well as the release of infectious particles. The replicative cycles of alpha- and flaviviruses have similarities, such as GAG-dependent attachment to the plasma membrane and entry via receptor mediated endocytosis (RME). ZIKV attachment is mediated by electrostatic interactions of positively-charged amino acids of the E protein with negativelycharged GAGs at the cell surface, having preferably long-chained and highly-sulfated HS (48, 49). There are also significant differences between the replicative cycles of alpha- and flaviviruses, like the mechanism of virion biogenesis. In the case of flaviviruses this relies on budding into the ER lumen (not at the plasma membrane as for CHIKV) and maturation during transit through the secretory pathway. Suramin could accumulate in the trans-Golgi network and hence impair the activity of enzymes (like furin or glycosyltransferases) required for the maturation of new ZIKV particles, or it could block virus release, through an unknown mechanism.

While we were investigating suramin's MoA against ZIKV, another research group made similar discoveries (50). Their findings in part corroborated ours, leading to the conclusion that suramin did not "inactivate viral particles, but interfered with virus adsorption, entry and post-infection events". However, some of their results appear to differ from ours and other previously published studies, but this might be due to their un-conventional experimental setup and the fact that also their controls behave different than published elsewhere in literature.

Suramin's inhibitory activity against the alphaviruses CHIKV, SFV and SINV (this thesis, (26, 51)), as well as against flaviviruses ZIKV (this thesis, (50)), DENV (52), BVDV (53), and HCV (54) appears to be based on a similar primary mechanism, i.e. the compound blocking the access to cell surface attachment factors. The bunyaviruses RVFV (55) and SFTSV (56) are inhibited by suramin at several steps during infection, but the main mechanism seems to be through the interaction with the nucleocapsid protein (N) and interfering with budding into the Golgi apparatus, which is crucial for particle assembly.

Suramin's spectrum extends beyond the alpha-, flavi- and bunyaviruses mentioned above. It can also inhibit DNA viruses, retroviruses and other RNA viruses, independent of whether they are enveloped or non-enveloped.

DNA viruses targeted by suramin at their entry step include HSV-1, CMV and duck hepatitis B virus (57-59). Suramin also inhibits entry of retroviruses, and HIV-1 was actually the first viral pathogen for which suramin's antiviral activity was evaluated in human patients (60, 61). Rous sarcoma virus is also sensitive to suramin, which blocks virus uptake or uncoating (59).

Suramin was shown to also inhibit non-enveloped viruses of the *Picornaviridae* family, more specifically the causative agent of hand foot and mouth disease, EV-A71 (62, 63). Interestingly, suramin's spectrum of activity was restricted to type A enteroviruses, and its MoA depended on competition with sulfated receptors for a binding site at the 5-fold vertex of the EV-A71 capsid, blocking virus attachment to cells (64).

Suramin also inhibits ebola virus (EBOV), another important enveloped RNA virus that has received a lot of public interest due to the recent serious epidemics in Africa. Suramin treatment also affected an early step of the EBOV infectious cycle in cell culture, as demonstrated with EBOV envelope glycoprotein pseudo-typed lentiviral vectors (51).

Preliminary (unpublished) results from our group have shown that suramin can also inhibit MERS-CoV (isolate EMC2012) and SARS-CoV (isolate Frankfurt-1) in CPE-based assays using Vero cells, with EC_{50} s of 50 and 100 µM, respectively. Even though the two coronaviruses have specific protein receptors (DPP4 and ACE2) for entering their target cells, MERS-CoV can also use attachment factors conjugated with sialic acid residues, which contribute to the negatively charged environment at the cell surface. Presumably by binding to the viral surface, suramin could block this interaction in Vero cells (65). In theory suramin could also affect the enzymatic activity of non-structural proteins (nsPs) by due to its high affinity for positively charged regions, such as those present in viral proteins interacting with negatively charged molecules, such as RNA. In cell culture, the effect on intracellular nsPs would likely be limited, since the compound accumulates in intracellular compartments and not in the cytosol or the replication organelles where most of those viral proteins would exert their activity.

In summary, we can conclude that suramin inhibits a wide variety of enveloped and nonenveloped RNA and DNA viruses, likely through a common mechanism that involves interfering with the electrostatic interactions between viruses and attachment factors at the cell surface.

6. Therapeutic strategies for CHIKV infections and outlook

As mentioned earlier, CHIKV causes a highly debilitating disease, for which vaccines or antiviral therapy are currently not on the market. Treatment of patients is mainly supportive at the moment (e.g. through the use of analgesics/painkillers). The development of a vaccine against CHIKV is of the utmost importance to prevent further spread of the virus and large epidemics as we have seen in the recent past. Antiviral compounds would be required, to treat people that have been already infected and are struggling with the persistent painful consequences (chronic arthritis for e.g.) of CHIKV infection. Compared to vaccines, antiviral compounds might be cheaper, easier to administer, stockpile, distribute and more suitable for certain target groups (that cannot be vaccinated) and emergency usage in new outbreak situations (to curb the outbreak). Considering the enormous costs involved in bringing a new drug to the market, broad-spectrum antivirals and repurposed compounds already on the market for other indications are of particular interest.

Suramin, a drug that has already been used for over 100 years to treat parasitic infections, was shown to have antiviral activity against CHIKV by us and others (Chapters 3 and 5 of this thesis, (26)). The drug was synthesized for the treatment of the early stages of trypanosomiasis and is still in use today and is even offered for free by the WHO in trypanosomiasis endemic areas. A course of suramin treatment for trypanosomiasis costs 27\$, and it comprises five 1-g intravenous doses, administered over 2 weeks, with minimal side effects. In November 2018, a Dutch tourist returning from Malawi developed trypanosomiasis and required urgent suramin treatment, which was not available in the Netherlands or Belgium and had to be imported from the Tropical institute in Basel. This illustrates that suramin has rightfully been placed on the WHO list of essential medicines, which each country should have available.

Suramin has not been tested in clinical trials for the treatment of CHIKV infections, but its efficacy was shown in a CHIKV mouse model (66). It has also been shown to be effective in mice and adult Rhesus monkeys for the treatment of EV-A71 infections, suggesting that suramin is a promising compound for the prevention and treatment of hand foot and mouth disease (63).

A highly efficient way to prevent drug-resistance during antiviral therapy of (rapidly mutating) RNA viruses, is the simultaneous use of multiple drugs with different viral targets, as exemplified by the combination of simeprevir and sofosbuvir to cure HCV infections or or the use of several multi-class combination drugs like Prezcobix, or Evotaz to control HIV infection (67, 68).

A dual-agent treatment, combining suramin and zanamivir (an anti-influenza drug), was tested *in vitro* against human parainfluenza infection (hPIV-3), which can cause serious respiratory illness in infants (69). The authors have shown that lower concentrations of both compounds can be used when they are combined, leading to higher levels of inhibition by simultaneously interacting with the haemagglutinin-neuraminidase (HN) protein. This exemplifies that combinatorial repurposing approaches of approved drugs can be a fast and rewarding way to develop new antiviral therapies.

Recently, it was shown that treatment of U2OS cells with a combination of suramin and epigallocatechin-3-gallate synergistically inhibited the replication of the African CHIKV strain S27 (70). Combining suramin with inhibitors that specifically target the activities of CHIKV nsPs might be a promising strategy towards the development of a combination treatment for CHIKV infections. Potential candidates are an inhibitor that targets the methyltransferase activity of CHIKV nsp1 (71) and favipiravir (72), a broad spectrum drug targeting the RdRp activity of CHIKV. Compounds that stimulate the host's natural antiviral mechanisms, e.g. by inducing viperin expression (73), could also be considered for combination treatment.

Hopefully the antiviral strategies mentioned above can be applied to treat patients and contain CHIKV epidemics. Nevertheless, both vaccines and antiviral drugs should be considered as pillars of a coordinated strategy during epidemics, regardless of the pathogen at hand (74).

Concluding remarks

This thesis describes the quest for compounds targeting alphavirus replication, which started with the development of an *in vitro* assays to study RNA replication and identify compounds with antiviral potential, but took an unexpected turn.

Suramin, which directly blocked CHIKV RNA synthesis in the *in vitro* assay, turned out to have a different mode-of-action in cell-based assays, a story with some parallels to early studies on the effect of suramin on HIV-1 infection (75).

In cell culture suramin primarily inhibits CHIKV binding and fusion with host membranes. Besides being an anti-parasitic drug with anti-cancer properties, suramin also inhibits a variety of viruses, and the work described in this thesis has demonstrated that its antiviral spectrum extends to alphaviruses and ZIKV.

Viruses have always been part of our existence, shaping human evolution and continuing to do so, probably even on a bigger scale, due to increased globalization, travel, changes in land use and expansion of human activities into previously uninhabited areas. Climate change and the rise in temperatures will lead to expansion of the distribution of insect vectors and likely will increase the incidence of outbreaks of "once tropical" diseases caused by DENV, ZIKA, CHIKV, *Plasmodium* sp., *Vibrio cholerae* etc. The design of better vaccines and development of new antiviral strategies, including those based on drug repurposing should enhance our preparedness for preventing and treating these infections.

References

- Tomar S, Hardy RW, Smith JL, Kuhn RJ. 2006. Catalytic core of alphavirus nonstructural protein nsP4 possesses terminal adenylyltransferase activity. J Virol 80:9962-9.
- Rupp JC, Jundt N, Hardy RW. 2011. Requirement for the amino-terminal domain of sindbis virus nsP4 during virus infection. J Virol 85:3449-60.
- de Groot RJ, Rumenapf T, Kuhn RJ, Strauss EG, Strauss JH. 1991. Sindbis virus RNA polymerase is degraded by the N-end rule pathway. Proc Natl Acad Sci U S A 88:8967-71.
- Rubach JK, Wasik BR, Rupp JC, Kuhn RJ, Hardy RW, Smith JL. 2009. Characterization of purified Sindbis virus nsP4 RNA-dependent RNA polymerase activity *in vitro*. Virology 384:201-8.
- 5. ESV. 2013. Book of Abstracts, p 175. In (ed), John Libbey Eurotext, Lyon.
- 6. Chen MW, Tan YB, Zheng J, Zhao Y, Lim BT, Cornvik T, Lescar J, Ng LFP, Luo D. 2017. Chikungunya virus nsP4 RNA-dependent RNA polymerase core domain displays detergent-sensitive primer extension and terminal adenylyltransferase activities. Antiviral Res 143:38-47.
- Lehmann KC, Gorbalenya AE, Snijder EJ, Posthuma CC. 2016. Arterivirus RNA-dependent RNA polymerase: Vital enzymatic activity remains elusive. Virology 487:68-74.
- van Hemert MJ, de Wilde AH, Gorbalenya AE, Snijder EJ. 2008. The *in vitro* RNA synthesizing activity of the isolated arterivirus replication/transcription complex is dependent on a host factor. J Biol Chem 283:16525-36.
- van Hemert MJ, van den Worm SH, Knoops K, Mommaas AM, Gorbalenya AE, Snijder EJ. 2008. SARScoronavirus replication/transcription complexes are membrane-protected and need a host factor for activity *in vitro*. PLoS Pathog 4:e1000054.
- Li ML, Wang H, Stollar V. 2010. *In vitro* synthesis of Sindbis virus genomic and subgenomic RNAs: influence of nsP4 mutations and nucleoside triphosphate concentrations. J Virol 84:2732-9.
- Pietila MK, van Hemert MJ, Ahola T. 2018. Purification of highly active alphavirus replication complexes demonstrates altered fractionation of multiple cellular membranes. J Virol doi:10.1128/JVI.01852-17.
- 12. Pietila MK, Albulescu IC, Hemert MJV, Ahola T. 2017. Polyprotein Processing as a Determinant for *in vitro* Activity of Semliki Forest Virus Replicase. Viruses 9.
- Wielgosz MM, Huang HV. 1997. A novel viral RNA species in Sindbis virus-infected cells. J Virol 71:9108-17.
- Levin JG, Friedman RM. 1971. Analysis of arbovirus ribonucleic acid forms by polyacrylamide gel electrophoresis. J Virol 7:504-14.
- Bruton CJ, Kennedy SI. 1975. Semliki Forest virus intracellular RNA: properties of the multi-stranded RNA species and kinetics of positive and negative strand synthesis. J Gen Virol 28:111-27.
- Barton DJ, Sawicki SG, Sawicki DL. 1988. Demonstration *in vitro* of temperature-sensitive elongation of RNA in Sindbis virus mutant ts6. J Virol 62:3597-602.
- 17. Shin G, Yost SA, Miller MT, Elrod EJ, Grakoui A, Marcotrigiano J. 2012. Structural and functional insights into alphavirus polyprotein processing and pathogenesis. Proc Natl Acad Sci U S A 109:16534-9.
- Li Y, Wang L, Li S, Chen X, Shen Y, Zhang Z, He H, Xu W, Shu Y, Liang G, Fang R, Hao X. 2007. Secopregnane steroids target the subgenomic RNA of alphavirus-like RNA viruses. Proc Natl Acad Sci U S A 104:8083-8.
- Stapleford KA, Rozen-Gagnon K, Das PK, Saul S, Poirier EZ, Blanc H, Vidalain PO, Merits A, Vignuzzi M. 2015. Viral Polymerase-Helicase Complexes Regulate Replication Fidelity To Overcome Intracellular Nucleotide Depletion. J Virol 89:11233-44.
- Mastrangelo E, Pezzullo M, Tarantino D, Petazzi R, Germani F, Kramer D, Robel I, Rohayem J, Bolognesi M, Milani M. 2012. Structure-based inhibition of Norovirus RNA-dependent RNA polymerases. J Mol Biol 419:198-210.
- Mukherjee S, Hanson AM, Shadrick WR, Ndjomou J, Sweeney NL, Hernandez JJ, Bartczak D, Li K, Frankowski KJ, Heck JA, Arnold LA, Schoenen FJ, Frick DN. 2012. Identification and analysis of hepatitis C virus NS3 helicase inhibitors using nucleic acid binding assays. Nucleic Acids Res 40:8607-21.

- 22. Karpe YA, Aher PP, Lole KS. 2011. NTPase and 5'-RNA triphosphatase activities of Chikungunya virus nsP2 protein. PLoS One 6:e22336.
- 23. Te HS, Randall G, Jensen DM. 2007. Mechanism of action of ribavirin in the treatment of chronic hepatitis C. Gastroenterol Hepatol (N Y) 3:218-25.
- 24. Wray SK, Gilbert BE, Noall MW, Knight V. 1985. Mode of action of ribavirin: effect of nucleotide pool alterations on influenza virus ribonucleoprotein synthesis. Antiviral Res 5:29-37.
- Crotty S, Cameron C, Andino R. 2002. Ribavirin's antiviral mechanism of action: lethal mutagenesis? J Mol Med (Berl) 80:86-95.
- Ho YJ, Wang YM, Lu JW, Wu TY, Lin LI, Kuo SC, Lin CC. 2015. Suramin Inhibits Chikungunya Virus Entry and Transmission. PLoS One 10:e0133511.
- Voogd TE, Vansterkenburg EL, Wilting J, Janssen LH. 1993. Recent research on the biological activity of suramin. Pharmacol Rev 45:177-203.
- Muller WE, Wollert U. 1976. Spectroscopic studies on the complex formation of suramin with bovine and human serum albumin. Biochim Biophys Acta 427:465-80.
- 29. Buys CH, Bouma JM, Gruber M, Wisse E. 1978. Induction of lysosomal storage by suramin. Naunyn Schmiedebergs Arch Pharmacol 304:183-90.
- Gritli-Linde A, Ruch JV, Mark MP, Lecolle S, Goldberg M. 1994. Effects of suramin, a polyanionic drug inducing lysosomal storage disorders on tooth germs *in vitro*. Biol Cell 81:143-52.
- Baghdiguian S, Boudier JL, Boudier JA, Fantini J. 1996. Intracellular localisation of suramin, an anticancer drug, in human colon adenocarcinoma cells: a study by quantitative autoradiography. Eur J Cancer 32A:525-32.
- Huang SS, Koh HA, Huang JS. 1997. Suramin enters and accumulates in low pH intracellular compartments of v-sis-transformed NIH 3T3 cells. FEBS Lett 416:297-301.
- 33. Jin J, Liss NM, Chen DH, Liao M, Fox JM, Shimak RM, Fong RH, Chafets D, Bakkour S, Keating S, Fomin ME, Muench MO, Sherman MB, Doranz BJ, Diamond MS, Simmons G. 2015. Neutralizing Monoclonal Antibodies Block Chikungunya Virus Entry and Release by Targeting an Epitope Critical to Viral Pathogenesis. Cell Rep 13:2553-2564.
- Ashbrook AW, Burrack KS, Silva LA, Montgomery SA, Heise MT, Morrison TE, Dermody TS. 2014. Residue 82 of the Chikungunya virus E2 attachment protein modulates viral dissemination and arthritis in mice. J Virol 88:12180-92.
- Coffey LL, Vignuzzi M. 2011. Host alternation of chikungunya virus increases fitness while restricting population diversity and adaptability to novel selective pressures. J Virol 85:1025-35.
- Voss JE, Vaney MC, Duquerroy S, Vonrhein C, Girard-Blanc C, Crublet E, Thompson A, Bricogne G, Rey FA. 2010. Glycoprotein organization of Chikungunya virus particles revealed by X-ray crystallography. Nature 468:709-12.
- Klimstra WB, Ryman KD, Johnston RE. 1998. Adaptation of Sindbis virus to BHK cells selects for use of heparan sulfate as an attachment receptor. J Virol 72:7357-66.
- Smit JM, Waarts BL, Kimata K, Klimstra WB, Bittman R, Wilschut J. 2002. Adaptation of alphaviruses to heparan sulfate: interaction of Sindbis and Semliki forest viruses with liposomes containing lipidconjugated heparin. J Virol 76:10128-37.
- Gardner CL, Hritz J, Sun C, Vanlandingham DL, Song TY, Ghedin E, Higgs S, Klimstra WB, Ryman KD.
 2014. Deliberate attenuation of chikungunya virus by adaptation to heparan sulfate-dependent infectivity: a model for rational arboviral vaccine design. PLoS Negl Trop Dis 8:e2719.
- 40. Gorchakov R, Wang E, Leal G, Forrester NL, Plante K, Rossi SL, Partidos CD, Adams AP, Seymour RL, Weger J, Borland EM, Sherman MB, Powers AM, Osorio JE, Weaver SC. 2012. Attenuation of Chikungunya virus vaccine strain 181/clone 25 is determined by two amino acid substitutions in the E2 envelope glycoprotein. J Virol 86:6084-96.
- Delogu I, Pastorino B, Baronti C, Nougairede A, Bonnet E, de Lamballerie X. 2011. *In vitro* antiviral activity of arbidol against Chikungunya virus and characteristics of a selected resistant mutant. Antiviral Res 90:99-107.

- Silva LA, Khomandiak S, Ashbrook AW, Weller R, Heise MT, Morrison TE, Dermody TS. 2014. A single-amino-acid polymorphism in Chikungunya virus E2 glycoprotein influences glycosaminoglycan utilization. J Virol 88:2385-97.
- Zhang R, Kim AS, Fox JM, Nair S, Basore K, Klimstra WB, Rimkunas R, Fong RH, Lin H, Poddar S, Crowe JE, Jr., Doranz BJ, Fremont DH, Diamond MS. 2018. Mxra8 is a receptor for multiple arthritogenic alphaviruses. Nature 557:570-574.

42.

- 44. Alsford S, Kelly JM, Baker N, Horn D. 2013. Genetic dissection of drug resistance in trypanosomes. Parasitology 140:1478-91.
- Scott AG, Tait A, Turner CM. 1996. Characterisation of cloned lines of Trypanosoma brucei expressing stable resistance to MelCy and suramin. Acta Trop 60:251-62.
- Alsford S, Eckert S, Baker N, Glover L, Sanchez-Flores A, Leung KF, Turner DJ, Field MC, Berriman M, Horn D. 2012. High-throughput decoding of antitrypanosomal drug efficacy and resistance. Nature 482:232-6.
- Wiedemar N, Graf FE, Zwyer M, Ndomba E, Kunz Renggli C, Cal M, Schmidt RS, Wenzler T, Maser P. 2018. Beyond immune escape: a variant surface glycoprotein causes suramin resistance in Trypanosoma brucei. Mol Microbiol 107:57-67.
- Kim SY, Zhao J, Liu X, Fraser K, Lin L, Zhang X, Zhang F, Dordick JS, Linhardt RJ. 2017. Interaction of Zika Virus Envelope Protein with Glycosaminoglycans. Biochemistry 56:1151-1162.
- Agrelli A, de Moura RR, Crovella S, Brandao LAC. 2019. ZIKA virus entry mechanisms in human cells. Infect Genet Evol 69:22-29.
- Tan CW, Sam IC, Chong WL, Lee VS, Chan YF. 2017. Polysulfonate suramin inhibits Zika virus infection. Antiviral Res 143:186-194.
- Henss L, Beck S, Weidner T, Biedenkopf N, Sliva K, Weber C, Becker S, Schnierle BS. 2016. Suramin is a potent inhibitor of Chikungunya and Ebola virus cell entry. Virol J 13:149.
- Chen Y, Maguire T, Hileman RE, Fromm JR, Esko JD, Linhardt RJ, Marks RM. 1997. Dengue virus infectivity depends on envelope protein binding to target cell heparan sulfate. Nat Med 3:866-71.
- 53. Iqbal M, Flick-Smith H, McCauley JW. 2000. Interactions of bovine viral diarrhoea virus glycoprotein E(rns) with cell surface glycosaminoglycans. J Gen Virol 81:451-9.
- 54. Garson JA, Lubach D, Passas J, Whitby K, Grant PR. 1999. Suramin blocks hepatitis C binding to human hepatoma cells *in vitro*. J Med Virol 57:238-42.
- Ellenbecker M, Lanchy JM, Lodmell JS. 2014. Inhibition of Rift Valley fever virus replication and perturbation of nucleocapsid-RNA interactions by suramin. Antimicrob Agents Chemother 58:7405-15.
- 56. Jiao L, Ouyang S, Liang M, Niu F, Shaw N, Wu W, Ding W, Jin C, Peng Y, Zhu Y, Zhang F, Wang T, Li C, Zuo X, Luan CH, Li D, Liu ZJ. 2013. Structure of severe fever with thrombocytopenia syndrome virus nucleocapsid protein in complex with suramin reveals therapeutic potential. J Virol 87:6829-39.
- Aguilar JS, Rice M, Wagner EK. 1999. The polysulfonated compound suramin blocks adsorption and lateral difusion of herpes simplex virus type-1 in vero cells. Virology 258:141-51.
- 58. Baba M, Konno K, Shigeta S, Wickramasinghe A, Mohan P. 1993. Selective inhibition of human cytomegalovirus replication by naphthalenedisulfonic acid derivatives. Antiviral Res 20:223-33.
- Petcu DJ, Aldrich CE, Coates L, Taylor JM, Mason WS. 1988. Suramin inhibits *in vitro* infection by duck hepatitis B virus, Rous sarcoma virus, and hepatitis delta virus. Virology 167:385-92.
- Schols D, Pauwels R, Desmyter J, De Clercq E. 1990. Dextran sulfate and other polyanionic anti-HIV compounds specifically interact with the viral gp120 glycoprotein expressed by T-cells persistently infected with HIV-1. Virology 175:556-61.
- 61. Yahi N, Sabatier JM, Nickel P, Mabrouk K, Gonzalez-Scarano F, Fantini J. 1994. Suramin inhibits binding of the V3 region of HIV-1 envelope glycoprotein gp120 to galactosylceramide, the receptor for HIV-1 gp120 on human colon epithelial cells. J Biol Chem 269:24349-53.
- 62. Ren P, Zheng Y, Wang W, Hong L, Delpeyroux F, Arenzana-Seisdedos F, Altmeyer R. 2017. Suramin interacts with the positively charged region surrounding the 5-fold axis of the EV-A71 capsid and inhibits multiple enterovirus A. Sci Rep 7:42902.

- 63. Ren P, Zou G, Bailly B, Xu S, Zeng M, Chen X, Shen L, Zhang Y, Guillon P, Arenzana-Seisdedos F, Buchy P, Li J, von Itzstein M, Li Q, Altmeyer R. 2014. The approved pediatric drug suramin identified as a clinical candidate for the treatment of EV71 infection-suramin inhibits EV71 infection *in vitro* and *in vivo*. Emerg Microbes Infect 3:e62.
- Nishimura Y, McLaughlin NP, Pan J, Goldstein S, Hafenstein S, Shimizu H, Winkler JD, Bergelson JM.
 2015. The Suramin Derivative NF449 Interacts with the 5-fold Vertex of the Enterovirus A71 Capsid to Prevent Virus Attachment to PSGL-1 and Heparan Sulfate. PLoS Pathog 11:e1005184.
- Li W, Hulswit RJG, Widjaja I, Raj VS, McBride R, Peng W, Widagdo W, Tortorici MA, van Dieren B, Lang Y, van Lent JWM, Paulson JC, de Haan CAM, de Groot RJ, van Kuppeveld FJM, Haagmans BL, Bosch BJ. 2017. Identification of sialic acid-binding function for the Middle East respiratory syndrome coronavirus spike glycoprotein. Proc Natl Acad Sci U S A 114:E8508-E8517.
- 66. Kuo SC, Wang YM, Ho YJ, Chang TY, Lai ZZ, Tsui PY, Wu TY, Lin CC. 2016. Suramin treatment reduces chikungunya pathogenesis in mice. Antiviral Res 134:89-96.
- 67. FDA. 2018. Antiretroviral drugs used in the treatment of HIV infection. https://www.fda.gov/ForPatients/ Illness/HIVAIDS/Treatment/ucm118915.htm. Accessed
- 68. FDA. 2017. Hepatitis B and C Treatments.
- 69. Bailly B, Dirr L, El-Deeb IM, Altmeyer R, Guillon P, von Itzstein M. 2016. A dual drug regimen synergistically blocks human parainfluenza virus infection. Sci Rep 6:24138.
- Lu JW, Hsieh PS, Lin CC, Hu MK, Huang SM, Wang YM, Liang CY, Gong Z, Ho YJ. 2017. Synergistic effects of combination treatment using EGCG and suramin against the chikungunya virus. Biochem Biophys Res Commun 491:595-602.
- Delang L, Li C, Tas A, Querat G, Albulescu IC, De Burghgraeve T, Guerrero NA, Gigante A, Piorkowski G, Decroly E, Jochmans D, Canard B, Snijder EJ, Perez-Perez MJ, van Hemert MJ, Coutard B, Leyssen P, Neyts J. 2016. The viral capping enzyme nsP1: a novel target for the inhibition of chikungunya virus infection. Sci Rep 6:31819.
- 72. Delang L, Abdelnabi R, Neyts J. 2018. Favipiravir as a potential countermeasure against neglected and emerging RNA viruses. Antiviral Res 153:85-94.
- Gizzi AS, Grove TL, Arnold JJ, Jose J, Jangra RK, Garforth SJ, Du Q, Cahill SM, Dulyaninova NG, Love JD, Chandran K, Bresnick AR, Cameron CE, Almo SC. 2018. A naturally occurring antiviral ribonucleotide encoded by the human genome. Nature 558:610-614.
- 74. Monto AS. 2006. Vaccines and antiviral drugs in pandemic preparedness. Emerg Infect Dis 12:55-60.
- 75. De Clercq E. 1987. Suramin in the treatment of AIDS: mechanism of action. Antiviral Res 7:1-10.

Appendix



Effect of suramin on ³⁵S-CHIKV uptake, at 37°C.

At 1 and 3 h p.i. of Vero E6 cells with CHIKV wt and S9, in the presence or absence of suramin, whole cell lysates were prepared in Laemmli sample buffer. Afterwards ³⁵S-labelled CHIKV proteins were separated by SDS-PAGE and visualized by phosphor imaging with a Typhoon scanner.

List of Commonly Used Abbreviations

CHIKV – Chikungunya virus

ZIKV – Zika virus

SINV – Sindbis virus

SFV – Semliki Forest virus

RTC – replication and transcription complex

RNA – ribonucleic acid

nsP - non-structural protein

RME - receptor mediated endocytosis

IVRA - in vitro replication assay

MoA – Mode of action

WT – wild-type

Summary

This thesis is focused on understanding the particularities of Chikungunya virus (CHIKV) replication, and the mechanism(s) by which it can be inhibited by suramin, a compound with a broad spectrum of activity.

Chikungunya virus (CHIKV) is a member of the *Togaviridae* family, and it can be transmitted to humans through the bite of infected mosquitoes. In the past, this virus, of African origin, has caused large epidemics in Asia, but the most recent one took place in 2014-2015 in a new territory, the Americas, where it caused over 1 million suspected and confirmed cases. The infection is manifested through acute joint and muscle pain (that can last for years), fever, and rash. To this day, there is no approved vaccine or treatment for CHIKV infection. A much broader introduction regarding CHIKV and its genome organization, protein function, replication, pathogenesis and preventive or therapeutic strategies can be found in Chapter 1, the general introduction of the thesis. Soon after the CHIKV epidemic was over, another virus, also mosquito-borne and originating from Africa, continued to cause serious health problems in Central and South America. This was the Zika virus (ZIKV), a member of the Flaviviridae family. Though ZIKV mostly causes asymptomatic infections or mild disease characterized by low fever, rash, conjunctivitis, and malaise, the epidemics in the Americas have also linked ZIKV infection to fetal malformations like microcephaly and to Guillain-Barré syndrome in adults. Besides this, previously unrecognized routes of motherto-child and sexual transmission were uncovered. Similarly, as in the case of CHIKV, no preventive or therapeutic strategy (vaccines or drugs) for treatment of ZIKV infection are available on the market. The compound suramin had been marketed for the treatment of parasitic infections, but its anti-cancer and antiviral potential were also discovered in the last 40 years. Therefore, we sought to test if suramin could inhibit the replication of CHIKV and ZIKV.

Chapter 2 describes the development of an *in vitro* replication assay (IVRA) that relies on CHIKV replication/transcription complexes isolated from infected cells. This assay can be used to study CHIKV RNA synthesis, as well as to identify inhibitors of this process and perform mode-of-action studies on these compounds. While optimizing the assay, we identified a new RNA species that is produced during CHIKV infection, alongside the genomic and subgenomic RNA that are required for the production of specific proteins involved in the replication and virus production process, but also in interaction with/ manipulation of the host cell. The new species was called RNAII, similarly to the one identified for the CHIKV-related Sindbis virus (SINV) in 1997 by Wielgosz and Huang. It is suspected to direct the replication/transcription complexes to later in infection predominantly produce the subgenomic RNA, which is required for the production of structural proteins that are used for progeny virus assembly. Using the IVRA, in **Chapter 3** we have identified suramin as a potent inhibitor of CHIKV replication. However, cell-based assays revealed that suramin's main mode-of-action is dependent on the inhibition of an early step of the replicative cycle, namely virus entry into the host cell. In addition, several suramin-related compounds were analyzed, and though these compounds were not more effective, they provided insight into the structural elements (symmetry of the molecule and the presence of negative charges) that are important for both inhibitory activities of suramin observed *in vitro* and in cell culture.

The antiviral effect of suramin is very broad, and in **Chapter 4** we show that it also inhibits the replication of the re-emerging ZIKV, by interfering both with its entry and biogenesis of progeny virions.

Subsequently, we explored how suramin can inhibit CHIKV entry steps. **Chapter 5** describes the identification of the CHIKV E2 envelope glycoprotein as the target of suramin and mode-of-action studies that suggest that by interacting with E2 suramin blocks virus attachment to cells, and subsequent fusion of the particle with cellular membranes. CHIKV can become more resistant to suramin by acquiring mutations in the E2 protein. The amino acid substitutions that we found were N5R and H18Q, which allowed the virus to replicate much better in the presence of suramin, as compared to the wild-type virus. However, a known CHIKV mutant, with a G82R substitution in E2 that adapts the virus for infecting mammalian cells (by interacting with a molecule on their cell surface, heparan sulfate), was more sensitive to suramin.

Lastly, **Chapter 6** summarizes and discusses the key findings presented in this thesis. Their implications in the context of the broader literature are presented, followed by a general discussion and conclusion.

Samenvatting

Titel: Het remmen van chikungunya-virus replicatie - Inzichten in de replicatie van het chikungunya-virus en de antivirale activiteit van suramin

Dit proefschrift is gericht op het beter begrijpen van de moleculaire biologie en replicatie van het chikungunya-virus (CHIKV), en het mechanisme waarmee suramine, een geneesmiddel met een breed spectrum, dit proces kan remmen.

Chikungunya-virus (CHIKV) is een lid van de Togaviridae-familie en kan op mensen worden overgedragen door de beet van geïnfecteerde muggen. Het virus is van Afrikaanse origine en heeft in het verleden grote epidemieën in Azië veroorzaakt, maar de meest recente uitbraak vond plaats in 2014-2015 in een nieuw gebied, Amerika, waar het meer dan 1 miljoen vermoedelijke en bevestigde gevallen veroorzaakte. De infectie manifesteert zich door acute gewrichts- en spierpijn (die jaren kan duren), koorts, en uitslag. Tot op heden is er geen goedgekeurd vaccin of behandeling voor CHIKV-infectie. Een meer uitgebreide inleiding met betrekking tot CHIKV en de virale genoomorganisatie, eiwitfuncties, replicatie, pathogenese en preventieve of therapeutische strategieën is te vinden in **hoofdstuk 1**, de algemene inleiding van het proefschrift. Kort nadat de CHIKVepidemie voorbij was, begon een ander virus, ook door muggen overgebracht en afkomstig uit Afrika, ernstige gezondheidsproblemen te veroorzaken in Midden- en Zuid-Amerika. Dit was het Zika-virus (ZIKV), een lid van de Flaviviridae-familie. Hoewel ZIKV meestal slechts asymptomatische infecties of milde ziekte veroorzaakt, gekenmerkt door lage koorts, uitslag, conjunctivitis en malaise, hebben de epidemieën in Amerika ZIKV-infectie ook in verband gebracht met foetale misvormingen zoals microcefalie en met het Guillain-Barrésyndroom bij volwassenen. Daarnaast werden eerder niet-herkende routes van moeder op kind transmissie en seksuele overdracht ontdekt. Net als in het geval van CHIKV, is er voor ZIKV infectie geen preventieve of therapeutische strategie (vaccins of geneesmiddelen) op de markt. Het geneesmiddel suramine was eerder op de markt gebracht voor de behandeling van parasitaire infecties, maar in de afgelopen 40 jaar is ontdekt dat het ook antikanker en antivirale potentie heeft. Daarom hebben we getest of suramine de replicatie van CHIKV en ZIKV in vitro kan remmen.

In **hoofdstuk 2** wordt de ontwikkeling van een *in vitro* replicatie assay (IVRA) beschreven die was gebaseerd op CHIKV-replicatie/transcriptiecomplexen die werden geïsoleerd uit geïnfecteerde cellen. Deze test kan worden gebruikt om CHIKV-RNA-synthese te bestuderen, en ook om remmers van dit proces te identificeren en om hun werkingsmethoden te bestuderen. Tijdens het optimaliseren van de test werd een nieuwe RNA-soort geïdentificeerd die tijdens CHIKV-infectie wordt geproduceerd, naast het genomische en subgenome RNA dat nodig is voor de productie van specifieke eiwitten die betrokken zijn bij de virale replicatie en virusproductie, maar ook bij interacties met en manipulatie van de gastheercel. De nieuwe soort is RNAII genoemd, naar analogie van
een vergelijkbaar RNA dat in 1997 door Wielgosz en Huang werd geïdentificeerd voor het aan CHIKV-verwante Sindbis-virus (SINV). Er wordt vermoed dat dit RNA de replicatie/ transcriptiecomplexen dirigeert om later in infectie hoofdzakelijk het subgenome RNA te produceren, wat is vereist voor de productie van de structurele eiwitten die worden gebruikt voor de assemblage van nieuwe virusdeeltjes.

Met behulp van de IVRA hebben we in **hoofdstuk 3** suramine geïdentificeerd als een krachtige remmer van CHIKV replicatie. Uit analyses van geïnfecteerde cellen bleek echter dat de werking van suramine afhankelijk is van de remming van een vroege stap van de virale cyclus, namelijk het binnenkomen van het virus in de gastheercel. Verder werden verschillende aan suramine gerelateerde verbindingen getest. Hoewel deze verbindingen niet effectiever waren, gaven ze wel inzicht in de structurele elementen (symmetrie van het molecuul en de aanwezigheid van negatieve ladingen) die belangrijk zijn voor de beide remmende activiteiten van suramine die zijn waargenomen *in vitro* en in celkweek.

Het antivirale effect van suramin is zeer breed en in **hoofdstuk 4** laten we zien dat het ook de replicatie van het nieuw opkomende virus ZIKV kan remmen, door de opname van virus toegang en de biogenese van nieuwe virions te verstoren.

Vervolgens probeerden we te begrijpen hoe suramine de vroege fase van CHIKV infectie kan remmen. **Hoofdstuk 5** beschrijft de identificatie van het CHIKV E2 envelop glycoproteïne als het doelwit van suramine. Mechanistische studies suggereren dat suramine door de interactie met E2 de virushechting aan cellen blokkeert, en de daaropvolgende fusie van het virion met cellulaire membranen. CHIKV kan resistenter worden tegen suramine door mutaties in het E2-eiwit te verwerven. De relevante aminozuursubstituties die we vonden waren N5R en H18Q, waardoor het virus veel beter kon repliceren in de aanwezigheid van suramine, in vergelijking met het wildtype virus. Echter, een bekende CHIKV-mutant met een G82R substitutie in E2, die het virus beter geschikt maakt voor het infecteren van zoogdiercellen (door interactie met een molecuul op hun celoppervlak, heparaan sulfaat), bleek gevoeliger voor suramine.

Tenslotte vat **hoofdstuk 6** de belangrijkste bevindingen van dit proefschrift samen. Hun implicaties in de context van de bredere literatuur worden gepresenteerd, gevolgd door een algemene discussie en conclusie.

Rezumat

Titlu: Inhibarea replicării virusului Chikungunya - Perspective asupra replicării acestui virus și modul de acțiune al compusului suramin

Această lucrare este axată pe înțelegerea particularităților replicării virusului Chikungunya (CHIKV) și a mecanismelor prin care acest proces poate fi inhibat de suramin, un compus cu spectru larg de activitate.

CHIKV este un membru al familiei Togaviridae care se transmite la oameni prin muscătura tântarilor infectati. În trecut acest virus de origine africană a provocat mari epidemii în Asia, dar cea mai recenta epidemie a avut loc între 2014-2015 in America, unde a cauzat peste 1 milion de cazuri confirmate. Persoanele infectate manifestă dureri articulare și musculare acute (care pot dura ani întregi), febră și erupții cutanate. În prezent nu există niciun vaccin sau tratament aprobat pentru infecția cu CHIKV. O introducere mult mai amplă în ceea ce privește CHIKV, organizarea genomului său, funcțiile proteinelor virale, replicarea, patogeneza și strategiile preventive sau terapeutice poate fi găsită în Capitolul 1, introducerea generală a tezei. La scurt timp după terminarea epidemiei cauzate de CHIKV, un alt virus, de asemenea transmis de tânțari și originar din Africa, a continuat să provoace grave probleme de sănătate în America Centrală si de Sud. Acesta este virusul Zika (ZIKV), un membru al familiei *Flaviviridae*. Desi ZIKV cauzează predominant infectii asimptomatice sau boli fără complicații caracterizate prin febră usoară, erupții cutanate, conjunctivită și stare de rău, epidemiile din America au asociat, de asemenea, infecția cu ZIKV de malformatii fetale, precum microcefalia, si sindromul Guillain-Barré la adulti. În plus, au fost descoperite rutele de transmitere a infectiei nerecunoscute anterior, precum de la mamă la copil și prin contact sexual. În mod similar, precum în cazul CHIKV, nu există pe piață o strategie preventivă sau terapeutică (vaccinuri sau medicamente) împotriva infecției cu ZIKV. Cercetările din ultimii 40 de ani au arătat ca suramin, un compus comercializat pentru tratamentul infecțiilor parazitare, are activitate anticanceroasă, dar și antivirală. Prin urmare, am dorit să testăm dacă acest compus ar putea inhiba replicarea celor două virusuri, CHIKV și ZIKV.

Capitolul 2 descrie dezvoltarea unui sistem de replicare *in vitro* (IVRA) care se bazează pe complexe de replicare/transcripție, aparținând virusului CHIKV, izolate din celulele infectate. Acest test poate fi utilizat pentru a studia sinteza ARN-ului viral, precum și pentru a identifica compuși care inhibă acest proces și a le studia modul de acțiune. În timpul optimizării acestui sistem, am identificat un nou tip de ARN care este produs în timpul infecției cu CHIKV, alături de ARN-ul genomic și subgenomic, specii/soiuri/tipuri care sunt necesare pentru sinteza proteinelor virale implicate în procesul de replicare sau în producția de virus nou, precum și în interacțiunea cu celula gazdă. Acest nou tip de RNA a fost numit RNAII, similar cu cel identificat de Huang și Wielgosz (1997) pentru virusul Sindbis (SINV), un virus înrudit cu CHIKV. Suspectăm că acest nou ARN dirijează mai

târziu în cursul infecției complexele de replicare/transcripție spre a produce predominant ARN subgenomic, necesar pentru generarea de proteine structurale care sunt apoi utilizate în asamblarea noilor particule virale.

Folosind acest sistem de replicare *in vitro*, în **capitolul 3** am identificat compusul suramin ca fiind un puternic inhibitor al replicării CHIKV. Cu toate acestea, testele bazate pe celule in cultură au demonstrat că suramin are ca principal mod de acțiune inhibarea unei etape timpurii a ciclului replicativ, și anume intrarea virusului în celula gazdă. În plus, au fost analizați mai mulți compuși înrudiți cu suramin și, deși acești compuși nu au fost mai eficienți, au oferit o perspectivă asupra elementelor structurale (simetria moleculei și prezența sarcinilor negative) care sunt importante pentru activitățile inhibitoare ale compusului suramin observate atât *in vitro*, cât și în culturi de celule mamaliene.

Efectul antiviral al suraminului are un spectru foarte larg, iar în **capitolul 4** arătăm că inhibă, de asemenea, și replicarea virusului ZIKV, interferând atât cu intrarea sa în celula gazdă, cât și cu biogeneza virionilor descendenți.

Ulterior, am explorat cum suraminul poate inhiba etapele prin care CHIKV intră în celula gazdă. **Capitolul 5** descrie identificarea glicoproteinei E2, expusă la suprafața anvelopei virale a lui CHIKV, ca una din țintele compusului. Experimentele din acest capitol sugerează că prin interacțiunea suraminului cu E2 este blocată atașarea virusului la membrana celulei gazdă, precum și fuziunea ulterioară a particulei cu membranele celulare (în urma endocitozei particulelor). CHIKV poate deveni mai rezistent la acțiunea compusului prin dobândirea de mutații în proteina E2. Substituțiile de aminoacizi N5R și H18Q, au permis virusului să se reproducă mult mai bine în prezența suraminului, în comparație cu virusul original. Cu toate acestea, un mutant CHIKV cunoscut în literatură, având substituția G82R în E2 care adaptează virusul pentru infectarea celulelor mamaliene (prin interacțiunea cu o moleculă de pe suprafața celulară a acestora, heparan sulfat), a fost mai sensibil la suramin. Spre final, **capitolul 6** rezumă principalele concluzie acestei lucrări și discută implicațiile acestora în contextul literaturii recente, și se încheie cu o concluzie generală.

Curriculum Vitae

Irina Cristina Albulescu (née Florea), was born on December 25th 1985, in Petroşani, Hunedoara county, Romania. She attended the National College "Mihai Viteazul" in Ploiesti, Prahova County, between 2000-2004, and graduated with a chemistry-biology profile. She continued with her Bachelor studies in Biochemistry (2004-2008) at the University of Bucharest, followed by a Master program in Biochemistry and Molecular Biology (2008-2010) at the same university. From December 2007, Irina began working at the Institute of Cellular Biology and Pathology (ICBP), under the supervision of Dr. Dorin Alexandru, and later of Dr. Anca Gafencu and Dr. Adrian Manea, where she also performed her Bachelor's and Master's thesis projects. During the four years spent at the ICBP, she was promoted from research assistant to scientific researcher.

In 2012, she moved to the Netherlands and volunteered as a scientist in the lab of Dr. Frank van Kuppeveld at Nijmegen Center for Molecular Life Sciences (NCMLS). Soon after, she obtained a PhD studentship in the lab of Dr. Eric Snijder at the Leiden University Medical Center (LUMC), under the direct supervision of Dr. Martijn van Hemert. In addition, she was also an early stage researcher (2012-2015) in the European Initial Training Network EUVIRNA. Her research was focused on host factors involved in the replication of alphaviruses and identification of compounds with antiviral activity. As part of her PhD research project, Irina performed an industrial training stage at Janssen Infectious Diseases in Beerse (Belgium), under the supervision of Dr. Florence Herschke. Between October 2016 and June 2018, she was involved in the EU-funded ZIKAlliance project, still under the supervision of Dr. Martijn van Hemert, working on the characterization of a ZIKV clinical isolate and the identification of antiviral compounds targeting ZIKV. From July 2018 until January 2019 she continued to work at the LUMC under the direct supervision of Dr. Marjolein Kikkert on the EU-funded Zoonoses Anticipation and Preparedness Initiative (ZAPI) project, which concerned the development of a yellow fever virus 17D-based vaccine platform.

In May 2019, Irina has re-joined the group of Dr. Frank van Kuppeveld at the Faculty of Veterinary Medicine in Utrecht and is now involved in the development of vaccination strategies that provide broad protection against antigenically variable pathogens.

List of Publications

Pietilä MK, **Albulescu IC**, Hemert MJV, Ahola T. Polyprotein Processing as a Determinant for *in vitro* Activity of Semliki Forest Virus Replicase. Viruses. 2017 Oct; 9(10).

Hwu JR, Gupta NK, Tsay SC, Huang WC, **Albulescu IC**, Kovacikova K, van Hemert MJ. Bis(benzofuran thiazolidinone)s and bis(benzofuran-thiazinanone)s as inhibiting agents for chikungunya virus. Antiviral Res. 2017 Oct; 146:96-101.

van Boheemen S, Tas A, Anvar SY, van Grootveld R, **Albulescu IC**, Bauer MP, Feltkamp MC, Bredenbeek PJ, van Hemert MJ. Quasispecies composition and evolution of a typical Zika virus clinical isolate from Suriname. Sci Rep. 2017 May; 7(1):2368.

Albulescu IC, Kovacikova K, Tas A, Snijder EJ, van Hemert MJ. Suramin inhibits Zika virus replication by interfering with virus attachment and release of infectious particles. Antiviral Res. 2017 Jul; 143:230-236.

Delang L, Li C, Tas A, Quérat G, **Albulescu IC**, De Burghgraeve T, Guerrero NA, Gigante A, Piorkowski G, Decroly E, Jochmans D, Canard B, Snijder EJ, Pérez-Pérez MJ, van Hemert MJ, Coutard B, Leyssen P, Neyts J. The viral capping enzyme nsP1: a novel target for the inhibition of chikungunya virus infection. Sci Rep. 2016 Aug; 6:31819.

Hoornweg TE, van Duijl-Richter MKS, Ayala Nuñez NV, **Albulescu IC**, van Hemert MJ, Smit JM. Dynamics of Chikungunya Virus Cell Entry Unraveled by Single-Virus Tracking in Living Cells. J Virol. 2016 Apr; 90(9):4745-4756.

Albulescu IC, van Hoolwerff M, Wolters LA, Bottaro E, Nastruzzi C, Yang SC, Tsay SC, Hwu JR, Snijder EJ, van Hemert MJ. Suramin inhibits chikungunya virus replication through multiple mechanisms. Antiviral Res. 2015 Sep; 121:39-46.

Manea A, Manea SA, Todirita A, **Albulescu IC**, Raicu M, Sasson S, Simionescu M. Highglucose-increased expression and activation of NADPH oxidase in human vascular smooth muscle cells is mediated by 4-hydroxynonenal-activated PPAR α and PPAR β/δ . Cell Tissue Res. 2015 Aug; 361(2):593-604.

Scholte FE, Tas A, **Albulescu IC**, Žusinaite E, Merits A, Snijder EJ, van Hemert MJ. Stress granule components G3BP1 and G3BP2 play a proviral role early in Chikungunya virus replication. J Virol. 2015 Apr; 89(8):4457-69.

Albulescu IC*, Tas A*, Scholte FE, Snijder EJ, van Hemert MJ. An *in vitro* assay to study chikungunya virus RNA synthesis and the mode of action of inhibitors. J Gen Virol. 2014 Dec; 95(Pt 12):2683-92. *Joint first authorship.

Olagnier D, Scholte FE, Chiang C, **Albulescu IC**, Nichols C, He Z, Lin R, Snijder EJ, van Hemert MJ, Hiscott J. Inhibition of dengue and chikungunya virus infections by RIG-Imediated type I interferon-independent stimulation of the innate antiviral response. J Virol. 2014 Apr; 88(8):4180-94.

Trusca VG, **Florea IC**, Kardassis D, Gafencu AV. STAT1 interacts with RXRα to upregulate ApoCII gene expression in macrophages. PLoS One. 2012; 7(7):e40463.

Manea A, Manea SA, **Florea IC**, Luca CM, Raicu M. Positive regulation of NADPH oxidase 5 by proinflammatory-related mechanisms in human aortic smooth muscle cells. Free Radic Biol Med. 2012 May; 52(9):1497-507.

Fenyo IM, **Florea IC**, Raicu M, Manea A. Tyrphostin AG490 reduces NAPDH oxidase activity and expression in the aorta of hypercholesterolemic apolipoprotein E-deficient mice. Vascul Pharmacol. 2011 Mar-Jun; 54(3-6):100-6.

Trusca VG, Fuior EV, **Florea IC**, Kardassis D, Simionescu M, Gafencu AV. Macrophagespecific up-regulation of apolipoprotein E gene expression by STAT1 is achieved via long range genomic interactions. J Biol Chem. 2011 Apr; 286(16):13891-904. +

ERRATA SHEET

This errata sheet lists oversights for the doctoral thesis of Irina Cristina Albulescu, titled "Targeting alphavirus replication – Insights into the chikungunya virus replication and the antiviral activity of suramin *in vitro*", Leiden University Medical Center, 2019, ISBN 97894-93184-16-9 (printed and pdf).

Location	Oversight
Page 11, Figure 1	Missing the citation of "the creative commons license, as CC-BY-SA 4.0
	https://creativecommons.org/licenses/by-sa/4.0/"
Page 14. Figure 3	Missing the text "Copyright Massachusetts Medical Society", after "adapted
	from [2]".
Page 18, Figure 4	Missing the citation of "the creative commons license, as CC-BY-SA 4.0
	https://creativecommons.org/licenses/by-sa/4.0/"
Page 122, Figure 1	Missing the citation of "Adapted from Adapted from
	https://talk.ictvonline.org/ictv-reports/ictv_online_report/positive-sense-
	rna-viruses/w/
	Togaviridae, under the creative commons license, as CC-BY-SA 4.0
	https://creativecommons.org/licenses/by-sa/4.0/"